POPULATION DYNAMICS MODELS ON THE RELATION OF SOCIAL NATURE TO THE EPIDEMICS

YING XIE C1ID1001

A thesis submitted in fulfillment of the requirements for the degree of Doctor of Philosophy (Information Sciences): Mathematical Biology, Department of Computer and Mathematical Sciences, Graduate School of Information Sciences, Tohoku University, Sendai, Japan



March 2024

CANDIDATE Ying Xie

candidate id C1ID1001

THESIS Population dynamics models on the relation of social nature to the epidemics

DEGREE AWARDED Doctor of Philosophy (Information Sciences): Mathematical Biology

SUPERVISOR Professor Hiromi Seno

AFFILIATION Department of Computer and Mathematical Sciences Graduate School of Information Sciences Tohoku University, Sendai, Japan

DISSERTATION REFEREES Professor Toshiyuki Sugawa Associate Professor Naoya Fujiwara Associate Professor Kei Funano

© March 2024

DECLARATION

I, Ying Xie, declare that this thesis titled, "Population dynamics models on the relation of social nature to the epidemics " and the work presented in it are my own. I confirm that:

- This work was done wholly or mainly while in candidature for a research degree at this University.
- Where any part of this thesis has previously been submitted for a degree or any other qualification at this University or any other institution, this has been clearly stated.
- Where I have consulted the published work of others, this is always clearly attributed.
- Where I have quoted from the work of others, the source is always given. With the exception of such quotations, this thesis is entirely my own work.
- I have acknowledged all main sources of help.
- Where the thesis is based on work done by myself jointly with others, I have made clear exactly what was done by others and what I have contributed myself.

Ying Xie Sendai, March 2024 Social nature could have a significant influence on the patterns and trends in the spread of infectious disease. Although it is difficult to quantify the social characteristics, mathematical models have become useful tools for understanding the relation of social nature to disease transmission. In this work, we consider three mathematical models incorporating social nature from the aspects of social response, community's policy, and detectability of the disease infection, respectively. The analysis of the model with the incorporation of social response, especially assuming that the infection rate is affected by such a response to the disease spread, shows that the social response could become a cause of recurring outbreaks whilst it must suppress the prevalence. From the model on the spread of a reinfectious disease in a community with the acceptance of visitors, particularly under the assumption that a certain proportion of visitors are immune, we find that the acceptance of visitors could have a significant influence on the disease's endemicity in the community, either suppressive or supportive. Furthermore, we consider a mathematical model on the disease spread by multiple strains with a competitive dominance, assuming the superinfection occurs when a more dominant strain takes over a host infected by a less dominant strain. This model shows that, strains could coexist with the existence of superinfection. Otherwise, the disease becomes eliminated or alternatively the endemic state arises with only the strain which has the largest strain-specific basic reproduction number while all the other strains get eliminated. Such theoretical/mathematical researches could provide better understanding of the complex interplay between social nature and the disease transmission.

Keywords: Epidemic dynamics, Social response, Public health, Recurring outbreaks, Reinfection, Multiple strains, Superinfection, Ordinary differential equations

ACKNOWLEDGE

I would like to express my gratitude to all those who helped me during the writing of this thesis.

My deepest gratitude goes first and foremost to my supervisor, Professor Seno, for his constant encouragement and guidance. His commitment to academic excellence has transformed my studies into an exhilarating and fulfilling experience. Without his consistent and illuminating instruction, I wouldn't have stuck it out and completed this thesis. The passion and commitment he puts into his students is undeserved but immensely appreciated.

Secondly, I would like to express my heartfelt gratitude to Professor Toshiyuki Sugawa, Professor Naoya Fujiwara, and Professor Kei Funano for serving as my dissertation referees.

A big thanks also goes out to Japan Science and Technology Agency (JST), the establishment of university fellowships towards the creation of science technology innovation. The Pioneering Research Support Project provided me with financial support so that I was encouraged to devote myself to conducting the current research with plenty of time.

I would like to thank the Department of Computer and Mathematical Sciences, Graduate School of Information Sciences, Tohoku University for making available a very stimulating environment for a fruitful academic journey, and the staff who have instructed and helped me a lot.

I am also greatly indebted to Ma Xiushuang and Victor Schneider, who are not only my seniors but also my friends for helping and inspiring me a lot. Besides, I thank my lab mates from across the world for making the lab more like a family: Fu zhiqiong, Xiao Yang, Goyal Akshat, Hattori Haruyuki.

Last my thanks would go to my beloved family for their loving considerations and great confidence in me all through these years. They always share my weal and woe. I feel much grateful and heartily owe my achievement to them.

v

CONTENTS

1	INTRODUCTION	1
	1.1 Characteristics of infectious disease spread	1
	1.2 Social natures on epidemics	2
	1.3 Classic epidemic dynamics models incorporating social natures	3
2	SOCIAL RESPONSE BECOMES A CAUSE FOR RECURRING EPIDEMIC	
	OUTBREAKS?	5
	2.1 Assumptions and modeling	5
	2.2 Basic reproduction number	7
	2.3 Mathematical results on the model	7
	2.4 Discussion	.6
3	THE ACCEPTANCE OF VISITORS PROMOTES THE DISEASE SPREAD? 1	8
	3.1 Assumptions and modeling	8
	3.2 Basic reproduction number	2
	3.3 Mathematical results on the model	24
	3.4 Discussion	\$7
4	MULTIPLE STRAINS SUPPORT THE ENDEMICITY? 3	9
	4.1 Assumptions and modeling	9
	4.2 Basic reproduction number	0
	4.3 Mathematical results on the model	ļ1
	4.4 Discussion	8
5	CONLUSION 5	;0
BI	BLIOGRAPHY 5	51
Α	APPENDIX FOR CHAPTER 2 6	i3
В	APPENDIX FOR CHAPTER 3 6	9
С	APPENDIX FOR CHAPTER 4 7	7

1.1 CHARACTERISTICS OF INFECTIOUS DISEASE SPREAD

1.1.1 Recurring epidemic outbreaks

Recurring outbreaks of transmissible diseases such as measles, chickenpox, mumps, and COVID-19 have been observed in our epidemic history^[1-6]. Such recurring outbreaks could have a significant impact on the society, including people's mental health, public health care costs, economies, and development^[7,8]. Observed in the Hong Kong influenza data and other related data, the influenza could show both seasonally periodic and non-periodic outbreaks^[9]. Not only by the seasonal variation of temperature, humidity, and resource availability, such a seasonality of epidemic outbreak could be caused also by a seasonal change in people's social behavior and contact rate^[10–13]. It has been indicated that the oscillation in epidemic dynamics could be induced by the feedback of the social behavior such as the ignorance when the disease prevalence decreases and the enforcement of containment measurements when the prevalence becomes cautious^[14–19].

1.1.2 Reinfectivity

The reinfectivity of disease in this paper means that the immunity gained by either vaccination or recovery is imperfect. For a spreading transmissible disease accompanied with a reinfectivity, the acceptance of visitors must influence the endemicity of such a disease in the community. Then the community's policy must take account of the risk of reinfection for both residents and visitors. Actually there are transmissible diseases with a reinfectivity, including influenza^[20–24], pertussis^[25,26], Lyme disease^[27], hand, foot and mouth disease^[28], malaria^[29,30], tuberculosis^[31–33], Ebola virus disease^[34,35], chronic lung diseases^[36], invasive pneumococcal disease^[37], meningococcal disease^[38], and COVID-19^[39–47], although the reinfectivity has been still requiring scientific researches to understand its kinetics and other nature.

1.1.3 Superinfection

Competition within a host individual for the limited resources between strains from the same species or from different species has been found in various transmissible diseases^[48,49]. Gupta *et al.*^[50] considered the competition between antigenically diverse strains through cross-reacting host immune response, a form of apparent competition, and showed that the competition for nonimmune hosts can shape the frequencies of strains. Le *et al.*^[51] analyzed a coinfection system with n strains from the strains characteristics including transmissibility, clearance rates of single infection and coinfection, and transmission probability from mixed coinfection. Both of Gupta *et al.*^[50] and Le *et al.*^[51] excluded the possibility of superinfection and focused on coinfection models which assume that hosts can be infected by multiple strains simultaneously.

2 INTRODUCTION

Superinfection implies a two-step process: a host individual previously infected by one strain becomes infected with another strain, after a time t, the second strain takes over the infected individual. Such a "takeover" of hosts by the second strain is assumed to be immediate^[52]. Different from the concept of coinfection with multiple strains, superinfection refers to the infection with only one strain at a time or the coexistence duration of two strains is relatively short and one of them is instantaneously excluded by another. Nowak and May^[53] proposed an SIS model with n strains following an order of virulence and assumed that a more virulent strain can superinfect a host who is already infected by a less virulent strain. Wu *et al.*^[54] considered two-strain models in a heterogenous population represented by scale-free networks, incorporating competing interaction between these two strains by superinfection.

1.2 SOCIAL NATURES ON EPIDEMICS

1.2.1 *Social response*

Protective measures such as vaccination and medicine are available to control the disease spreading^[55,56]. In some scientific works, the social behavior has been considered as a key factor to understand the epidemic dynamics ^[57–61]. When people become aware of the spread of a transmissible disease in a community, various media (e.g., TV, newspaper, and SNS) may provide information to alert the presence of a disease spreading over the community^[62]. Then the community may promote or control some behaviors of its members, for instance, wearing a mask, limiting the number of contacts with others, and taking medication or vaccination $[6_3-6_5]$. Such information could induce some qualitative or quantitative changes in the quotidian behavior, which in turn may reduce the susceptibility to the disease^[66]. Especially the report broadcasted by the media on a significant number of infected individuals is very likely to urge people's caution to take such preventive behaviors^[67]. One of the typical behaviors is the social distancing, which is particularly useful and little costly to slow down the epidemic until a vaccine or medicine becomes widely available^[68]. In this paper, we shall call the collective effect of such people's cautious behaviors on the epidemic dynamics by social response.

On the other hand, even when a disease has been spread in a community, people may not respond to the disease due to, for example, the cost or inconvenience to fight the disease spread^[69–71]. We describe here that such a community is *insensitive* if the community is unaware or unresponsive to the disease spread. Even in such a case, a sufficiently large number of infectives, or severe symptoms by the disease may lead the community to show a social response about the disease spread.

1.2.2 Community's policy

Since the globalization in business and tourism becomes crucial more and more for the economical sustainability of local communities, the condition about the acceptance of visitors would be an important part of the community's policy for the public health about a spreading transmissible disease in and out of it. There must be such a decision on the policy by the host community as whether to accept visitors or not, the number of acceptable visitors, and the condition for acceptable visitors. Actually the importance of such a

3

policy on the tourism regulation has been recognized more and more in the post-COVID-19 period^[72–76].

1.2.3 Detectability of strains

Genetic changes in strains can affect the virus's characteristics, such as transmissibility, virulence, antigenicity, and in some cases, the efficacy of vaccines and treatments. For some viruses, like the influenza virus, genetic variations are common and can lead to the emergence of new seasonal flu strains each year^[77,78]. Although the characterization of multiple genotypes strains could contribute to identifying the disease infection, the emergence of mutant or novel strains of infectious diseases has provided a challenge to clinical diagnosis due to the lack of knowledge about the novel strains or the limitation of testing techniques^[79,80].

1.3 CLASSIC EPIDEMIC DYNAMICS MODELS INCORPORATING SOCIAL NATURES

1.3.1 Models incorporating social behaviors

Funk et al.^[82] quantified the impact on the endemicity of a disease in a well-mixed population under the variation of different disease parameters as a consequence of growing awareness in the population. They consider the spread of the awareness in response to the spread of an infectious disease and divide the population into two compartments: aware and unaware. Cabrera et al.^[81] incorporated the social distancing behavior into an SIR model and argued that an effective social distancing could reduce the disease transmission and its effectiveness depends on the nature of the society. Agaba et al.^[83] considered how the dissemination of private awareness arising from direct contact between unaware/aware individuals and that of public awareness through population-wide campaigns affect the disease spread. Both works focused on a substantial fraction of the population, while the effect of social response was not taken into account. Misra et al.^[84] proposed a nonlinear mathematical model to discuss the effect of awareness about a disease spread. Their results indicate that the awareness programs through the media campaign can decrease the disease spread by isolating a fraction of susceptibles from infectives. Basir *et al.*^[85] assumed the rate of becoming aware (resp. unaware) depends on the media campaign, and showed that increasing the rate of implementation of awareness program through the media could reduce the number of infectives.

1.3.2 Models incorporating people's displacement

There have been many investigations concerning the effect of a people's displacement due to social and political unrest as well as the natural migration of disease vectors to new areas on the epidemic outbreak, and especially conducted have been many theoretical/mathematical studies taking into account the possibility of individuals becoming infective during transportation and contributing significantly to transport-related infection (see Wilson^[86] and references therein; especially for the SARS virus transmission, see Wang^[87]). Not only the particular transportation with a long travel, but also the human quotidian mobility as a common phase of the human activity

4 INTRODUCTION

can be considered as one of relevant factors that could cause the spread of a transmissible disease such as influenza^[88,89]. So is the case of today's pandemic of COVID-19 in local regions of every country^[90–92]. In the work presented by Parikh *et al.*^[93], a synthetic population model of the Washington DC metro area was extended to include leisure and business travelers classified as transients. The final size of the epidemic among residents was found to be remarkably higher when transients were included in the simulation of a flu-like disease outbreak. In considering the emerging diseases of wildlife, Tompkins *et al.*^[94] show that the key drivers of such diseases are agents from domestic sources and human-assisted exposure to infectious agents from wild populations. Talking about swine fever otherwise known as hog cholera, wild boar populations are known to serve as reservoir for the disease thereby constituting a great challenge for domestic pig farmers, veterinarians and other stakeholders^[95,96].

SOCIAL RESPONSE BECOMES A CAUSE FOR RECURRING EPIDEMIC OUTBREAKS?

2.1 ASSUMPTIONS AND MODELING

2.1.1 Assumptions

When a transmissible disease spreads in a community, the behavioral change of the members in response to the outbreak may affect the course of its spread. For our mathematical modeling about the epidemic dynamics considered here under the social response, we set up the following assumptions:

- The spreading disease is non-fatal, and the disease-induced death can be negligible (for example, the common cold).
- The recovered individual cannot acquire a long-lasting effective immunity and becomes susceptible again in a sufficiently short period after the recovery.
- The demographic change about the community is negligible in the time scale of considered epidemic dynamics.
- The stronger social response makes the infection rate smaller, for example, with a decrease of individual contact rate.
- The social response follows a natural decay, while the fact of disease spread in the community tends to arouse the response.
- The disease spread may not cause the social response unless the number of infectives becomes enough to concern the people about it. Such a situation defines the social insensitivity. It may depend on the educational or cultural backgrounds of the community members.

2.1.2 Modeling

Let S(t) and I(t) be the susceptible and infective population densities in the community at time t, while M(t) represents the strength of the social





response at time t. With the above assumptions, we shall consider the following model with ordinary differential equations (Figure 1):

$$\frac{dS}{dt} = -\beta(M)IS + qI$$

$$\frac{dI}{dt} = \beta(M)IS - qI$$

$$\frac{dM}{dt} = \Gamma(I) - \mu M,$$
(1)

where q is the recovery rate and μ is the natural decay rate of the social response. The coefficient of disease transmission $\beta = \beta(M)$ is given by a decreasing, positive, and differentiable function of $M \in [0, \infty)$. The initial condition is given by S(0) > 0, I(0) > 0, and M(0) = 0, which means that there is no social response at the beginning of the disease spread. Then people are unconcern about the disease spread. Thus $\beta_0 := \beta(0)$ denotes the coefficient of disease transmission in such a situation of the community with no social response.

The social sensitivity function $\Gamma(I)$ represents the nature of the social response according to its sensitivity to the disease spread.

we assume that the social response does not arise as long as the infective population density is not beyond a threshold value I_c :

$$\Gamma(I) := \begin{cases} 0 & \text{for } I \leqslant I_c; \\ \gamma(I - I_c) & \text{for } I > I_c. \end{cases}$$
(2)

Positive parameter γ is the social sensitivity coefficient, and I_c is the threshold value for infective population density to raise the social response. The threshold value I_c can be regarded as representing the social insensitivity to characterize the community. Parameter γ characterizes the responsiveness of the community that becomes aware of the epidemic situation. As γ gets larger, the strength of social response more sensitively increases with the increase of infectives in the community.

This kind of a switch-off change in the social collective behavior depending on the epidemic situation has been taken into account for the mathematical modeling also in some previous works with somewhat different contexts (for example, [97-101]), while we introduce it here as the nature of social insensitivity as described above. On the other hand, like our model (1) with (2) for I_c \in (0, N), the system with a state-dependent switch in the dynamical nature may be regarded as a *piecewise smooth system* (PSS), or what is sometimes called *Filippov system* or *switching system* [102-107] and references therein, while we will not need to use any specific mathematical technique or knowledge for such a piecewise smooth system in the following analysis.

According to our assumptions, we have S(t) + I(t) = N with a positive constant N for any $t \ge 0$. Then the system (1) can be reduced to the following two-dimensional one:

$$\frac{dI}{dt} = \beta(M)(N-I)I - qI$$

$$\frac{dM}{dt} = \Gamma(I) - \mu M$$
(3)

with the initial condition given by $I(0)\in (0,N)$ and M(0)=0. Note that $I(t)\in (0,N)$ for any $t\geqslant 0.$

We have a special case with $I_c \ge N$. This is the case where $\Gamma(I)$ is always zero for any $I \in (0, N)$ in the epidemic dynamics given by the system (1).

The alternative special case is such that $\gamma = 0$, which makes $\Gamma(I)$ always zero as well. These special cases correspond to the model with no social response which can be regarded as a reference model to our full model (1) with (2).

We have the other special case with $I_c = 0$. In this case, the social response arises for any number of infectives, so that it is regarded as the model for an epidemic dynamics in a community with no insensitivity about the disease spread. This case is mathematically equivalent to a particular case of the model analytically investigated in^[108,109]. There are some other works^[110–120] related to^[108,109], and in part to this work.

2.2 BASIC REPRODUCTION NUMBER

The basic reproduction number \mathscr{R}_0 is defined as the expected number of secondary infectives generated by a single infective individual in a community consisting only of susceptible individuals in the duration of the infectivity of the initial infective individual [52,121-123]. To derive \mathscr{R}_0 for the model (1), we use the condition that $dI/dt|_{t=0} > 0$ for $I(0) \ll 1$ and $S(0) \approx N$. This condition corresponds to the situation in which the basic reproduction number \mathscr{R}_0 could be defined as the supremum for the number of secondary infectives^[124]. From M(0) = 0, we can easily obtain the following condition with the definition of \mathscr{R}_0 for the model (1):

$$\mathscr{R}_0 = \frac{\beta_0 N}{q} > 1,$$

when the number of infectives increases in an early period after the disease invasion. If $\Re_0 < 1$, the number of infectives decreases after the invasion. If $\Re_0 > 1$, the disease is able to spread in the community at least for a while after its invasion.

2.3 MATHEMATICAL RESULTS ON THE MODEL

2.3.1 Non-dimensionalization

Since the total population size N is constant for our model (3) independently of time, we introduce the following non-dimensional transformation of variables and parameters:

$$u = \frac{S}{N}; \ v = \frac{I}{N}; \ \tau = qt; \ \mathscr{R}_0 = \frac{\beta_0 N}{q}; \ \eta = \frac{\gamma N}{q}; \ \theta_c = \frac{I_c}{N}; \ \delta = \frac{\mu}{q},$$

and u = 1 - v. Then the system (3) can be rewritten as

$$\frac{d\nu}{d\tau} = \frac{\beta(M)}{\beta_0} \mathscr{R}_0 \nu (1 - \nu) - \nu$$

$$\frac{dM}{d\tau} = G(\nu) - \delta M$$
(4)

with

$$G(\nu) := \begin{cases} 0 & \text{for } \nu \leqslant \theta_c; \\ \eta(\nu - \theta_c) & \text{for } \nu > \theta_c. \end{cases}$$
(5)

In the subsequent sections, we investigate the dynamical nature of the nondimensionalized system (4) with (5).

2.3.2 Model with no social response

We consider the system (4) when no social response arises, that is, when $M(t) \equiv 0$ for any $t \ge 0$. This corresponds to the case where $\eta = 0$ or $\theta_c \ge 1$ in (5) with M(0) = 0 as mentioned in Section 2.1. From (4), we have

$$\frac{\mathrm{d}\nu}{\mathrm{d}\tau} = \mathscr{R}_0 \nu (1 - \nu) - \nu \tag{6}$$

which mathematically corresponds to the Verhulst model^[125], being wellknown today as the logistic equation. The ordinary equation (6) with the initial condition $v(0) = v_0 > 0$ is given as

$$\nu(\tau) = \begin{cases} \left(1 - \frac{1}{\mathscr{R}_0}\right) \frac{\nu_0}{\nu_0 + \{(1 - 1/\mathscr{R}_0) - \nu_0\}e^{-\tau/(\mathscr{R}_0 - 1)}} & \text{for } \mathscr{R}_0 \neq 1; \\ \frac{1}{\tau + 1/\nu_0} & \text{for } \mathscr{R}_0 = 1. \end{cases}$$

When $\mathscr{R}_0 \leq 1$, $\nu(\tau)$ is monotonically decreasing and approaches the disease-free equilibrium with $\nu = 0$. When $\mathscr{R}_0 > 1$, $\nu(\tau)$ monotonically approaches the endemic equilibrium with $\nu = \nu^* = 1 - 1/\mathscr{R}_0 > 0$.

2.3.3 Model without social insensitivity

In this section, we consider the system (4) without the social insensitivity, that is, with $\theta_c = 0$. This is the case where the community always has a social response whenever a transmissible disease exists in it. From (4) and (5), the model now becomes

$$\frac{d\nu}{d\tau} = \frac{\beta(M)}{\beta_0} \mathscr{R}_0 \nu (1-\nu) - \nu$$

$$\frac{dM}{d\tau} = \eta \nu - \delta M.$$
(7)

This system has been analytically investigated also in^[108,109], as already mentioned about the full model(1) with $I_c = 0$ for (2) at the end of Section 2.1.

The model (7) always has the disease-free equilibrium $E_0(0,0)$ and may have an endemic equilibrium $E_+(v^*, M^*)$ which satisfies that

$$\frac{\beta(M^*)}{\beta_0} \mathscr{R}_0(1-\nu^*) - 1 = 0; \quad \eta \nu^* - \delta M^* = 0.$$
(8)

Then we can get the following result on the existence of equilibria E_0 and E_+ (Appendix A.1):

Lemma 2.1. For the model (4) without the social insensitivity, that is, for the system (7),

- (*i*) *if and only if* $\Re_0 \leq 1$ *, there is only the disease-free equilibrium* E_0 *;*
- (ii) if and only if $\mathscr{R}_0 > 1$, there are the disease-free equilibrium E_0 and the endemic equilibrium E_+ uniquely determined by (8), where $0 < v^* < 1 1/\mathscr{R}_0$.

With this lemma, we can obtain the following result on the stability of existing equilibrium:

Theorem 2.1. For the model (4) without the social insensitivity,

9

- (*i*) *if and only if* $\Re_0 \leq 1$ *, the unique equilibrium* E_0 *is globally asymptotically stable;*
- (*ii*) *if and only if* $\mathscr{R}_0 > 1$ *, there are two equilibria* E_0 *and* E_+ *, where* E_0 *is unstable and* E_+ *is globally asymptotically stable.*

This result corresponds to Theorem 5.1 in^[108] or Theorem 5.2 in^[109], which is on the SIS model with the recruitment of susceptibles to balance the death in the population. Thus our Theorem 2.1 can be proved by the arguments corresponding to their proof with a Lyapunov function for their model with their parameter $\mu = 0$.

2.3.4 Model with social insensitivity

In this section, we consider the model (4) with a social insensitivity, that is, with $\theta_c \in (0, 1)$. The model (4) always has the disease-free equilibrium $E_0(0,0)$ and may have an endemic equilibrium (ν^*, M^*) which satisfies that

$$\frac{\beta(M^*)}{\beta_0}\mathscr{R}_0(1-\nu^*) - 1 = 0; \quad G(\nu^*) - \delta M^* = 0.$$
(9)

We can get the following result on the existence of equilibria (Appendix A.2):

Lemma 2.2. For the model (4) with a social insensitivity,

- (*i*) *if and only if* $\Re_0 \leq 1$ *, there is only the disease-free equilibrium* $E_0(0, 0)$ *;*
- (ii) if and only if $1 < \Re_0 \leq (1 \theta_c)^{-1}$, there are the disease-free equilibrium $E_0(0,0)$ and the endemic equilibrium $E_{+0}(1 1/\Re_0, 0)$;
- (iii) if and only if $\Re_0 > (1 \theta_c)^{-1}$, there are the disease-free equilibrium $E_0(0,0)$ and the endemic equilibrium $E_{++}(v^*, M^*)$ uniquely determined by

$$v^* = 1 - \frac{1}{\mathscr{R}_0} \frac{\beta_0}{\beta(M^*)}; \quad M^* = \frac{\eta}{\delta} (v^* - \theta_c), \tag{10}$$

where $\theta_c < v^* < 1 - 1/\Re_0$.

(iv) $E_{++} \rightarrow E_{+0}$ as $\mathscr{R}_0 \rightarrow (1-\theta_c)^{-1} + 0$.

The equilibrium E_{+0} is the endemic state that people are unconcerned about the disease persisting in the community, while E_{++} is the endemic state that the persisting disease concerns people in the community.

With Lemma 2.2, we can obtain the following result on the stability of existing equilibrium (Appendix A.3):

Theorem 2.2. For the model (4) with a social insensitivity,

- (*i*) *if and only if* $\Re_0 \leq 1$ *, the unique equilibrium* E_0 *is globally asymptotically stable;*
- (*ii*) *if and only if* $1 < \Re_0 \leq (1 \theta_c)^{-1}$ *, there are two equilibria* E_0 *and* E_{+0} *, where* E_0 *is unstable and* E_{+0} *is globally asymptotically stable;*
- (iii) if and only if $\Re_0 > (1 \theta_c)^{-1}$, there are two equilibria E_0 and E_{++} , where E_0 is unstable and E_{++} is globally asymptotically stable.

Independently of whether the social insensitivity exists or not, the model (4) has an endemic equilibrium when and only when $\Re_0 > 1$ while it always has the disease-free equilibrium E_0 . Then there exists the endemic equilibrium E_{+0} or E_{++} , depending on the social insensitivity. In the absence of the social insensitivity (formally, $\theta_c = 0$), the disease and social response coexist when people keep a response to the disease (M > 0). In the presence of a social insensitivity ($0 < \theta_c < 1$), there is a critical condition for \Re_0 . If $1 < \Re_0 \leq (1 - \theta_c)^{-1}$, the disease persists in the community whereas people do not take care of it at the endemic equilibrium. If $\Re_0 > (1 - \theta_c)^{-1}$, the disease and social response coexist at the endemic equilibrium.

From Theorem 2.2, we find that, when $1 < \Re_0 \leq (1 - \theta_c)^{-1}$, the model (4) approaches the endemic equilibrium $E_{+0}(1 - 1/\Re_0, 0)$. In this case, the endemic size of infectives is determined only by the basic reproduction number \Re_0 , that is, $1 - 1/\Re_0$, while there is no social response at the endemic equilibrium. In contrast, for the case where $\Re_0 > (1 - \theta_c)^{-1}$, we can obtain the following result on the parameter dependence of the endemic size v^* and the strength of social response M^* at the endemic equilibrium E_{++} which the system approaches:

Corollary 2.2.1. The endemic size v^* at the endemic equilibrium E_{++} is increasing in terms of \mathscr{R}_0 , δ and θ_c , while it is decreasing in terms of η . The strength of social response M^* at E_{++} is increasing in terms of \mathscr{R}_0 and η , while it is decreasing in terms of δ and θ_c .

The proof of this corollary can be obtained straightforwardly from the sign of the partial derivatives of v^* and M^* with respect to each parameter, where we used $d\beta(M)/dM < 0$ from the assumption for $\beta(M)$.

For a disease with high transmissibility (i.e., large \Re_0), although people show a strong social response to the disease spread, there is a large number of infectives at the endemic equilibrium state due to the high transmissibility of the disease. As for a community of people more sensitive to the disease spread (i.e., small θ_c or large η), or more persistently keeping the social response (i.e., small δ), people show a stronger social response to the disease spread, and such a response can reduce the number of infectives at the endemic equilibrium state.

Furthermore as shown in Appendix A.3, we can obtain the following result on the behavior of the system to approach the equilibrium:

Corollary 2.2.2. *The system* (4) *approaches*

- (*i*) the equilibrium E_0 in a monotonic manner when $\mathscr{R}_0 \leq 1$;
- (ii) the equilibrium E_{+0} in a monotonic manner when $1 < \Re_0 \leq (1 \theta_c)^{-1}$;
- (iii) the equilibrium E_{++} in the following manner when $\Re_0 > (1 \theta_c)^{-1}$:

 $\left\{ \begin{array}{l} a \text{ monotonic manner if } \Delta \geqslant 0; \\ an \text{ oscillatory manner if } \Delta < 0, \end{array} \right.$

where

$$\Delta := \left\{ \frac{\beta(M^*)}{\beta_0} \mathscr{R}_0 \nu^* + \delta \right\}^2 + 4\eta \nu^* \frac{d\ln\beta(M)}{dM} \bigg|_{M=M^*}.$$
 (11)

The behavior of the system (4) to approach E_{++} could have two different manners: monotonic or oscillatory. This is because the second term of (11) is

negative and its absolute value is determined by the derivative of β . Hence the sign of Δ essentially depends on the detailed features of function $\beta(M)$.

As a consequence, we have found the possibility that the social sensitivity could induce a damped oscillation in the epidemic dynamics. The emergence of a damped oscillation caused by the social response has been shown also in some literatures on the epidemic dynamics model (for example,^[14,19], and references therein). Generally, independent of which an epidemic dynamics is approaching a disease-free or endemic state, an oscillatory variation in the infective population size must concern the public health in the community since it appears as recurring outbreaks. In the following part, we shall focus on the condition for the occurrence of such an oscillatory behavior of the system (4) with a social insensitivity.

2.3.5 Extremely fast/slow social response

To understand further the nature of epidemic dynamics by the model (1), we consider here a specific case where the strength of social response changes extremely faster or slower than the epidemic dynamics, and vice versa. Then we mathematically apply the quasi-stationary state approximation (QSSA) for the temporal change in M or v, that is to use $dM/d\tau \approx 0$ or $dv/d\tau \approx 0$ respectively for the system (4)^[124,126,127]. As proved in Appendix A.4, we can find the following nature of the epidemic dynamics with such an extremely fast/slow social response:

Theorem 2.3. *In the case of extremely fast/slow social response, Theorem 2.2 holds, and the system* (4) *necessarily approaches an equilibrium in a monotonic manner.*

Theorem 2.3 indicates that the system (4) does not approach the equilibrium in any oscillatory manner with the extremely fast/slow social response. In other mathematical words, any damped oscillation toward the equilibrium does not occur for the system (4) with the parameter value η and δ sufficiently larger/smaller than any other parameter values. Therefore it is implied that a damped oscillation toward the endemic equilibrium E_{++} could occur only for a finite range of parameter value η or δ when $\Re_0 > (1 - \theta_c)^{-1}$, following Corollary 2.2.2. We will consider the relation of social response and insensitivity to the occurrence of such a damped oscillation in more detail for the model (1) with a specific function $\beta(M)$ in the next section.

2.3.6 A specific model

We consider the model (1) with the following specific function $\beta(M)$:

$$\beta(M) = \frac{\beta_0}{1 + aM}.$$
 (12)

Positive constant a is the efficiency coefficient of the social response with respect to the transmission rate. The larger a means that the social response has the higher efficiency to reduce the risk of infection. The same relation of the social response to the infection coefficient was introduced also in the modified SIR and SEIQR models of ^[14,19,113,118,119].



Figure 2: Bifurcation diagrams about v^* and M^* for the system (13). Solid curves are for stable equilibria E_0^s , E_{+0}^s and E_{++}^s in (a) and (b), for E_{++}^s when $\theta_c < 1 - 1/\Re_0$ and E_{+0}^s when $\theta_c > 1 - 1/\Re_0$ in (c). Dashed lines in (a) and (b) are for the unstable equilibrium E_0^u . Numerically drawn with (a) and (b) $\theta_c = 0.6$; (c) $\Re_0 = 2.5$, and $\alpha = 5.0$; $\delta = 10.0$; $\eta = 5.0$.

The model (1) with (12) becomes

$$\begin{aligned} \frac{\mathrm{dS}}{\mathrm{dt}} &= -\frac{\beta_0}{1+aM}\mathrm{IS} + q\mathrm{I} \\ \frac{\mathrm{dI}}{\mathrm{dt}} &= \frac{\beta_0}{1+aM}\mathrm{IS} - q\mathrm{I} \\ \frac{\mathrm{dM}}{\mathrm{dt}} &= \Gamma(\mathrm{I}) - \mu\mathrm{M}, \end{aligned}$$

with $\Gamma(I)$ defined by (2). The non-dimensionalized system (4) becomes

$$\frac{d\nu}{d\tau} = \frac{\mathscr{R}_{0}}{1+aM}(1-\nu)\nu - \nu$$

$$\frac{dM}{d\tau} = G(\nu) - \delta M,$$
(13)

with G(v) given by (5). From Theorem 2.2, if $\Re_0 > (1 - \theta_c)^{-1}$, there is a globally asymptotically stable endemic equilibrium E_{++} with

$$v^* = \frac{(\mathscr{R}_0 - 1)\delta + a\eta\theta_c}{\mathscr{R}_0\delta + a\eta}; \quad M^* = \frac{(\mathscr{R}_0 - 1 - \mathscr{R}_0\theta_c)\eta}{\mathscr{R}_0\delta + a\eta}.$$
 (14)

We can obtain the bifurcation diagrams as shown in Figure 2 for the system (13) (refer to Corollary 2.2.1). From (14), we can find that v^* is monotonically decreasing in term of an. Since a is the efficiency coefficient of the social response with respect to the transmission rate, the endemic size v^* must become smaller as the social response more efficiently works to reduce the infection risk (see Figure 3).

From Corollary 2.2.2, only when E_{++} exists, the system (13) may approach it in an oscillatory manner (see Figure 4). Since the temporal oscillation of infective population size means recurring outbreaks of the disease spread as already mentioned before, it is worthwhile investigating what condition causes such an oscillatory behavior in the epidemic dynamics by the system (13).

The discriminant (11) of the characteristic equation for the endemic equilibrium E_{++} of the system (13) becomes

$$\Delta = \left(\frac{\nu^*}{1-\nu^*}\right)^2 - 2\left(\delta + \frac{2}{\mathscr{R}_0}a\eta\right)\frac{\nu^*}{1-\nu^*} + \delta^2 \tag{15}$$

from (12) and (14). Then we derive the following result (Appendix A.5):



Figure 3: Parameter dependence of the endemic size v^* at the endemic equilibrium which is E_{++} if $\theta_c \leq 1 - 1/\Re_0$, and E_{+0} if $\theta_c > 1 - 1/\Re_0$ respectively. Numerically drawn with (a) $\delta = 2.0$; (b) $a\eta = 5.0$; (c) $\theta_c = 0.3$, and $\Re_0 = 3.0$.



Figure 4: Temporal variations of v and M for the model (13) with a = 5.0; $\mathcal{R}_0 = 4.0$; $\eta = 5.0$; $\theta_c = 0.3$; (a) $\delta = 10.0$; (b) $\delta = 0.1$. The initial condition is given as $(\nu(0), M(0)) = (0.001, 0.0)$.

Theorem 2.4. When $\Re_0 > (1 - \theta_c)^{-1}$, the system (13) approaches E_{++} with a damped oscillation if and only if $\theta_-^c < \theta_c < \theta_+^c$, where

$$\theta_{\pm}^{c} := \frac{(\delta + a\eta)x_{\pm} - (\mathscr{R}_{0} - 1)\delta}{a\eta(1 + x_{\pm})}$$
(16)

with

$$\mathbf{x}_{\pm} := \left(\delta + \frac{2a\eta}{\mathscr{R}_0}\right) \pm \sqrt{\left(\delta + \frac{2a\eta}{\mathscr{R}_0}\right)^2 - \delta^2}.$$
 (17)

If $\theta_c \leqslant \theta^c_-$ or $\theta_c \geqslant \theta^c_+$, the system approaches an equilibrium in a monotonic manner.

From Theorem 2.4, we can obtain the following detailed result of the condition about the occurrence of a damped oscillation for the system (13) (Appendix A.6):

Corollary 2.4.1. For the system (13), a damped oscillation occurs

- (i) when $1 < \mathscr{R}_0^{inf} < \mathscr{R}_0 < \mathscr{R}_0^c$ and $0 < \theta_-^c < \theta_c < 1 1/\mathscr{R}_0$;
- (ii) when $1 < \mathscr{R}_0^{cc} < \mathscr{R}_0 < \mathscr{R}_0^{sup}$ and $0 < \theta_c < \theta_+^c < 1 1/\mathscr{R}_0$;
- (iii) when $\mathscr{R}_0^c < \mathscr{R}_0 < \mathscr{R}_0^{cc}$ and $0 < \theta_c < 1 1/\mathscr{R}_0$,

where \mathscr{R}_0^{\inf} , \mathscr{R}_0^c , \mathscr{R}_0^{cc} , and \mathscr{R}_0^{sup} are uniquely determined with $\alpha\eta$ and δ , satisfying that $\mathscr{R}_0^{sup} > \mathscr{R}_0^{cc} > \mathscr{R}_0^c > \mathscr{R}_0^{\inf} > 1$. When $\mathscr{R}_0 < \mathscr{R}_0^{\inf}$, $\mathscr{R}_0 > \mathscr{R}_0^{sup}$, or $\theta_c > 1 - 1/\mathscr{R}_0^{cc}$, the damped oscillation does not occur.



Figure 5: (\mathscr{R}_0, θ_c)-dependence of the occurrence of a damped oscillation around the endemic equilibrium E_{++} . For the region out of the filled, the system (13) approaches an equilibrium (either disease-free or endemic) in a monotonic manner, as shown in Corollary 2.2.2. Numerically drawn with (a) $\delta = 1.5$; (b) $\delta = 0$, and $\alpha \eta = 0.5$.

Mathematical definitions of \mathscr{R}_0^{inf} , \mathscr{R}_0^c , \mathscr{R}_0^{cc} and \mathscr{R}_0^{sup} are given in Appendix A.6.

As indicated by Figure 5(a), the disease spread with sufficiently large or sufficiently small \mathscr{R}_0 does not show any damped oscillation, independently of the social sensitivity to it. Only if \mathscr{R}_0 is in a specific range greater than 1, a damped oscillation may occur. Moreover, if the community is insensitive so as to have $\theta_c > 1 - 1/\mathscr{R}_0^{cc}$, the recurring outbreaks do not occur for any \mathscr{R}_0 . The community with sufficiently weak insensitivity (i.e., small θ_c) is very likely to have recurring outbreaks, even though such a community would have a relatively small endemic size at the equilibrium state as shown in Figure 2(c).

For a specific case where the social response never decays with $\delta = 0$, we can get the following simpler result (Appendix A.7):

Corollary 2.4.2. For the system (13) with $\delta = 0$, a damped oscillation occurs

(*i*) when $1 < \Re_0 \leq \Re_0^{cc} = (1 + \sqrt{1 + 16a\eta})/2$ and $0 < \theta_c < 1 - 1/\Re_0$;

(ii) when $\mathscr{R}_0 > \mathscr{R}_0^{cc}$ and $0 < \theta_c < \theta_+^c = 4a\eta/(\mathscr{R}_0 + 4a\eta) < 1 - 1/\mathscr{R}_0$.

Otherwise, the damped oscillation does not occur.

In such a specific case without the decay of the social response, a damped oscillation can occur for a sufficiently weak insensitivity for any $\Re_0 > 1$, as indicated in Figure 5(b). Hence the community which keeps the social response longer would be more likely to show an oscillation in epidemic dynamics for a wide range of the basic reproduction number \Re_0 .

With a decay of the social response, that is, with $\delta > 0$, we can find the following result in the case of extremely poor/effective social response or the case of its extremely fast decay (Appendix A.8):

Corollary 2.4.3. For $a\eta \gg 1$, a damped oscillation occurs, while it does not for $a\eta \ll 1$ or $\delta \gg 1$.

In the extreme case where the community is sufficiently sensitive (i.e., large η) and the social response has sufficiently high efficiency in reducing the susceptibility (i.e., large a), recurring outbreaks necessarily occur. Recurring outbreaks do not occur in the opposite extreme case or in the case where the social response decays so fast (i.e., large δ).



Figure 6: (δ, \mathfrak{an}) -dependence of the occurrence of a damped oscillation. (a) $1 < \Re_0 \leq 2$; (b) $2 < \Re_0 \leq 4$; (c) $\Re_0 > 4$. A damped oscillation could occur only for the region $\Omega_0 \cup \Omega_+ \cup \Omega_-$, where the definitions of Ω_0, Ω_+ , and Ω_- are given in the main text. For the blank region, the damped oscillation does not occur. Boundary curves between Ω_- and Ω_0 , between Ω_0 and Ω_+ , between Ω_+ and blank region, between blank region and Ω_- correspond to $\theta_-^c = 0, \theta_+^c = 1 - 1/\Re_0, \theta_+^c = 0$ and $\theta_-^c = 1 - 1/\Re_0$, respectively.

From Theorem 2.4 and Corollaries 2.4.1–2.4.3, we can get the result shown in Figure 6 on the $(\delta, a\eta)$ -dependence of the occurrence of a damped oscillation (A.6), where the parameter region is classified into four subregions:

$$\begin{aligned} \Omega_{-} &:= \{ (\delta, a\eta) | 0 < \theta_{-}^{c} < 1 - 1/\mathscr{R}_{0} < \theta_{+}^{c} \}; \\ \Omega_{+} &:= \{ (\delta, a\eta) | \theta_{-}^{c} < 0 < \theta_{+}^{c} < 1 - 1/\mathscr{R}_{0} \}; \\ \Omega_{0} &:= \{ (\delta, a\eta) | \theta_{-}^{c} < 0 < 1 - 1/\mathscr{R}_{0} < \theta_{+}^{c} \}, \end{aligned}$$

and the rest. For parameter regions Ω_{-} and Ω_{+} , sufficiently large or small θ_{c} does not cause the damped oscillation. In contrast, for Ω_{0} , a damped oscillation occurs whenever E_{++} exists.

For the other specific case with no social insensitivity, that is, with $\theta_c = 0$, we can get the following result too (Appendix A.9):

Corollary 2.4.4. For the system (13) with $\theta_c = 0$, a damped oscillation occurs when

- (*i*) $\delta = 4(1-1/\Re_0)$ with $\Re_0 > 2$ and $a\eta > (\Re_0 4)^2(\Re_0 1)/\{2\Re_0(\Re_0 2)\} > 0$;
- (*ii*) $4(1-1/\mathscr{R}_0) < \delta < \min\{\mathscr{R}_0, (\mathscr{R}_0+2)(1-1/\mathscr{R}_0)\}$ with $\mathscr{R}_0 > 2$ and $0 < (a\eta)_-^c < a\eta < (a\eta)_+^c$;
- (iii) $\delta < 4(1 1/\Re_0)$ and $a\eta > (a\eta)^c_+ > 0$,

where

$$(\mathfrak{a}\eta)_{\pm}^{\mathbf{c}} = \frac{\mathscr{R}_{0}\delta(\mathscr{R}_{0}-1-\delta) + 2(\mathscr{R}_{0}-1)\{\delta \pm \sqrt{\delta(\mathscr{R}_{0}-1)}\}}{\mathscr{R}_{0}\delta - 4(\mathscr{R}_{0}-1)}.$$
 (18)

With no social insensitivity, a damped oscillation is more likely to occur for sufficiently small δ and sufficiently large $a\eta$. For a community in which people persistently keep the response to the disease spread and such a response can effectively work to reduce the infection risk, recurring outbreaks of epidemic dynamics would be more likely to occur. If people easily lose attention to the disease spread, recurring outbreaks would occur only if the efficiency of such a response is in a certain range. With a social insensitivity, that is, with $\theta_c > 0$, the damped oscillation does not occur for sufficiently small $\alpha\eta$, while it must occur for sufficiently large $\alpha\eta$, as already shown in Corollary 2.4.3. If the effect of social response is persistent with sufficiently slow decay, recurring outbreaks occur only in the community with a sufficiently weak insensitivity, that is, with a sufficiently strong sensitivity for the disease spread.

As shown in Corollary 2.2.1 and in the early part of this section about the $\alpha\eta$ -dependence of ν^* , it is certain that people's attention to disease and attempt to prevent its further spread necessarily suppress the endemicity to make the endemic size ν^* smaller: With the smaller δ or larger $\alpha\eta$, the endemic size ν^* becomes smaller. Hence the above-mentioned likeliness of recurring outbreaks indicates that a damped oscillation in the temporal variation in terms of the number of infectives would occur toward a relatively low level of endemicity. Since we use the word 'outbreaks' here as repeating peaks in the temporal variation about the number of infectives, someone may think the above arguments counter-intuitive. However note that it is not the case.

2.4 DISCUSSION

In this model, we focused on the relation of social response to the likeliness of recurring outbreaks of a spreading disease in a community. For our SIS model, recurring outbreaks may occur only when the system approaches an endemic equilibrium at which the social response remains active. In another endemic case where the social response disappears, the system approaches it in a monotonic manner, that is, the temporal variation of infective population size is monotonic around the endemic equilibrium.

Our model is based on the simplest SIS epidemic dynamics model. As shown in Section 2.3.2, the model with a constant infection coefficient β does not show any oscillatory behavior in the temporal variation of variables. Our modeling is to introduce the effect of social response only on the infection coefficient in the SIS model. Thus, the social response has an effect to vary only the velocity of the state transition from susceptible to infective. As a result of such our modeling, being seen with respect to the closed two dimensional system (3), the temporally varying social response can be regarded as a factor to lead to a temporal change of the intrinsic growth rate in a modified logistic equation with the constant carrying capacity N and the term of a proportional harvesting (qI), which is the ordinary differential equation to govern the temporal change of I. From this mathematical structure about our model, it has no sustained (nondecaying) periodic oscillations, whereas this result implies a possibility of the emergence of a sustained periodic oscillations in some similar models with different assumptions for the epidemic dynamics (for example, see^[14,18,61,100,116–120,128,129]). As we have shown in our analysis on the model, the social response does not alter the endemicity of spreading disease although it can certainly suppress the endemic size. This is because it tends to fade out when the disease becomes about to be eliminated, which we could expect for any human community.

The social response could induce the recurring outbreaks in the epidemic dynamics, while it can suppress the endemic size. The results on our model implied that such recurring outbreaks hardly occur for a disease with sufficiently low or sufficiently high transmissibility. Only for a disease which has a certain intermediate range of transmissibility, such recurring outbreaks may occur, the community approaching an endemic equilibrium at which the social response remains active. A community more sensitive to the disease spread is more likely to have such recurring outbreaks. Moreover, if the social response is more efficient to reduce the risk of infection, the recurring outbreaks are more likely to occur. In contrast, the recurring outbreaks hardly occur for the community much insensitive to the disease spread, while the endemic/epidemic size would become large.

Depending on the characteristics of the community and the nature of the transmissible disease, the social response could become a cause of recurring outbreaks whilst it must suppress the prevalence. Since the nature of actual social response for a disease spread must be one of interesting problems in social sciences, such a possibility that it could be an important factor to cause recurring outbreaks would become more important to our preparation for the future epidemic outbreaks after our experience of the recent pandemic of COVID-19.

THE ACCEPTANCE OF VISITORS PROMOTES THE DISEASE SPREAD?

3.1 ASSUMPTIONS AND MODELING

3.1.1 Assumptions

We consider the spread of a transmissible disease during a short-term period in time after the community starts to accept visitors from the outside, satisfying the following assumptions on the epidemic dynamics:

- [H1] The demographic change in the resident population is negligible in the season.
- [H2] The fatality of disease on the resident and visitor populations is negligible in the season.
- [H₃] The community starts the acceptance of a number of temporal visitors from the outside in the season after a transmissible disease has invaded in it.
- [H4] The entry flow of visitors is constant, that is, the net entry rate is constant independently of time.
- [H5] The exit of visitors from the community follows a constant per capita exit rate.
- [H6] No infected visitor is accepted by the community (i.e., the perfect quarantine), so that every accepted visitor is susceptible or immune to the disease at the entry into the community.
- [H7] A given proportion of visitors is immune at the entry into the community.
- [H8] Only the susceptible residents can get the vaccination to become immune, and it is not available for any visitor staying in the community.
- [H9] Every immune visitor or resident has a possibility to get reinfected (i.e., the *imperfect* or *partial* immunity) during its stay in the community.
- [H10] Infected visitor has the same exit rate as the susceptible visitor, that is, we neglect any influence of the infection on the visitor during the stay in the community.

Assumption H1 indicates a time-independent constant size of resident population during the season in which the epidemic dynamics is going on. We then ignore the death due to the transmissible disease under consideration in the epidemic dynamics too, as indicated by the assumption H2. Assumption H3 indicates that the community accepts the visitors, even undergoing the spread of a transmissible disease, since the fatality of the disease is negligible with the assumption H2. No disease invasion with the visitors is assumed, as indicated by the assumption H6. From the assumptions H4 and H6, the community carries out the perfect regulation for the visitors at the entry according the entry number and the quarantine. Assumption H5 mathematically means that the exit of a visitor from the community follows the homogeneous Poisson process. In a model with ordinary differential equations, it can be introduced with a constant exit rate per visitor. Assumption H7 is to reflect the situation of public health out of the community, applying the mean-field approximation for the proportion of immune visitors at the entry.

Since we assume that the community undergoes the disease spread, the assumption H8 gives the existence of a vaccination program for the residents, while it is not applied to the visitors. However, since the disease is reinfectious as assumed by the assumption H9, the immunity obtained by the vaccination or the recovery of the infection works only to reduce the risk of reinfection. Hence the state transition in terms of the disease follows the susceptible–infective–recovered/immunized–infective (SIRI) structure in our modeling, as used for example in^[130–143].

Remark that the assumed reinfection is not caused by the waning or loss of immunity, which must take a certain period after getting it by the infection or vaccination. As already mentioned in the introduction section, we assume instead the imperfectness of immunity obtained by the infection or vaccination. Hence we do not introduce any specific period or time scale to get reinfected after getting the immunity. Since the infection or vaccination generates an immunity against the disease, the assumption H9 indicates that the immunity is imperfect or partial against the infection, for example, due to the multiplicity of pathogen types (e.g., mutated variants)^[24,132]. Because the cross-immunity is well-known for such similar pathogens by the antigen for a type of pathogen, the reinfection may be suppressed or fail to cause an effective symptom to reproduce and discharge the pathogen to the environment.

For a simplification, the assumption H10 indicates that the exit of visitor is independent of whether the visitor is infected or not during the stay in the community. This assumption would be appropriate when the expected duration of the visitor is sufficiently shorter than the latent period, whereas it may be less appropriate when it is long. As assumed by the assumption H2, we consider a transmissible disease with little serious symptom, so that the assumption H10 would be applicable for visitors infected by such a disease.

3.1.2 Modeling

Generic model

With these assumptions given in the previous section, we shall consider the following model of ordinary differential equations (Figure 7):

Dynamics for the visitor population:

$$\begin{cases} \frac{dS_{v}}{dt} = (1-\rho)\Lambda - \beta \frac{I_{r} + I_{v}}{N+m} S_{v} - qS_{v}; \\ \frac{dI_{v}}{dt} = \beta \frac{I_{r} + I_{v}}{N+m} S_{v} + \varepsilon \beta \frac{I_{r} + I_{v}}{N+m} R_{v} - \gamma I_{v} - qI_{v}; \\ \frac{dR_{v}}{dt} = \rho \Lambda + \gamma I_{v} - \varepsilon \beta \frac{I_{r} + I_{v}}{N+m} R_{v} - qR_{v}; \end{cases}$$
(19)

Dynamics for the resident population:

$$\begin{cases} \frac{dS_{r}}{dt} = -\beta \frac{I_{r} + I_{v}}{N + m} S_{r} - \sigma S_{r}; \\ \frac{dI_{r}}{dt} = -\beta \frac{I_{r} + I_{v}}{N + m} S_{r} + \varepsilon \beta \frac{I_{r} + I_{v}}{N + m} R_{r} - \gamma I_{r}; \\ \frac{dR_{r}}{dt} = \sigma S_{r} + \gamma I_{r} - \varepsilon \beta \frac{I_{r} + I_{v}}{N + m} R_{r}, \end{cases}$$
(20)

where S_v , I_v , and R_v are the subpopulation sizes of susceptible, infective, and immune visitors respectively. Similarly S_r , I_r , and R_r are the corresponding subpopulation sizes about the residents. The population sizes of residents and visitors staying in the community are denoted by $N = S_r + I_r + R_r$ and $m = S_v + I_v + R_v$ respectively. The resident population size N is constant independently of time t, as seen from $d(S_r + I_r + R_r)/dt = 0$ for any t by the system (20). From Assumption H₃ in Section 3.1.1, the community starts the acceptance of visitors from the outside in the considered season after a transmissible disease has already invaded in it. The visitor population size m could be reasonably assumed to be less than the population size of residents N: m < N, whereas we shall not specifically assume so but consider the mathematically general case of m in the subsequent sections without any constraint except for $m \ge 0$. On the other hand, as given in the following part, we will take an assumption on the visitor population size m accompanying with a confinement for the net entry rate of visitors Λ .

All parameters are positive. Parameter ρ is the proportion of immune visitors at the entry ($0 \le \rho \le 1$). Proportion $1 - \rho$ of visitors is susceptible at the entry. Parameter q is the per capita exit rate of visitor. Thus the expected duration of a visitor's stay in the community is given by 1/q.

Parameter $\epsilon\beta$ is the reinfection coefficient for immune resident and visitor, while β is the infection coefficient for susceptible ones. Then the infection forces for the susceptible individual and the immune individual are respectively given by $\beta(I_r + I_v)/(N + m)$ and by $\epsilon\beta(I_r + I_v)/(N + m)$ for both resident and visitor. Remark that, in the setup for our modeling, the visitors do not form any specific subcommunity distinct from the resident population. From Assumption H₃, they are temporal visitors for tourism, business etc. For this setup, we could assume that most of visitors are independent of the others. Thus, for a mathematical simplification, the influence of their movement on the epidemic dynamics is introduced in the epidemic dynamics by the mean-field approximation. Further, although the visitors' contribution to the infection forces would be different from the residents' one because



Figure 7: Scheme of the model for the epidemic dynamics in a community accepting temporal visitors, given by the system of (19) and (20).

of the difference in the mobility/behavioral pattern, the infection forces have their same contributions in our modeling here. This modeling may be regarded as an oversimplification, though we think that our modeling would still worth being considered to get a cue for the discussion about the influence of visitors on the epidemic dynamics within a community.

From Assumption H9 in Section 3.1.1, our modeling assumes that the immunity is imperfect or partial against the infection. Because of the crossimmunity by the obtained antigen, we reasonably assume that $0 \leq \varepsilon \leq$ 1 in our model, so that the reinfection coefficient $\epsilon\beta$ is not beyond the coefficient for susceptible β . That is, the reinfection after the vaccination or recovery from the disease generally has a smaller likelihood than that for the susceptible. For the extremal case of $\epsilon = 1$, the vaccination or recovery does not work at all to reduce the risk of reinfection. For $\varepsilon=0,$ the recovery and vaccination give the perfect immunity so that there is no likelihood of reinfection. Thus the parameter ε means an index for the likelihood of reinfection after the recovery or vaccination, so that it can be regarded as an index for the risk of reinfection. Remark here again that the reinfection in our modeling is assumed to be not due to the waning of immunity (like for the SIRS models) but due to the imperfect immunity, and hence also the vaccinated individual has a risk to get infected, as introduced by Assumption H9.

Parameter γ is the recovery rate of an infective individual, and the recovered individual gets immunity, which is however imperfect. Only the susceptible residents can get the vaccination, with rate σ , and it is not available for any visitor staying in the community. Since the vaccination is imperfect from Assumption H9, it works to reduce the risk of infection but is unable to protect the vaccinated individual from the infection.

Assumption for the visitor population size in the community

According to the dynamics for the visitor population (19), we have

$$\frac{\mathrm{d}\mathfrak{m}}{\mathrm{d}\mathfrak{t}} = \Lambda - \mathfrak{q}\mathfrak{m}$$

where $m = m(t) := S_v(t) + I_v(t) + R_v(t)$ is the visitor population size at time t, Λ the net entry rate of visitors, and q the per capita exit rate of visitor. Now let us consider the stationary situation with respect to the temporal change of visitor population size. This means that the number of visitors is assumed to be stationary, which may be regarded as a consequence of the regulation of their entry by the community, following the assumptions with H4 and H6 given in Section 3.1.1. Therefore we mathematically assume the

situation to satisfy that dm/dt = 0. Hence we put $\Lambda = qm$, and hereafter treat the visitor population size m as a positive constant.

The initial condition

Following the assumption of the stationary visitor population size with $\Lambda = qm$ as given in the above, we have the following dynamics for the visitor population at the disease-free state:

$$\begin{cases} \frac{dS_{v}}{dt} = (1-\rho)qm - qS_{v};\\ \frac{dR_{v}}{dt} = \rho qm - qR_{v}, \end{cases}$$

where R_{ν} means the subpopulation size of immune visitors when they enter in the disease-free community. It can be easily found that this dynamics results in an eventual approach to the equilibrium state such that $(S_v, R_v) \rightarrow$ $((1 - \rho)m, \rhom)$ as $t \rightarrow \infty$ for any non-negative initial condition such that $S_v(0) \ge 0$ and $R_v(0) = m - S_v(0) \ge 0$. For this reason, let us assume the following initial condition for the epidemic dynamics with the model given by the system of (19) and (20):

$$(S_{v}, I_{v}, R_{v}, S_{r}, I_{r}, R_{r}) = ((1 - \rho)m, 0, \rho m, S_{r0}, I_{r0}, R_{r0}),$$
(21)

where $S_{r0} + I_{r0} + R_{r0} = N$ (a positive constant) with $S_{r0} > 0$, $I_{r0} > 0$ and $R_{r0} \ge 0$. This initial condition defines the situation when the community starts the acceptance of visitors from the outside, even under the existence of disease in it. The setup of this initial condition as our modeling follows Assumption H₃ in Section 3.1.1.

3.2 BASIC REPRODUCTION NUMBER

For our model given by the system of (19) and (20), we can derive the following formula of the basic reproduction number \mathcal{R}_0 (Appendix B.1):

$$\begin{aligned} \mathscr{R}_{0} &= \underbrace{\frac{1}{\gamma}}_{\substack{\text{the expected}\\\text{duration}\\\text{infectivity.}}}} \times \left[\underbrace{\beta \frac{N}{N+m}}_{\substack{\text{the supremum of the}\\\text{expected new cases}\\\text{per unit time for the}}_{\substack{\text{resident.}}} + \underbrace{\left\{ \beta \frac{(1-\rho)m}{N+m} + \varepsilon \beta \frac{\rho m}{N+m} \right\}}_{\substack{\text{the supremum of the expected new cases per \\\text{unit time for the visitor, given by the sum of secondary infections for susceptible and \\\text{immune visitors.}}} \right] \\ &= \mathscr{R}_{00} \Big\{ 1 - (1-\varepsilon)\rho \frac{\mu}{1+\mu} \Big\}, \end{aligned}$$
(22)

where $\mu := m/N$, and for a convenience in the following arguments, we define $\mathscr{R}_{00} := \beta/\gamma$, which is the basic reproduction number for the community when no visitor comes in (i.e., m = 0). Note that this basic reproduction number is fundamentally for the epidemic dynamics in the community after it starts the acceptance of visitors.

From the formula (22), we can immediately find that the basic reproduction number \mathscr{R}_0 is less than 1 independently of the nature of accepted visitors if $\mathscr{R}_{00} < 1$. Hence, if the disease fails its invasion in the community with $\mathscr{R}_{00} < 1$ before starting the acceptance of visitors, the number of infectives in the community cannot turn to increase in the early period after the acceptance of visitors starts. As we will see in the later sections of the analysis on our model, this is valid only in the early period after the acceptance of visitors starts.



Figure 8: The dependence of the basic reproduction number \Re_0 given by (22) on parameters $(1 - \epsilon)\rho$ and $\mu := m/N$. Numerically drawn with $\Re_{00} = 1.4$.

As for the dependence of \Re_0 on the nature of accepted visitors, we note that \Re_0 is monotonically decreasing in terms of μ when the visitors contain some immune ones (i.e., $\rho > 0$). Moreover, \Re_0 becomes smaller as the proportion of immune visitors at the entry ρ gets larger. If any visitor is susceptible, that is, when $\rho = 0$, there is no contribution of the visitors to the basic reproduction number \Re_0 , that is, $\Re_0 = \Re_{00}$.

As an extremal case, if the immunity gained by the vaccination or recovery from the disease does not work at all to reduce the risk of reinfection, that is, if $\epsilon = 1$, the basic reproduction number \Re_0 is independent of the acceptance of visitors. This is easily understandable, since the reinfection is regarded as the same as the infection for the susceptible so that the immune individual is regarded as equivalent to the susceptible according to the epidemic dynamics when $\epsilon = 1$. Such an extreme case may be regarded as corresponding to an SIS type of the epidemic dynamics, where the state transition in terms of the disease follows the susceptible–infective–susceptible structure.

As the other extremal case, if the immunity is perfectly effective to make the immune individual unable to be reinfected, that is, if $\epsilon = 0$, the entry of immune visitors works to reduce the value of \mathscr{R}_0 for the community. This extremal case may be regarded as corresponding to an SIR type of the epidemic dynamics, where the state transition in terms of the disease follows the susceptible–infective–removed structure.

From these arguments with the basic reproduction number \mathscr{R}_0 given by (22), We can get the following result on the influence of the acceptance of visitors at the early stage of the disease invasion in the community (Figure 8):

Theorem 3.1. The acceptance of visitors influences the basic reproduction number \mathcal{R}_0 given by (22) for the epidemic dynamics with the system of (19) and (20) as follows:

- (i) The acceptance of visitors makes \$\mathcal{R}_0\$ smaller than \$\mathcal{R}_{00}\$ if and only if the visitors contain some immune, and its decrease becomes bigger as the number of accepted visitors gets larger;
- (ii) When $\Re_{00} > 1$, if

$$\rho \leqslant \rho_{\infty}^{0} \coloneqq \frac{1}{1 - \epsilon} \Big(1 - \frac{1}{\mathscr{R}_{00}} \Big),$$

then $\Re_0 > 1$ independently of the number of accepted visitors;

(iii) When $\mathscr{R}_{00} > 1$, if and only if $\rho > \rho_{\infty}^{0}$, the acceptance of visitors so many as

$$\mu > \frac{1 - 1/\mathscr{R}_{00}}{(1 - \varepsilon)\rho - (1 - 1/\mathscr{R}_{00})}$$

makes $\mathscr{R}_0 < 1$.

The result (*i*) in Theorem 3.1 means that the acceptance of visitors does not help the invasion success of a transmissible disease as long as $\Re_{00} \leq 1$, and instead it could work to suppress the invasion if the community accepts a sufficiently large number of visitors with a sufficiently large proportion of immune, as indicated by the results (*ii*) and (*iii*).

Note that these arguments and result are about the effect of the acceptance of visitors on the temporal change of the number of infectives only in the early period after the community starts the acceptance of visitors. They cannot be necessarily applied for its long-term temporal change. It may be possible that the number of infectives turns to increase in a later period, independently of what final state the epidemic dynamics approaches, as we will actually see in the later sections of the analysis on our model.

3.3 MATHEMATICAL RESULTS ON THE MODEL

3.3.1 Non-dimensional transformation of the system

Since the population sizes of visitors and residents are assumed constant independently of time, the above six dimensional system of (19) and (20) can be mathematically reduced to the following closed four dimensional one, making use of $S_v + I_v + R_v = m$ and $S_r + I_r + R_r = N$:

$$\begin{split} \frac{dS_v}{dt} &= (1-\rho)qm - \beta \frac{I_r + I_v}{N+m} S_v - qS_v; \\ \frac{dI_v}{dt} &= -\beta \frac{I_r + I_v}{N+m} S_v + \varepsilon \beta \frac{I_r + I_v}{N+m} (m - S_v - I_v) - \gamma I_v - qI_v; \\ \frac{dS_r}{dt} &= -\beta \frac{I_r + I_v}{N+m} S_r - \sigma S_r; \\ \frac{dI_r}{dt} &= -\beta \frac{I_r + I_v}{N+m} S_r + \varepsilon \beta \frac{I_r + I_v}{N+m} (N - S_r - I_r) - \gamma I_r. \end{split}$$

Now we apply the following transformation of variables and parameters for this four dimensional system:

$$\begin{split} \tau &:= \gamma t; \; x_v(t) := \frac{S_v(t)}{m}; \; y_v(t) := \frac{I_v(t)}{m}; \; x_r(t) := \frac{S_r(t)}{N}; \; y_r(t) := \frac{I_r(t)}{N}; \\ \mu &:= \frac{m}{N}; \; c := \frac{q}{\gamma}; \; \omega := \frac{\sigma}{\gamma}, \end{split}$$

and then we can derive the following non-dimensinalized system:

$$\begin{split} \frac{dx_{v}}{d\tau} &= (1-\rho)c - \mathscr{R}_{00}\frac{y_{r} + \mu y_{v}}{1+\mu}x_{v} - cx_{v};\\ \frac{dy_{v}}{d\tau} &= -\mathscr{R}_{00}\frac{y_{r} + \mu y_{v}}{1+\mu}x_{v} + \varepsilon\mathscr{R}_{00}\frac{y_{r} + \mu y_{v}}{1+\mu}(1-x_{v} - y_{v}) - (1+c)y_{v};\\ \frac{dx_{r}}{d\tau} &= -\mathscr{R}_{00}\frac{y_{r} + \mu y_{v}}{1+\mu}x_{r} - \omega x_{r};\\ \frac{dy_{r}}{d\tau} &= -\mathscr{R}_{00}\frac{y_{r} + \mu y_{v}}{1+\mu}x_{r} + \varepsilon\mathscr{R}_{00}\frac{y_{r} + \mu y_{v}}{1+\mu}(1-x_{r} - y_{r}) - y_{r}, \end{split}$$
(23)

where $\Re_{00} := \beta/\gamma$ as before. Remark that the symbol \Re_{00} is formally introduced now as a dimensionless parameter for the non-dimensionalized system given by (23), while its meaning is given in Section 3.2 as the basic

reproduction number for the community when no visitor comes in. The initial condition (21) now becomes

$$(x_v(0), y_v(0), x_r(0), y_r(0)) = (1 - \rho, 0, x_{r0}, y_{r0})$$

with $x_{r0} := S_{r0}/N = 1 - I_{r0}/N = 1 - y_{r0}$. In the following sections, we shall analyze the non-dimensionalized system (23) to investigate the nature of the epidemic dynamics by our model with the system of (19) and (20).

3.3.2 Dynamics without reinfection

In this section, we consider the system without reinfection, that is, with $\epsilon = 0$, while we will consider our model only with $\epsilon > 0$ in the subsequent sections. For the system (23) with $\epsilon = 0$, we can easily find that $x_r \to 0$ and $y_r \to 0$ as $\tau \to \infty$. In other words, since the epidemic dynamics for the resident population is governed by an SIR model with the continuous vaccination for the susceptibles, the disease eventually disappears in the resident population, and the residents come to make no contribution to the epidemic dynamics. This means that, for our model of (19) and (20) *without reinfection*, no endemic state can be established as long as the community does not accept any visitor from the outside. Thus we consider the case of $\mu > 0$ hereafter in this section, when the community accepts the visitors.

By the local stability analysis with the eigenvalues of Jacobi matrix at the equilibrium, we can easily find that the endemic equilibrium $E_{+0}(\tilde{x}_v^*, \tilde{y}_v^*, 0, 0)$ with

$$\tilde{x}_{v}^{*} = \frac{(1+c)(1+\mu)}{\mathscr{R}_{00}\mu}; \quad \tilde{y}_{v}^{*} = c\left(\frac{1-\rho}{1+c} - \frac{1+\mu}{\mathscr{R}_{00}\mu}\right)$$
(24)

is locally asymptotically stable when it exists. Then we can get the following theorem on the epidemic dynamics given by (23) with $\epsilon = 0$ (Appendix B.2):

Theorem 3.2. For the system (23) with $\epsilon = 0$, if and only if the condition

$$\mathscr{R}_{00} > \frac{1}{1-\rho} \Big(1 + \frac{1}{\mu} \Big) (1+c)$$
 (25)

is satisfied, the endemic equilibrium $E_{+0}(\tilde{x}_v^*, \tilde{y}_v^*, 0, 0)$ with (24) uniquely exists, and it is globally asymptotically stable. Otherwise, the disease-eliminated equilibrium $E_{00}(1 - \rho, 0, 0, 0)$ is globally asymptotically stable.

See Figure 9(a, b) for numerical examples.

From the basic reproduction number \mathscr{R}_0 given by (22) with $\epsilon = 0$, we can easily find that $\mathscr{R}_0 > 1$ if the condition (25) is satisfied, while the inverse does not necessarily hold. Hence we can get the following result:

Corollary 3.2.1. Even when the disease successfully invades in the community with $\Re_0 > 1$, the disease without reinfection eventually gets eliminated unless the condition (25) is satisfied.

When the condition (25) is unsatisfied with $\Re_0 > 1$, the number of infectives increases at the initial stage of disease spread in the community, and then it eventually turns to decrease toward zero, as numerically exemplified in Figure 9(a).

For $\rho = 1$, the condition (25) cannot be satisfied:



Figure 9: Temporal variations by the system (23). Numerically drawn with (a) $(\varepsilon, \mu, \rho) = (0.0, 0.2, 0.1)$ ($\mathscr{R}_0 = 7.87$); (b) $(\varepsilon, \mu, \rho) = (0.0, 0.5, 0.1)$ ($\mathscr{R}_0 = 7.73$); (c) $(\varepsilon, \mu, \rho) = (0.1, 0.2, 0.1)$ ($\mathscr{R}_0 = 7.88$); (d) $(\varepsilon, \mu, \rho) = (0.1, 0.5, 0.8)$ ($\mathscr{R}_0 = 6.08$); and commonly $\mathscr{R}_{00} = 8.0$; c = 1.0; $\omega = 1.0$; $(x_v(0), y_v(0), x_r(0), y_r(0)) = (1 - \rho, 0.0, 0.99, 0.01)$. In (a, d), the system approaches the disease-eliminated equilibrium, and in (b, c), it approaches the endemic equilibrium.

Corollary 3.2.2. *If the community accepts only immune visitors, the epidemic dynamics without reinfection necessarily approaches the disease-eliminated equilibrium.*

Therefore, when the reinfection is impossible/negligible, the acceptance of visitors with a high immune proportion at the entry does not cause the endemicity of disease.

We must remark that the endemic equilibrium E_{+0} is sustained only by the visitor subpopulation, while no resident contributes to the epidemic dynamics at the equilibrium because all residents have eventually become immune by the past infection or vaccination (Figure 9(a, b)). From Theorem 3.2, for the disease with a sufficiently high infectivity, the acceptance of many visitors with a sufficiently small immune proportion at the entry can cause such an apparent endemicity in the community. Therefore, as seen in Figure 9(a, b), when the community successfully controls and reduces the number of visitors to make μ sufficiently small, or if the community suspends accepting the visitors, the endemic state can be disrupted, and then the disease gets eliminated in the community. However, as we will see in the subsequent sections, if the disease is accompanied by a reinfectivity, this could not be the case (Figure 9(c, d)).

3.3.3 Dynamics with no visitor

Next we consider the model of (19) and (20) with reinfection, that is, with $\varepsilon > 0$, when the community does not accept any visitor from the outside. Thus we analyze the following system derived from (23) with $x_v = y_v \equiv 0$ and $\mu = 0$:

$$\frac{dx_{r}}{d\tau} = -\mathscr{R}_{00}y_{r}x_{r} - \omega x_{r};$$

$$\frac{dy_{r}}{d\tau} = (1 - \varepsilon)\mathscr{R}_{00}y_{r}x_{r} - \varepsilon \mathscr{R}_{00}y_{r}^{2} - (1 - \varepsilon \mathscr{R}_{00})y_{r}.$$
(26)



Figure 10: Application of the isocline method for the system with no visitor (26) when (a) $\epsilon \mathscr{R}_{00} < 1$; (b) $\epsilon \mathscr{R}_{00} = 1$; (c) $\epsilon \mathscr{R}_{00} > 1$.

As already mentioned in Section 3.2, the basic reproduction number for this epidemic dynamics is defined as $\Re_0 = \Re_{00} = \beta/\gamma$. It is easy to find that there are two feasible equilibria for this system of (x_r, y_r) : the disease-eliminated equilibrium $E_0(0,0)$ and the endemic equilibrium $E_+(0,1-1/(\epsilon \Re_{00}))$. The endemic equilibrium E_+ exists when and only when $\epsilon \Re_{00} > 1$.

Making use of the local stability analysis for the equilibrium of the system (26), we can easily find that the equilibrium E_0 is locally asymptotically stable if $\epsilon \mathscr{R}_{00} < 1$, and unstable if $\epsilon \mathscr{R}_{00} > 1$. The endemic equilibrium E_+ is locally asymptotically stable whenever it exists. Taking account of the result on the local stability of equilibria, the isocline method for the two dimensional system (26) can further give the following result (see Figure 10):

Theorem 3.3. For the system (26) with no visitor,

- (*i*) *if and only if* $\epsilon \mathscr{R}_{00} \leq 1$ *, the disease-eliminated equilibrium* E_0 *is globally asymptotically stable;*
- (ii) if and only if $\epsilon \mathscr{R}_{00} > 1$, the endemic equilibrium E_+ exists and is globally asymptotically stable, while E_0 is unstable.

This result was shown for a mathematically equivalent SIRI model in^[131].

From this result, we find that, even with the basic reproduction number $\Re_{00} > 1$, the community approaches the disease-eliminated equilibrium E_0 if $\Re_{00} \leq 1/\epsilon$ with $0 < \epsilon < 1$. In such a case, the number of infectives increases at the initial stage of disease spread in the community, and then it eventually turns to decrease toward zero. If and only if the basic reproduction number is sufficiently large as $\Re_{00} > 1/\epsilon$, the disease becomes endemic in the community.

As shown in the previous section, no endemicity arises in the community with a non-reinfectious disease when no visitor is accepted. Now the result obtained in this section indicates that the endemicity of a disease can arise in the community even with no visitor if the disease has both sufficiently high infectivity ($\Re_{00} > 1$) and sufficiently high reinfectivity ($\varepsilon > 1/\Re_{00}$).

3.3.4 Dynamics with visitors

Disease-eliminated equilibrium

For the system (23) with visitors, we can get the following result on the local stability of the disease-eliminated equilibrium $E_{00}(1-\rho, 0, 0, 0)$ (Appendix B.3):

Theorem 3.4. The disease-eliminated equilibrium E_{00} is unstable if

$$\epsilon \mathscr{R}_{00} > G(\mu, \rho) := \left[\left\{ \frac{1}{\epsilon} (1-\rho) + \rho \right\} \frac{1}{1+\epsilon} \frac{\mu}{1+\mu} + \frac{1}{1+\mu} \right]^{-1}, \quad (27)$$
while it is locally asymptotically stable if the inverse inequality of (27) is satisfied.

It can be easily seen that the condition (27) becomes equivalent to (25) as $\epsilon \rightarrow 0$. This shows a mathematical consistency of Theorem 3.4 to Theorem 3.2.

Moreover we can prove the following result about the relation of the basic reproduction number \mathscr{R}_0 defined by (22) to the condition (27) (Appendix B.4):

Corollary 3.4.1. When $\mathscr{R}_0 \leq 1$, the disease-eliminated equilibrium E_{00} is locally asymptotically stable.

This result is consistent with the biological/epidemiological meaning of the basic reproduction number with respect to the invasion success of a disease in a population, which was described in Section 3.2 and references therein. Under the condition that $\Re_0 < 1$ at the initial stage of a disease invasion in a population, the number of infectives is expected to decrease toward the disease elimination. In a sense of epidemic dynamics, such a decline of the infective subpopulation toward the elimination must follow the locally asymptotic stability of the disease-free equilibrium (as referred in most literatures), which corresponds here to the disease-eliminated equilibrium E_{00} . The result of Corollary 3.4.1 shows this consistency.

Endemic equilibrium

From the equations of (23), if the endemic equilibrium $E_{++}(x_v^*, y_v^*, x_r^*, y_r^*)$ with $y_v^* > 0$ and $y_r^* > 0$ exists for $\rho < 1$, it must satisfy that

$$\begin{aligned} \mathscr{R}_{00} \frac{y_{r}^{*} + \mu y_{v}^{*}}{1 + \mu} &= c \frac{1 - \rho - x_{v}^{*}}{x_{v}^{*}}, \\ y_{v}^{*} &= \frac{1 - \rho - x_{v}^{*}}{x_{v}^{*}} \frac{(1 - \epsilon) x_{v}^{*} + \epsilon}{1 + 1/c + \epsilon (1 - \rho - x_{v}^{*})/x_{v}^{*}}, \\ y_{r}^{*} &= \frac{1 - \rho - x_{v}^{*}}{x_{v}^{*}} \frac{\epsilon}{1/c + \epsilon (1 - \rho - x_{v}^{*})/x_{v}^{*}}, \end{aligned}$$
(28)

and $x_r^* = 0$. In contrast, at the endemic equilibrium E_{++} for $\rho = 1$, we have

$$\begin{split} & \epsilon \mathscr{R}_{00} \frac{y_r^* + \mu y_v^*}{1 + \mu} (1 - y_v^*) - (1 + c) y_v^* = 0, \\ & \epsilon \mathscr{R}_{00} \frac{y_r^* + \mu y_v^*}{1 + \mu} (1 - y_r^*) - y_r^* = 0, \end{split} \tag{29}$$

and $x_v^* = 0$, $x_r^* = 0$, instead of (28). We can obtain the following result on the existence of E₊₊ (Appendix B.5):

Theorem 3.5. *The endemic equilibrium* E_{++} *uniquely exists if and only if the condition* (27) *is satisfied.*

Hence, when the disease-eliminated equilibrium E_{00} is locally asymptotically stable, the endemic equilibrium E_{++} does not exist, and when E_{00} is unstable, E_{++} uniquely exists.

For the local stability of the endemic equilibrium E_{++} for (23), we can get the following result:

Theorem 3.6. When the endemic equilibrium E_{++} exists, it is locally asymptotically stable.

This theorem can be proved by the eigenvalue analysis on the Jacobi matrix for (23) at the endemic equilibrium E_{++} , applying the Routh-Hurwitz

criterion (Appendix B.6). Although we could not get any analytical result on the global stability of the endemic equilibrium E_{++} , our numerical calculations imply that it is globally asymptotically stable when it exists. We then have the mathematical consistency of Theorems 3.4, 3.5, and 3.6 to Theorem 3.2 as for the case with no reinfection, $\epsilon = 0$.

3.3.5 Influence of the acceptance of visitors

Shift in endemicity

First we can easily find the following features of $G(\mu, \rho)$ defined in (27):

- $G(0, \rho) = 1.$
- $G(\mu, \rho)$ is monotonically increasing in terms of ρ for any $\mu > 0$.

• $G(\mu, \rho)$ is $\begin{cases}
monotonically increasing in terms of <math>\mu$ if $\rho > \rho_s := 1 - \frac{c}{1/\varepsilon - 1}; \\
constant (\equiv 1) independently of <math>\mu$ if $\rho = \rho_s; \\
monotonically decreasing in terms of <math>\mu$ if $\rho < \rho_s.
\end{cases}$

• $G(\mu, \rho) < 1$ for any positive μ and $\rho < \rho_s$ if and only if $\varepsilon < 1/(1+c)$.

• $G(\mu, 0)$ is $\begin{cases}
\text{monotonically increasing in terms of } \mu \text{ if } \rho_s < 0, \text{ that is, } \varepsilon > \frac{1}{1+c}; \\
\text{constant } (\equiv 1) \text{ independently of } \mu \text{ if } \rho_s = 0, \text{ that is, } \varepsilon = \frac{1}{1+c}; \\
\text{monotonically decreasing in terms of } \mu \text{ if } \rho_s > 0, \text{ that is, } \varepsilon < \frac{1}{1+c}.
\end{cases}$

•
$$G(\mu, 0) \begin{cases} > 1 \text{ for any } \mu > 0 \text{ and } \varepsilon > \frac{1}{1+c}; \\ > 1 \text{ for any } \mu > 0 \text{ and } \varepsilon < \frac{1}{1+c}. \end{cases}$$

•
$$\inf_{\mu \in (0,\infty)} G(\mu, \rho) = \begin{cases} G(0, \rho) = 1 \text{ for } \varepsilon \ge \frac{1}{1+c}; \\ \lim_{\mu \to \infty} G(\mu, \rho) = G_{\infty}(\rho) \coloneqq \frac{1+c}{(1-\rho)/\varepsilon + \rho} < 1 \text{ for } \varepsilon < \frac{1}{1+c} \end{cases}$$

•
$$\sup_{\mu \in (0,\infty)} G(\mu, \rho) = \begin{cases} G_{\infty}(\rho) \ge 1 \text{ for } \rho \ge \rho_{s}; \\ G(0, \rho) = 1 \text{ for } \rho < \rho_{s}. \end{cases}$$

•
$$\inf_{\substack{(0,\infty) \times (0,1)}} G(\mu, \rho) = \begin{cases} G(0, 0) = 1 \text{ for } \varepsilon \ge \frac{1}{1+c}; \\ \lim_{\mu \to \infty} G(\mu, 0) = \varepsilon(1+c) < 1 \text{ for } \varepsilon < \frac{1}{1+c}. \end{cases}$$

•
$$\sup_{\substack{(0,\infty) \times (0,1)}} G(\mu, \rho) = \lim_{\mu \to \infty} G(\mu, 1) = 1 + c. \end{cases}$$

Then from these mathematical features of $G(\rho, \mu)$, and Theorems 3.4, 3.5, and 3.6, we can get the following result on the disease endemicity in the community accepting the visitors (see Figures 11 and 12):



Figure 11: Parameter region and boundary indicated by the condition (27). The boundary curve is given by $G(\mu, \rho)$. (a) $\varepsilon < 1/(1 + c)$; (b) $\varepsilon = 1/(1 + c)$; (c) $\varepsilon > 1/(1 + c)$. Numerically drawn with (a) $\varepsilon = 0.10$; (b) $\varepsilon = 0.25$; (c) $\varepsilon = 0.40$, commonly for c = 3.0. Solid curves are for $\mu = 0.25$, 0.5, 0.75, 1.0 in each figure. Dotted curve indicates $G_{\infty}(\rho)$.



Figure 12: (ϵ , \Re_{00})-dependence of the endemicity, derived from the condition (30) in Theorem 3.7. Numerically drawn for (a) $\rho = 0.0$; (b) $\rho = 0.6$; (c) $\rho = 1.0$, commonly with c = 1.0. For the region Ω_+ , the acceptance of visitors may change the endemic situation of the community for the disease-eliminated equilibrium as described in Theorem 3.8, while, for the region Ω_- , it may drive the situation of the community approaching the disease-eliminated equilibrium toward the endemic equilibrium as described in Theorem 3.9. For the region out of Ω_- and Ω_+ , the endemicity is independent of whether the community accepts visitors or not.

Theorem 3.7. Independently of whether the community accepts the visitors or not, it necessarily approaches an endemic equilibrium if $\epsilon \mathscr{R}_{00} \ge \max [1, G_{\infty}(\rho)]$, while it necessarily approaches the disease-eliminated equilibrium if $\epsilon \mathscr{R}_{00} \le \min [G_{\infty}(\rho), 1]$, where

$$G_{\infty}(\rho) := \lim_{\mu \to \infty} G(\mu, \rho) = \frac{1+c}{(1-\rho)/\epsilon + \rho}$$

Only when

$$\min\left[\mathsf{G}_{\infty}(\rho), 1\right] < \varepsilon \mathscr{R}_{00} < \max\left[1, \, \mathsf{G}_{\infty}(\rho)\right],\tag{30}$$

the endemicity could significantly depend on the acceptance of visitors.

The inequality (30) gives a necessary condition for which the acceptance of visitors could change the epidemic situation in the community from the endemic equilibrium to the disease-eliminated equilibrium or vice versa. The corresponding parameter regions are shown as Ω_{-} and Ω_{+} in Figure 11. Numerical examples of such a change of endemicity by the acceptance of visitors are given in Figure 13(a, c).

Further from the monotonicity of $G(\mu, \rho)$ in terms of μ and ρ as described in the above, we find the following result on the condition with respect to the influence of the acceptance of visitors on the endemicity in the community:



Figure 13: Temporal variations of infective subpopulations y_r and y_v by the systems (23) and (26). Numerically drawn for the model (26) until $\tau = 40$ until $\tau = 40$ and model (23) for $\tau > 40$, with (a) (ε , μ , ρ) = (0.2, 0.9, 0.3) ($\Re_0 = 3.55$; $\varepsilon \Re_{00} = 0.8$); (b) (ε , μ , ρ) = (0.3, 0.9, 0.3) ($\Re_0 = 3.60$; $\varepsilon \Re_{00} = 1.2$); (c) (ε , μ , ρ) = (0.3, 0.9, 0.9) ($\Re_0 = 2.81$; $\varepsilon \Re_{00} = 1.2$); and commonly $\Re_{00} = 4.0$; c = 1.0; $\omega = 1.0$; ($x_r(0), y_r(0)$) = (0.99, 0.01); ($x_v(40), y_v(40)$) = (1 - ρ , 0.0). In (a) and (c), the endemicity is changed before and after starting the acceptance of visitors, while in (b) the system remains at an endemic state before and after it.

Corollary 3.7.1. *Independently of whether the community accepts the visitors or not, it necessarily approaches an endemic equilibrium if*

$$\epsilon \mathscr{R}_{00} \geqslant \sup_{(0,\infty) \times (0,1)} \mathrm{G}(\mu,\rho) = 1 + c,$$

while it necessarily approaches the disease-eliminated equilibrium if

$$\varepsilon \mathscr{R}_{00} \leqslant \inf_{(0,\infty) \times (0,1)} G(\mu,\rho) = \min \big[1, \ \varepsilon (1+c) \big].$$

Only when

$$\min\left[1, \ \epsilon(1+c)\right] < \epsilon \mathcal{R}_{00} < 1+c, \tag{31}$$

the endemicity could significantly depend on the acceptance of visitors.

As seen in Figures 11 and 12, such an influence to cause a change of endemicity depends on the nature of accepted visitors (i.e., the number, the immune proportion, and the duration of stay).

Moreover, from the features of $G(\mu, \rho)$ described in the above, we find that, if $\epsilon \mathscr{R}_{00} \ge G_{\infty}(\rho) \ge 1$, the disease is endemic independently of how many visitors the community accepts even under the condition (31), as seen in Figure 11. Thus, in comparison to the result for the community with no visitor (i.e., $\mu = 0$) given by Theorem 3.3, we can get the following result (see Figure 12):

Theorem 3.8. Suppose that the disease was endemic under the condition that $1 < \varepsilon \mathscr{R}_{00} < 1 + c$ before the community accepts visitors. If the community accepts visitors with an immune proportion

$$\rho \leqslant \rho_{\infty} := \frac{1}{1 - \epsilon} \left(1 - \frac{1 + c}{\mathscr{R}_{00}} \right), \tag{32}$$

the disease remains endemic independently of how many visitors are accepted. If the community accepts visitors with an immune proportion $\rho > \rho_{\infty}$, then the acceptance of visitors so many as

$$\mu > \mu_{c} := \frac{1 - 1/(\varepsilon \mathscr{R}_{00})}{1/(\varepsilon \mathscr{R}_{00}) - \{(1 - \rho)/\varepsilon + \rho\}/(1 + c)}$$
(33)



Figure 14: (ρ, μ) -dependence of the endemicity, derived from the condition (27) with the results given by Theorems 3.4, 3.5, and 3.6: (a, b) $\epsilon < 1/(1 + c)$; (c) $\epsilon > 1/(1 + c)$. Numerically drawn for (a) $(\mathscr{R}_{00}, \epsilon, c) = (4.0, 0.2, 1.0)$ $(\epsilon \mathscr{R}_{00} = 0.8)$; (b) $(\mathscr{R}_{00}, \epsilon, c) = (4.0, 0.3, 1.0)$ $(\epsilon \mathscr{R}_{00} = 1.2)$; (c) $(\mathscr{R}_{00}, \epsilon, c) = (1.8, 0.8, 1.0)$ $(\epsilon \mathscr{R}_{00} = 1.44)$, each of which satisfies the condition (31) in Corollary 3.7.1.

makes the community approach the disease-eliminated equilibrium. Even if $\rho > \rho_{\infty}$, the acceptance of visitors with $\mu \leq \mu_c$ does not sufficiently shift the endemicity, and the disease remains endemic.

The critical value ρ_{∞} satisfies the equation $\epsilon \mathscr{R}_{00} = G_{\infty}(\rho_{\infty})$. The latter case of $\rho > \rho_{\infty}$ in Theorem 3.8 corresponds to the parameter region Ω_+ in Figure 12. Figure 14(b) shows a numerical example of the (ρ, μ) -dependence in such a case when $\epsilon \mathscr{R}_{00} > 1$.

Since μ_c defined by (33) is monotonically decreasing in terms of ρ when $\epsilon \mathscr{R}_{00} > 1$, we note that

$$\mu_{c} \ge \mu_{c} \big|_{\rho=1} = \mu_{c1} := \frac{1 - 1/(\epsilon \mathscr{R}_{00})}{1/(\epsilon \mathscr{R}_{00}) - 1/(1+c)},$$
(34)

where $\mu_{c\,1}>0$ for $1<\varepsilon\mathscr{R}_{00}<1+c.$ Hence we get the following corollary:

Corollary 3.8.1. When $1 < \varepsilon \mathscr{R}_{00} < 1 + c$, if the community accepts visitors few enough to have $\mu < \mu_{c1}$, the disease remains endemic independently of how much proportion of visitors is immune at the entry.

This result is numerically pointed out in Figure 14(b, c).

Moreover we note that, when ρ_{∞} defined by (32) is negative, that is, when $\mathscr{R}_{00} < 1 + c$, the first case in Theorem 3.8 does not occur. Then the (ρ, μ) -dependence becomes as shown by Figure 14(c), where there is a finite value of μ beyond which the community approaches the disease-eliminated equilibrium, independently of the immune proportion in the visitors at the entry:

$$\mu_{c0} := \mu_c \big|_{\rho=0} = \frac{1 - 1/(\varepsilon \mathscr{R}_{00})}{1/(\varepsilon \mathscr{R}_{00}) - 1/\{\varepsilon (1+c)\}}.$$
(35)

Corollary 3.8.2. When $1 < \epsilon \mathscr{R}_{00} < \epsilon(1 + c)$, if the community accepts visitors so many as $\mu > \mu_{c0}$, the disease goes eliminated independently of how much proportion of visitors is immune at the moment of their immigration. When $\epsilon(1 + c) \leq \epsilon \mathscr{R}_{00} < 1 + c$, only the acceptance of visitors with $\rho > \rho_{\infty}$ and $\mu > \mu_c$ can change the endemicity and lead the community to the disease-eliminated equilibrium.

Hence the value μ_{c0} gives a sufficient number of accepted visitors which is effective to suppress the spread of disease in the community when $1 < \epsilon \Re_{00} < \epsilon (1 + c)$. See the numerical examples in Figure 13(b, c).



Figure 15: Parameter region and boundary indicated by the condition (27) with the results given by Theorems 3.4, 3.5, and 3.6 when the community accepts only immune visitors with $\rho = 1$. Numerically drawn with $\Re_{00} = 4.0$ and c = 1.0.

In contrast, when the risk of reinfection is so weak as $\epsilon < 1/(1 + c)$, the acceptance of visitors may cause the inverse influence on the epidemic dynamics, as numerically indicated by Figure 14(a):

Theorem 3.9. Suppose that the disease was getting eliminated under the condition that $\epsilon(1 + c) < \epsilon \mathscr{R}_{00} < 1$ before the community accepts visitors. If the community accepts visitors with an immune proportion $\rho \ge \rho_{\infty}$, the disease keeps getting eliminated independently of how many visitors are accepted. If the community accepts visitors with an immune proportion $\rho < \rho_{\infty}$, then the acceptance of visitors so many as $\mu > \mu_c$ induces the endemicity, and the disease becomes endemic. Even if $\rho < \rho_{\infty}$, the acceptance of visitors so few as $\mu \le \mu_c$ does not induce the endemicity, and the disease keeps getting eliminated.

The situation considered in this theorem corresponds to the parameter region Ω_{-} in Figure 12, and is numerically exemplified by Figure 13(a). Theorem 3.9 indicates that, if the proportion of immune visitors is so low as $\rho < \rho_{\infty}$, there exists the upper threshold μ_{c} for the number of accepted visitors to suppress the revival of the disease spread after starting the acceptance of visitors in the community where the disease was getting eliminated.

Acceptance of only immune visitors

When the community accepts only immune visitors, that is, when $\mu > 0$ with $\rho = 1$, $G(\mu, 1)$ is necessarily greater than 1 and monotonically increasing in terms of μ . Then, from Theorem 3.8, we can find that only the acceptance of visitors so many as $\mu > \mu_{c1}$ can induce the disease-eliminated equilibrium in the community where the disease was endemic before starting the acceptance of visitors. As defined by (34), the critical value μ_{c1} depends on the risk of reinfection, and then, from Theorems 3.7 and 3.8, we can find the following result (see Figures 14 and 15):

Corollary 3.8.3. Suppose that the disease was endemic under the condition that $\epsilon \mathscr{R}_{00} > 1$ before the community accepts only immune visitors. If $\epsilon \mathscr{R}_{00} \ge 1 + c$, the endemicity remains independently of how many visitors the community accepts. If $1 < \epsilon \mathscr{R}_{00} < 1 + c$, the acceptance of visitors so many as $\mu > \mu_{c1}$ is effective to make the disease eliminated.

Therefore the community under an endemic situation could have a preferable influence to suppress the endemicity by accepting only immune visitors only when the reinfectivity is sufficiently low as indicated by Figure 15.

If the community was approaching the disease-eliminated equilibrium with the risk of reinfection so low as $\epsilon \mathscr{R}_{00} \leq 1$, the community keeps



Figure 16: Parameter region and boundary indicated by the condition the condition (27) with the results given by Theorems 3.4, 3.5, and 3.6 when all visitors accepted by the community is susceptible with $\rho = 0$: (a) $\Re_{00} > 1 + c$; (b) $\Re_{00} = 1 + c$; (c) $\Re_{00} < 1 + c$. Numerically drawn with (a) c = 1.0; (b) c = 3.0; (c) c = 5.0, and commonly $\Re_{00} = 4.0$.

approaching the disease-eliminated equilibrium even after starting the acceptance of only immune visitors, independently of how many visitors the community accepts.

Acceptance of only susceptible visitors

Now let us consider the case where all visitors accepted by the community are susceptible, that is, when $\mu > 0$ with $\rho = 0$. Then, from Theorem 3.9, we find three different cases according to the influence of the acceptance of visitors as shown in Figure 16, taking account of the features of G(μ , 0) given at the beginning of this section. For this case, we can get the following result:

Corollary 3.9.1. When the disease was getting eliminated under the condition that $\epsilon \mathscr{R}_{00} \leq 1$, the acceptance of only susceptible visitors induces

 $\begin{cases} \text{no endemicity if } \varepsilon \mathscr{R}_{00} \leqslant \varepsilon (1+c); \\ \text{no endemicity if } \varepsilon \mathscr{R}_{00} > \varepsilon (1+c) \text{ and } \mu \leqslant \mu_{c0}; \\ \text{the endemicity if } \varepsilon \mathscr{R}_{00} > \varepsilon (1+c) \text{ and } \mu > \mu_{c0}. \end{cases}$

In contrast, when the disease was endemic under the condition that $\epsilon \mathscr{R}_{00} > 1$, the acceptance of only susceptible visitors induces

(no change in the endemicity if $\epsilon \mathscr{R}_{00} \ge \epsilon(1+c)$; no change in the endemicity if $\epsilon \mathscr{R}_{00} < \epsilon(1+c)$ and $\mu \le \mu_{c0}$; the elimination of disease if $\epsilon \mathscr{R}_{00} < \epsilon(1+c)$ and $\mu > \mu_{c0}$.

The critical value μ_{c0} is defined by (35).

Therefore the acceptance of only susceptible visitors could have the counter effect according to the endemicity, depending on the infectivity of disease. Only for a moderately high infectious disease such that $1/\epsilon < \Re_{00} < 1 + c$, the acceptance of only susceptible visitors so many as $\mu > \mu_{c0}$ can lead the community to the disease-eliminated equilibrium. For the disease with a low reinfectivity such that $1 + c < \Re_{00} \leq 1/\epsilon$, the acceptance of only susceptible visitors so many as $\mu > \mu_{c0}$ can lead the community to the disease-eliminated equilibrium.

Further, as indicated by Figure 16, we find that there is a sufficient value of μ which determines the epidemic situation in the community:



Figure 17: The μ -dependence of endemic sizes. Numerically drawn by (28) with (a) (ϵ , ρ) = (0.2, 0.4) ($\epsilon \mathscr{R}_{00} = 0.8$, $\mu_c = 0.56$); (b) (ϵ , ρ) = (0.25, 0.4) ($\epsilon \mathscr{R}_{00} = 1.0$, $\mu_c = 0.0$, $\rho_c = 0.67$), (c) (ϵ , ρ) = (0.3, 0.1) ($\epsilon \mathscr{R}_{00} = 1.2$, $\mu_c < 0$, $\rho_c = 0.40$), (d) (ϵ , ρ) = (0.3, 0.8) ($\epsilon \mathscr{R}_{00} = 1.2$, $\mu_c = 1.67$, $\rho_c = 0.40$), and commonly $\mathscr{R}_{00} = 4.0$; c = 1.0.

Corollary 3.9.2. If $\mathscr{R}_{00} > 1 + c$, the acceptance of only susceptible visitors so many as

$$\mu \geqslant \frac{1+c}{\mathscr{R}_{00}-(1+c)}$$

necessarily makes the disease endemic. In contrast, if $\Re_{00} < 1 + c$, the acceptance of only susceptible visitors so many as

$$\mu \ge \frac{(\mathscr{R}_{00} - 1)(1 + c)}{(1 + c) - \mathscr{R}_{00}}$$

necessarily makes the disease eliminated.

The former case means an unpreferable influence of the sufficiently large number of visitors for the community with the spread of a highly infectious disease, while the latter does a preferable influence for the community with the spread of a moderately infectious disease.

Change in endemic size

Figure 17 shows the numerically drawn μ -dependence of endemic sizes y_r^* and y_v^* at the endemic equilibrium given by (28). As the figure implies, the endemic size necessarily has a monotonic dependence on the number of accepted visitors, represented now by μ , about which we can get the following analytical result (Appendix B.7):

Theorem 3.10. The endemic sizes y_r^* , y_v^* , and the total endemic size

$$z^* := \frac{y_r^* + \mu y_v^*}{1 + \mu} = \frac{I_r^* + I_v^*}{N + m}$$
(36)

are monotonically increasing in terms of μ if and only if $\epsilon \mathscr{R}_{00} \leq 1$ or

$$\begin{cases} \epsilon \mathscr{R}_{00} > 1; \\ \rho < \rho_{c} := \frac{1 - \epsilon^{2} \mathscr{R}_{00} - \epsilon c}{(1 - \epsilon) \epsilon \mathscr{R}_{00}}. \end{cases}$$
(37)



Figure 18: (ρ, μ) -dependence of the endemic size y_r^* . Numerically drawn contour maps for three cases correspond to those in Figure 14: (a) $\epsilon \mathscr{R}_{00} = 0.8$; (b) $\epsilon \mathscr{R}_{00} = 1.2$ and $\rho_c = 0.40$; (c) $\epsilon \mathscr{R}_{00} = 1.44$ and $\rho_c = -3.31$, where the parameter values are respectively the same as in Figure 14.

It can be easily found that $\rho_c < 1$ when $\varepsilon \mathscr{R}_{00} > 1$. See the numerically drawn (ρ, μ) -dependence of the endemic size y_r^* in Figure 18.

We remark that, as shown in this section, the endemic equilibrium for $\rho = 1$ exists only when $\epsilon \mathscr{R}_{00} > 1$. Then we can find the following result too (Appendix B.8):

Corollary 3.10.1. When the community accepts only immune visitors (i.e., $\rho = 1$), the endemic size is monotonically decreasing in terms of μ .

This result could be regarded as included in Theorem 3.10 because any condition given in Theorem 3.10 cannot be applicable when $\rho = 1$. We can see the numerical examples in Figure 18(b, c).

When ρ_c defined by (37) is non-positive with $\epsilon \mathscr{R}_{00} > 1$, any ρ cannot be smaller than ρ_c , so that the endemic size is necessarily monotonically decreasing in terms of μ :

Corollary 3.10.2. If $\epsilon \mathscr{R}_{00} > \max[1, 1/\epsilon - c]$, the endemic size is necessarily monotonically decreasing in terms of μ .

The numerical example of Figure 18(c) illustrates the case.

For the critical case of $\rho = \rho_c > 0$ with $\epsilon \mathscr{R}_{00} > 1$, we can derive the explicit values at the endemic equilibrium E_{00} from (28) (Appendix B.7):

$$x_{v}^{*} = \frac{c}{(1-\epsilon)\mathscr{R}_{00}}; \ y_{v}^{*} = y_{r}^{*} = z^{*} = 1 - \frac{1}{\epsilon\mathscr{R}_{00}}.$$
 (38)

Hence the endemic sizes are independent of the number of accepted visitors in this case:

Corollary 3.10.3. For $\rho = \rho_c > 0$ with $\epsilon \mathscr{R}_{00} > 1$, the endemic sizes y_r^* , y_v^* , and the total endemic size z^* are determined independently of μ .

A numerical example is given in Figure 18(b). We note it necessary for $\rho_c > 0$ with $\epsilon \mathscr{R}_{00} > 1$ that $1/\epsilon - c > 1$, that is, $\epsilon(1 + c) < 1$. Moreover the case of Corollary 3.10.3 can appear only when $1 < \epsilon \mathscr{R}_{00} < 1/\epsilon - c$.

Additionally we can find the following relations among the specific values ρ_s , ρ_{∞} , and ρ_c for the immune proportion of accepted visitors at the entry:

Corollary 3.10.4. It holds that $\rho_c < \rho_\infty$ and $\rho_c < \rho_s$.

The proof is easy by calculating the differences $\rho_{\infty} - \rho_c$ and $\rho_s - \rho_c$ and showing them positive. Numerical calculation of Figure 18(b) demonstrates this result.



Figure 19: Classification of the parameter region of (ϵ, ρ) according to the μ dependence of the change in the endemic size. Numerically drawn with $\Re_{00} = 4.0$ and c = 1.0. Regions Ω_{\pm} correspond to those in Figure 12.

Consequently, as indicated by Figure 19, the larger number of accepted visitors makes the endemic size bigger only when the immune proportion of accepted visitors at the entry is sufficiently small under the epidemic situation with a sufficiently low risk of infection.

3.4 DISCUSSION

The results of our model imply that the acceptance of temporal visitors from the outside may induce a significant change of the epidemic state in the community. Contrary to an intuitive expectation, the acceptance of visitors does not necessarily make the epidemic situation worse in the community. Only when the reinfectivity of the disease is sufficiently weak, the acceptance of visitors may induce the endemicity if the community accepts the visitors only with a sufficiently low immune proportion. Furthermore, when the reinfectivity is high, the acceptance of a sufficiently large number of visitors may induce the elimination of the disease if the community can regulate to accept the visitors with a sufficiently high immune proportion.

The visitors certainly play a role of recruitment of hosts for the disease spread in the community. The visitors with a higher susceptible proportion could be regarded as a larger supply of highly susceptible individuals in the community, and they provide a fast recruitment of new infectives. In contrast, the visitors with a high immune proportion cause only a slow recruitment of new infectives with the reinfection. For these reasons, the influence of the visitor acceptance on the epidemic dynamics with a reinfectious disease must depend on the immune proportion in the visitors at the entry. On the other hand, the entry of many visitors could induce a dilution of the infective density in the community at the same time, which is regarded as an advantageous influence of the visitor acceptance against the disease spread. In the epidemic dynamics with our model, a balance of these counteractive factors of the visitor acceptance with respect to the disease spread could significantly affect the consequence of epidemic dynamics in the community.

As a result, a preferable acceptance of visitors must be regulated to have a sufficiently large immune proportion according to the public health in the community. In this sense, the best policy for the visitor acceptance would be to allow the entry only for the immune visitors. From the results on our model, such an acceptance of only immune visitors may lower the endemic size, and further suppress the endemicity to induce the elimination of the disease spread in the community.

In contrast, when the community was on the way to the disease-eliminated equilibrium before starting the acceptance of visitors, the acceptance of visitors without any epidemiological regulation may cause the revival of the disease spread in the community. Such a case would occur by reduced cautiousness of the disease before starting the acceptance of visitors, which is caused by the reason that the number of infective residents became rather small in comparison to that at the outbreak.

Our model could be regarded as a consideration on the epidemic dynamics in a season. In this sense, the number of visitors may be beyond the number of residents in the community (i.e., $\mu > 1$), as some popular touristic local places like Venice in the vacation season, or a certain place attracting visitors like a newly found gold mine. As another example, we could consider a community accepting many evacuees from a certain calamity. Even though the number of visitors would be smaller than the number of residents in most cases (i.e., $\mu < 1$), our results imply that the influence of the visitor acceptance could depend on the infectivity and reinfectivity in the epidemic dynamics, and the regulation on the epidemiological nature of accepted visitors.

Since the infectivity and reinfectivity are not only determined by the nature of disease itself but also by social custom, the sanitary condition, and the people's behavior^[57–59,144–148], the influence of the visitor acceptance could depend also on social factors in the community which accepts the visitors under the epidemic dynamics. Such social factors could be affected by the situation of disease spread during the epidemic dynamics in the community. For example, some strategic/non-strategic transmission of information about the disease spread or a public health campaign to prevent the further disease spread could alter people's social behavior, and subsequently the risk of infection/reinfection. Hence if the infectivity and reinfectivity would be changeable in the epidemic dynamics, the influence of the visitor acceptance would be qualitatively changed.

The results from our model imply such a possibility that a shift of the infectivity and/or reinfectivity to the weaker would induce an epidemic situation in which the acceptance of visitors causes the increase in the epidemic size or the revival of disease spread even with the endemicity. If so, there would be repetitive revivals of disease spread in the community, driven by a temporal shift of the infectivity and/or reinfectivity which could bring a feedback influence on the policy to control the disease spread in the community. Such theoretical/mathematical researches on the relation between the disease spread and the nature of hosts are interesting and require further development.

4.1 ASSUMPTIONS AND MODELING

4.1.1 Assumptions

The strains from the same species or different species interact with competition. They compete not only for the host population by infecting as many healthy individuals as possible, but also for the growth-limiting resources within the host individual^[149–151]. For example, they compete for the healthy cells, which could be one of the identical resources within the host individual, and only one strain can persist. We consider the superinfection as one of the results of strains competition within the host individual. For our mathematical modeling, we set up the following assumptions:

- Strains compete for the host population. We assume that the infection force of infectives with strain k for the healthy individuals is given by $\beta_k I_k$.
- Strains compete for the resources within a host following a strict hierarchy order of competitive dominance.
- Superinfection occurs when a more dominant strain j takes over a host infected by a less dominant strain k. For the infectives who hold strain k, we assume that the infection force of infectives I_j with strain j is given by ε_{jk}β_jI_j under the possibility ε_{jk}.
- The quarantine efficiency is determined by the detectability of strain.
- The recovered individual gets immunity lasting in the epidemic season under consideration.
- The influence of disease in demographic change is negligible in the time scale of considered epidemic dynamics.

Although there are some studies have shown that the more virulent strains of rodent malaria, the more competitive advantage within the hosts^[152], there is no necessary relationship between the competitive dominance, the transmissibility, and the virulence^[153,154]. We shall focus on the distribution of detectability and consider that the transmissibility and the possibility of superinfection are independent of strains (i.e., $b_k = b$ and $\varepsilon_{jk} = \varepsilon$ for all k and j < k).



Figure 20: The state transition in the epidemic dynamics of our model. S, I_k , Q_k , and R (k = 1, 2, ..., n) are population densities of susceptibles, infectives who hold strain k as the strain of the highest dominance, corresponding isolated and recovered individuals respectively, where $1 \le i < j < k$.

4.1.2 Modeling

With the assumptions in Section 4.1.1, our model is given by the following system of ordinary differential equations (Figure 20):

$$\begin{array}{ll} \frac{dS}{dt} &= \mu N - \sum\limits_{k=1}^{n} \beta I_k S - \mu S; \\ \frac{dI_1}{dt} &= \beta I_1 S + \sum\limits_{k=2}^{n} \varepsilon \beta I_1 I_k - \sigma_1 I_1 - \rho_1 I_1 - \mu I_1; \\ \frac{dI_j}{dt} &= \beta I_j S + \sum\limits_{k=j+1}^{n} \varepsilon \beta I_k I_j - \sum\limits_{k=1}^{j-1} \varepsilon \beta I_j I_k \\ &- \sigma_j I_j - \rho_j I_j - \mu I_j \quad (1 < j < n); \\ \frac{dI_n}{dt} &= \beta I_n S - \sum\limits_{k=1}^{n-1} \varepsilon \beta I_k I_n - \sigma_n I_n - \rho_n I_n - \mu I_n; \\ \frac{dQ_k}{dt} &= \sigma_k I_k - \alpha_k Q_k - \mu Q_k \quad (k = 1, 2, \dots, n); \\ \frac{dR}{dt} &= \sum\limits_{k=1}^{n} \rho_k I_k + \sum\limits_{k=1}^{n} \alpha_k Q_k - \mu R, \end{array}$$

where S, I_k, Q_k, and R are population densities of susceptibles, infectives who hold strain k, corresponding isolated and recovered individuals respectively, The total population size is denoted by N = S + $\sum_{k=1}^{n} I_k + \sum_{k=1}^{n} Q_k + R$. βI_k gives the infection force of strain k for the susceptible with the coefficient β , and $\epsilon\beta I_j$ gives that for the infective with strain j of the lower dominance. Parameter σ_k is the quarantine rate for the infective who holds strain k, which reflects the detectability of strain k. Parameters α_k and ρ_k are the recovery rates for the infective with strain k under and out of the isolation, respectively. μ is the natural death rate.

4.2 BASIC REPRODUCTION NUMBER

To derive the basic reproduction number for the model (39), we use the next-generation method. The model (39) has the disease-free equilibrium

 $E_0 = (N, 0, ..., 0)$. We can derive the next-generation matrix $G = FV^{-1}$, where F and G are diagonal matrices given by

$$\begin{split} F &= diag(\beta N, \beta N, \cdots, \beta N); \\ V &= diag(\sigma_1 + \rho_1 + \mu, \sigma_2 + \rho_2 + \mu, \cdots, \sigma_n + \rho_n + \mu) \end{split}$$

Thus we have

 $FV^{-1} = diag(\mathscr{R}_{0,1}, \mathscr{R}_{0,2}, \cdots, \mathscr{R}_{0,n}),$

where

$$\mathscr{R}_{0,k} := \frac{\beta N}{\sigma_k + \rho_k + \mu} \quad (k = 1, 2, \cdots, n).$$
(40)

The basic reproduction number for the system (39) mathematically given by

$$\mathscr{R}_0 = \max\{\mathscr{R}_{0,1}, \mathscr{R}_{0,2}, \cdots, \mathscr{R}_{0,n}\}.$$

The basic reproduction number \mathscr{R}_0 for the system (39) is given by the maximum of the expected number of new cases. This \mathscr{R}_0 is meaningful only if we consider all strains are included in the initial stage with $0 < I_k(0) \ll 1$ (k = 1, 2, ..., n). The number of new cases infected by per infective individual is defined for each strain, which is given by $\mathscr{R}_{0,k}$ (k = 1, 2, ..., n). At the initial stage, if the initial invasion includes only strain k, we have $dI_k(t)/dt > 0$ with $I_k(0) > 0$, $I_k(0) \ll 1$, and $I_j(0) = 0$ (j \neq k). In such a case, we have the basic reproduction number given by $\mathscr{R}_{0,k}$. Let us call $\mathscr{R}_{0,k}$ the *strain-specific basic reproduction number*.

For the mathematical convenience, we define a set $\Omega := \{k \mid \mathscr{R}_{0,k} > 1\}$, which represents the strain with $\mathscr{R}_{0,k} > 1$. The set Ω is empty when $\mathscr{R}_0 \leq 1$.

4.3 MATHEMATICAL RESULTS ON THE MODEL

4.3.1 Non-dimensionalization

We apply the following parameter transformation to non-dimensionalize the model (39) ($k = 1, 2, \dots, n$):

$$\begin{aligned} \tau &:= \mu t; \ u := \frac{S}{N}; \ \nu_{k} := \frac{I_{k}}{N}; \ q_{k} := \frac{Q_{k}}{N}; \ w := \frac{R}{N}; \\ b &:= \frac{\beta N}{\mu}; \ \gamma_{k} := \frac{\sigma_{k}}{\mu}; \ a_{k} := \frac{\alpha_{k}}{\mu}; \ \eta_{k} := \frac{\rho_{k}}{\mu}, \end{aligned}$$
(41)

then our model can be rewritten as:

$$\begin{split} \frac{du}{d\tau} &= 1 - b \Big(\sum_{k=1}^{n} \nu_k \Big) u - u; \\ \frac{dv_1}{d\tau} &= b \nu_1 u + \varepsilon b \Big(\sum_{k=2}^{n} \nu_k \Big) \nu_1 - (1 + \gamma_1 + \eta_1) \nu_1; \\ \frac{dv_j}{d\tau} &= b \nu_j u + \varepsilon b \Big(\sum_{k=j+1}^{n} \nu_k \Big) \nu_j - \varepsilon b \Big(\sum_{k=1}^{j-1} \nu_k \Big) \nu_j \\ &- (1 + \gamma_j + \eta_j) \nu_j \quad (1 < j < n); \\ \frac{dv_n}{d\tau} &= b \nu_n u - \varepsilon b \Big(\sum_{k=1}^{n-1} \nu_k \Big) \nu_n - (1 + \gamma_n + \eta_n) \nu_n; \\ \frac{dq_k}{d\tau} &= \gamma_k \nu_k - a_k q_k - q_k \quad (k = 1, 2, \dots, n); \\ \frac{dw}{d\tau} &= \sum_{k=1}^{n} \eta_k \nu_k + \sum_{k=1}^{n} a_k q_k - w \end{split}$$

with the initial condition that $u(0) = u^0 > 0$, $v_1(0) = v_1^0 > 0$, $v_j(0) = v_j^0 > 0$, $v_n(0) = v_n^0 > 0$, $q_k(0) = 0$, w(0) = 0 (1 < j < n), and $u^0 + \sum_{k=1}^n v_k^0 = 1$. From (40), the strain-specific reproduction number is given by

$$\mathscr{R}_{0,k} = \frac{b}{1 + \gamma_k + \eta_k}$$
 (k = 1, 2, ..., n). (43)

Hereafter, we assume that $\Re_{0,j} \neq \Re_{0,k}$ for $j \neq k$.

4.3.2 Model without superinfection

In this section, we consider that superinfection never happens ($\epsilon = 0$). This is the case when there is no such order of competition dominance among strains. Once a host is infected with a strain, such strain is considered as completely owning that host. From (41), model (39) without superinfection can be nondimensionalized as:

$$\begin{aligned} \frac{du}{d\tau} &= 1 - b \left(\sum_{k=1}^{n} v_k \right) u - u; \\ \frac{dv_k}{d\tau} &= b v_k u - (1 + \gamma_k + \eta_k) v_k \quad (k = 1, 2, \dots, n); \\ \frac{dq_k}{d\tau} &= \gamma_k v_k - a_k q_k - q_k; \\ \frac{dw}{d\tau} &= \sum_{k=1}^{n} \eta_k v_k + \sum_{k=1}^{n} a_k q_k - w. \end{aligned}$$

$$(44)$$

We can obtain the following result on the stability of the equilibrium for the model without superinfection (Appendix C.1):

Theorem 4.1. For the model without superinfection ($\epsilon = 0$), that is, for the model (44),

- (*i*) if Ω is empty, E_0 is is globally asymptotically stable. Otherwise, E_0 is unstable;
- (ii) if Ω is not empty, we have a unique locally asymptotically stable single strain endemic equilibrium E_{ℓ} given by

$$u^{*} = \frac{1}{\mathscr{R}_{0,\ell}}; v_{\ell}^{*} = \frac{\mathscr{R}_{0,\ell} - 1}{b}; v_{k}^{*} = 0; q_{\ell}^{*} = \frac{\gamma_{\ell}v_{\ell}^{*}}{a_{\ell} + 1}; q_{k}^{*} = 0, \quad (45)$$

where ℓ such that $\mathscr{R}_{0,\ell} = \mathscr{R}_0$.

When Ω is empty, that is, when $\Re_{0,k} \leq 1$ for any k, the system (44) necessarily approaches the disease-free equilibrium. When Ω is not empty, that is, when there exists ℓ such that $\Re_{0,\ell} > 1$, if the system (44) approaches the endemic equilibrium, it consists of the infectives with the strain which has the largest strain-specific basic reproduction number. Therefore, If and only if Ω is not empty, when there is a single strain ℓ which has the basic reproduction number $\Re_{0,\ell} > 1$ or when all strains with the basic reproduction number greater than one have the same basic reproduction number, there is a single endemic equilibrium.

On the other hand, if we consider that there exist some j such that $\Re_{0,j} = \Re_{0,k} > 1$ (j $\neq k$), we have an endemic equilibrium consisting of both infectives with strain j and infectives with strain k, while v_j^* and v_k^* are alternatively positive or zero at the equilibrium.

From (45), we have the equilibrium endemic size z^* at E_{ℓ} is given by

$$z^{*} := v_{\ell}^{*} + q_{\ell}^{*} = \frac{1 + \gamma_{\ell}/(a_{\ell} + 1)}{1 + \gamma_{\ell} + \eta_{\ell}} \left(1 - \frac{1}{\mathscr{R}_{0,\ell}} \right)$$
$$= \frac{1 + \gamma_{\ell}/(a_{\ell} + 1)}{1 + \gamma_{\ell} + \eta_{\ell}} \left(1 - \frac{1 + \gamma_{\ell} + \eta_{\ell}}{b} \right).$$
(46)

It is easy to see that the endemic size z^* at E_{ℓ} is monotonically decreasing in terms of $1/\Re_{0,\ell}$. We can also have the following results about the γ_{ℓ} dependence on the equilibrium endemic size z^* at E_{ℓ} (Appendix C.2):

Theorem 4.2. For the model without superinfection ($\epsilon = 0$), that is, for the model (44) with $\Re_{0,\ell} = \Re_0 > 1$,

- (*i*) if $b(\eta_{\ell} \alpha_{\ell}) \leq (1 + \eta_{\ell})^2$, the equilibrium endemic size z^* at E_{ℓ} is monotonically decreasing in terms of γ_{ℓ} ;
- (ii) if $b(\eta_{\ell} a_{\ell}) > (1 + \eta_{\ell})^2$, the equilibrium endemic size z^* at E_{ℓ} takes a maximum value for a specific value of γ_{ℓ} which is given by $-(1 + \eta_{\ell}) + \sqrt{b(\eta_{\ell} a_{\ell})}$.

Moreover, the results shown in Theorems 4.1 and 4.2 hold even if the transmissibility depends on strains (i.e., $b_j \neq b_k$ for $j \neq k$).

4.3.3 Model with superinfection

Boundedness for the solution

Since the first (n + 1) equations in the model (42) is closed, it is sufficient to consider the stability with the reduced closed system of model (42). We set the following mathematical results on the boundedness for the solution of the reduced system of model (42) (Appendix C.3):

Lemma 4.1. For the initial condition such that $u(0) = u^0 > 0$, $v_1(0) = v_1^0 > 0$, $v_j(0) = v_j^0 > 0$ (j = 2, 3, ..., n-1), $v_n(0) = v_n^0 > 0$, and $u(0) + \sum_{k=1}^n v_k(0) = u^0 + \sum_{k=1}^n v_k^0 = 1$, the solution of the reduced closed system of model (42) belongs to the region D for any $\tau > 0$, where

$$D := \left\{ (u, v_1, \dots, v_n) \, \big| \, u > 0, v_k > 0 \, (k = 1, 2, \dots, n), u + \sum_{k=1}^n v_k < 1 \right\}.$$
(47)

Feasible equilibrium

First we can get the following result on the reduced closed system of model (42) (Appendix C.4):

Lemma 4.2. If $\mathscr{R}_{0,k} \leq 1$, the strain necessarily die out as $\tau \to \infty$.

If the strain with the strain-specific reproduction number less than or equal to 1, such a strain can not persist in the population. On the other hand, it is necessary to have the strain-specific reproduction number greater than 1 to persist such a strain. For model (42), we have the disease-free equilibrium E_0 always exists. Making use of the eigenvalue analysis, we can easily get the following result about the local stability of E_0 (Appendix C.5):

Lemma 4.3. The disease-free equilibrium E_0 is locally asymptotically stable if Ω is empty. If Ω is not empty, it is unstable.

Making use of the result in Lemmas 4.1, 4.2, and 4.3, we can get the following result about the stability of E_0 :

Theorem 4.3. *The disease-free equilibrium* E_0 *is globally asymptotically stable in* D *if* Ω *is empty, where* D *is given in* (47)*. Otherwise, if* Ω *is not empty, it is unstable.*

When Ω is not empty, that is, when there are some strain k such that $\Re_{0,k} > 1$, we have some endemic equilibria where at least one strain persists. If there exists a strain ℓ such that $\Re_{0,\ell} > 1$, we have a single strain endemic equilibrium E_{ℓ} given by

$$u^{*} = \frac{1}{\mathscr{R}_{0,\ell}}; \ v_{\ell}^{*} = \frac{\mathscr{R}_{0,\ell} - 1}{b_{\ell}}; \ v_{k}^{*} = 0; \ q_{\ell}^{*} = \frac{\gamma_{\ell}v_{\ell}^{*}}{a_{\ell} + 1}; \ q_{k}^{*} = 0; \ w^{*} = \eta_{\ell}v_{\ell}^{*} + a_{\ell}q_{\ell}^{*},$$
(48)

where $k = 1, 2, \ldots, n$ and $k \neq \ell$.

Single strain endemic equilibrium

We can get the following result about the stability of E_{ℓ} given in (48) (Appendix C.6):

Theorem 4.4. When the equilibrium E_{ℓ} exists with $\mathscr{R}_{0,\ell} > 1$ and satisfies

$$\min_{\mathbf{k}<\ell}\left\{\frac{1}{\mathscr{R}_{0,\mathbf{k}}}\right\} > \frac{1}{\mathscr{R}_{0,\ell}} + \frac{\varepsilon}{b}(\mathscr{R}_{0,\ell}-1);$$
(49)

$$\min_{k>\ell}\left\{\frac{1}{\mathscr{R}_{0,k}}\right\} > \frac{1}{\mathscr{R}_{0,\ell}} - \frac{\varepsilon}{b}(\mathscr{R}_{0,\ell} - 1), \tag{50}$$

it is globally asymptotically stable in D *given in* (47). *Otherwise, if the condition* (49) or (50) *is unsatisfied,* E_{ℓ} *is unstable even when it exists.*

When $\mathscr{R}_{0,k} > 1$ for all k = 1, 2, ..., n, if there exists a globally asymptotically stable single strain endemic equilibrium E_{ℓ} , all the other single strain endemic equilibria E_k ($k \neq \ell$) are unstable even they exist.

From Theorem 4.4, we have the following results:

Corollary 4.4.1. For the strain which has $\mathscr{R}_{0,j} = \min\{\mathscr{R}_{0,k}\}$ with $\mathscr{R}_{0,j} > 1$ (k = 1, 2, ..., n), the single strain endemic equilibrium E_j is necessarily unstable.

From Theorems 4.3 and 4.4, when there is more than one strain with strain-specific reproduction number greater than 1, if the system does not approach a single strain endemic equilibrium, it will approach an endemic state persisting more than one strains.

4.3.4 Dependence on the distribution of strain-specific basic reproduction number

From Corollary 4.2, we know that a strain k may persist only if it has the strain-specific reproduction number $\Re_{0,k} > 1$. In this section, we assume that $\Re_{0,k} > 1$ for any k = 1, 2, ..., n.

Monotonically decreasing

If the strain-specific basic reproduction number is monotonically decreasing following the strain's order of competitive dominance (i.e., $\mathcal{R}_{0,k+1} < \mathcal{R}_k$ for all k), we have the single strain endemic equilibrium E_1 is necessarily globally asymptotically stable since the condition (50) with $\ell = 1$ is always satisfied. From Corollary 4.4.1, we have E_k are unstable for all k > 1. Therefore, we can get the following result:



Figure 21: Temporal variations for n strain model with n = 6. Numerically drawn with (a) γ_1 = 1.5; (b) γ_1 = 3.0; (c) γ_1 = 4.5. Commonly, b = 10.0; γ_k = 0.8; η_k = 1.0 (k = 2, 3, ..., 6); ϵ = 0.6; (u(0), v_1(0), v_2(0), v_3(0), v_4(0), v_5(0), v_6(0)) = (0.94, 0.01, 0.01, 0.01, 0.01, 0.01).

Theorem 4.5. If $\mathscr{R}_{0,k+1} > \mathscr{R}_k$ with $\mathscr{R}_{0,k} > 1$ for all k = 1, 2, ..., n, the system (42) necessarily approaches the single strain endemic equilibrium E_1 .

If the more dominant strain has a lower strain-specific basic reproduction number, only the most dominant strain (i.e., strain 1) can persist while any other subordinate strains disappear. In this case, it is impossible to have more than one strains at the endemic state.

Monotonically increasing

If the strain-specific basic reproduction number is monotonically decreasing following the strain's order of competitive dominance (i.e., $\mathscr{R}_{0,k+1} > \mathscr{R}_k$ for all k), we have the system (42) may approach a single strain endemic equilibrium, depending on the conditions (49) and (50). When the condition (49) or (49) is unsatisfied for any $\ell = 1, 2, ..., n$, we have the system (42) approaches an endemic state where more than one strains persist.

Let us assume that $\gamma_1 > 0$, $\gamma_k = 0$, $\eta_1 = \eta_k = \eta$ for $k \neq 1$, then we have $\mathscr{R}_{0,1} < \mathscr{R}_{0,k}$ for $k \neq 1$. We can obtain the following result:

Theorem 4.6. When $\gamma_1 > 0$, $\gamma_k = 0$, and $\eta_1 = \eta_k = \eta$ with $\mathscr{R}_{0,k} > 1$ for any $k \neq 1$, the system (42) approaches an endemic state where more than one strains persist if

$$\frac{1}{2}\bigg\{-(1+\eta+\varepsilon)+\sqrt{(1+\eta-\varepsilon)^2+4\varepsilon b}\bigg\}<\gamma_1<\varepsilon\bigg(\frac{b}{1+\eta}-1\bigg).$$

When the recovery rate is independent of strains, if the most dominant strain (i.e., strain 1) is the only detectable strain while any other subordinate strains are undetectable, the system may approach an endemic equilibrium at which the infectives holding different strains persist. If the detectability for the infection with strain 1 is sufficiently low, the system approaches the endemic equilibrium E_1 (see Figure 21(a)). On the other hand, if the detectability for the infection with strain 1 is sufficiently high, the system approaches the endemic equilibrium E_2 (see Figure 21(c)). Only if the detectability for the infection with strain 1 is in an intermediate range, the system approaches an endemic equilibrium where strain 1 and strain 2 persist, while the other strains get eliminated (see Figure 21(b)).

If $\gamma_j > 0$ and $\gamma_k = 0$ for $j \neq 1$ and $k \neq j$, we have the endemic equilibrium E_1 is necessarily locally asymptotically stable, while other endemic equilibria E_k ($k \neq 1$) are unstable. If there is the only detectable strain j which is not the

most dominant strain ($j \neq 1$), while other strains k ($k \neq j$) are undetectable, the system approaches an endemic equilibrium where there are only the infectives with strain 1, and no infectives with strain k ($k \neq 1$) exists.

4.3.5 Two strain model

In this section, we shall consider the model (42) of two strains (n = 2):

$$\frac{du}{d\tau} = 1 - bv_1 u - bv_2 u - u;$$

$$\frac{dv_1}{d\tau} = bv_1 u + \varepsilon bv_1 v_2 - (1 + \gamma_1 + \eta_1)v_1;$$

$$\frac{dv_2}{d\tau} = bv_2 u - \varepsilon bv_1 v_2 - (1 + \gamma_2 + \eta_2)v_2;$$

$$\frac{dq_1}{d\tau} = \gamma_1 v_1 - a_1 q_1 - q_1;$$

$$\frac{dq_2}{d\tau} = \gamma_2 v_2 - a_2 q_2 - q_2;$$

$$\frac{dw}{d\tau} = \eta_1 v_1 + \eta_2 v_2 + a_1 q_1 + a_2 q_2 - w,$$
(51)

with $u + v_1 + v_2 + q_1 + q_2 + w = 1$ and the initial condition that

$$(\mathfrak{u}(0), \mathfrak{v}_1(0), \mathfrak{v}_2(0), \mathfrak{q}_1(0), \mathfrak{q}_2(0), \mathfrak{w}(0)) = (\mathfrak{u}^0, \mathfrak{v}_1^0, \mathfrak{v}_1^0, \mathfrak{0}, \mathfrak{0}, \mathfrak{0})$$

where $u^0 > 0$, $v_1^0 > 0$, $v_2^0 > 0$, and $u^0 + v_1^0 + v_2^0 = 1$. We suppose that $\Re_{0,1} \neq \Re_{0,2}$, where $\Re_{0,1}$ and $\Re_{0,2}$ are defined by (43). The system (51) has the disease-free equilibrium $E_0 = (1, 0, 0, 0, 0, 0)$ and may have the endemic equilibria $E_1 = (u^*, v_1^*, 0, q_1^*, 0, w^*)$, $E_2 = (u^*, 0, v_2^*, 0, q_2^*, w^*)$, and $E_{12} = (u^*, v_1^*, v_2^*, q_1^*, q_2^*, w^*)$ where $u^* > 0$, $v_1^* > 0$, $v_2^* > 0$, $q_1^* > 0$, $q_2^* > 0$, and $w^* > 0$.

In the absent of superinfection (i.e., $\epsilon = 0$), from the results derived in Section 4.3.2, we know that, if and only if $\Re_{0,1} \leq 1$ and $\Re_{0,2} \leq 1$, the system (51) necessarily approaches the disease-free equilibrium E_0 . If $\Re_{0,1} > 1$ and $\Re_{0,2} \leq 1$, the system (51) approaches the single strain endemic equilibrium E_1 . If $\Re_{0,1} \leq 1$ and $\Re_{0,2} > 1$, it approaches the single strain endemic equilibrium E_2 . If $\Re_{0,1} > 1$ and $\Re_{0,2} > 1$, the system (51) approaches E_1 when $\Re_{0,1} > \Re_{0,2}$, while it approaches E_2 when $\Re_{0,2} > \Re_{0,1}$.

With the presence of superinfection where $0 < \epsilon < 1$, we can get the following result on the existence of the equilibrium for the system (51) (Appendix C.7):

Lemma 4.4. For the system (51),

- (*i*) the disease-free equilibrium E_0 always exists;
- (ii) the endemic equilibrium E_1 exists if $\mathscr{R}_{0,1} > 1$, where

$$u^* = \frac{1}{\mathscr{R}_{0,1}}; v_1^* = \frac{\mathscr{R}_{0,1} - 1}{b}; v_2^* = 0; q_1^* = \frac{\gamma_1 v_1^*}{a_1 + 1}; q_2^* = 0; w^* = \eta_1 v_1^* + a_1 q_1^*$$

(iii) the endemic equilibrium E_2 exists if $\Re_{0,2} > 1$, where

$$u^{*} = \frac{1}{\mathscr{R}_{0,2}}; v_{1}^{*} = 0; v_{2}^{*} = \frac{\mathscr{R}_{0,2} - 1}{b}; q_{1}^{*} = 0; q_{2}^{*} = \frac{\gamma_{2}v_{2}^{*}}{a_{2} + 1}; w^{*} = \eta_{2}v_{2}^{*} + a_{2}q_{2}^{*};$$



Figure 22: Temporal variations for the two strain model. Numerically drawn with (a) $\gamma_2 = 0.9$; (b) $\gamma_2 = 0.5$; (c) $\gamma_2 = 0.2$. Commonly, $\gamma_1 = 0.8$; $\eta_1 = 0.6$; $a_1 = 2.0$; $\eta_2 = 0.6$; $a_2 = 3.0$; b = 3.0; $\varepsilon = 0.8$; $(u(0), v_1(0), v_2(0), q_1(0), q_2(0), w(0)) = (0.9, 0.05, 0.05, 0, 0, 0)$.

(iv) the endemic equilibrium E_{12} exists if and only if $\Re_{0,2} > \Re_{0,1} > 1$ and

$$\frac{1}{1-\hat{\varepsilon}} \left(\frac{1}{\mathscr{R}_{0,1}} - \hat{\varepsilon} \right) < \frac{1}{\mathscr{R}_{0,2}}$$
(52)

$$\frac{1}{\mathscr{R}_{0,2}} < \frac{1}{(1-\hat{\varepsilon}) + \hat{\varepsilon}\mathscr{R}_{0,1}} \cdot \frac{1}{\mathscr{R}_{0,1}}$$
(53)

are satisfied, where

$$\hat{\varepsilon} := \frac{\varepsilon}{1 + \gamma_2 + \eta_2}.$$
(54)

;

and

$$u^{*} = \left\{ 1 + \frac{b}{\varepsilon} \left(\frac{1}{\mathscr{R}_{0,1}} - \frac{1}{\mathscr{R}_{0,2}} \right) \right\}^{-1}; v_{1}^{*} = \frac{1}{\varepsilon} \left(u^{*} - \frac{1}{\mathscr{R}_{0,2}} \right); v_{2}^{*} = \frac{1}{\varepsilon} \left(\frac{1}{\mathscr{R}_{0,1}} - u^{*} \right)$$
$$q_{1}^{*} = \frac{\gamma_{1}v_{1}^{*}}{a_{1} + 1}; q_{2}^{*} = \frac{\gamma_{2}v_{2}^{*}}{a_{2} + 1}; w^{*} = \eta_{1}v_{1}^{*} + \eta_{2}v_{2}^{*} + a_{1}q_{1}^{*} + a_{2}q_{2}^{*}.$$
(55)

If $\Re_{0,1} \leq 1$ and $\Re_{0,2} \leq 1$, there is not any equilibrium other than the disease-free equilibrium E_0 . If $\Re_{0,1} > 1$ or $\Re_{0,2} > 1$, we may have three endemic equilibria with the persistence of a single strain or the coexistence of two strains (Figure 22).

From Theorem 4.3, we have the disease-free equilibrium E_0 is globally asymptotically stable if and only if $\Re_{01} \leq 1$ and $\Re_{02} \leq 1$. We can also obtain the following result about the stability of E_1 , E_2 , and E_{12} for the two strain model (51) (Appendix C.8):

Theorem 4.7. For the two strain model (51),

- *(i)* E₁ *is globally asymptotically stable only if the condition given by the inverse inequality in (*53*) is satisfied;*
- *(ii)* E₂ *is globally asymptotically stable only if the condition given by the inverse inequality in (*52*) is satisfied;*
- (iii) E_{12} is always globally asymptotically stable whenever it exists.

From the results in Theorem 4.7, when E_1 or E_2 is globally asymptotically stable, we have E_{12} does not exists. When E_{12} exists, we have E_1 and E_2 are unstable. We can also get the following results from Theorem 4.7:





Corollary 4.7.1. When E_1 is globally asymptotically stable, E_2 is unstable and vice versa.

Corollary 4.7.2. When E_{12} exists, both of E_1 and E_2 exist, and they are unstable.

From Theorem 4.7 with (52) and (53), we can draw the (γ_1, γ_2) -dependence of the existence and stability of equilibria for the two strain model (Figure 23). As shown by Figure 23, strain 2 can persist only if strain 1 has a sufficiently high detectability. The harder the strain 2 is to detect, the more likely it can persist. Moreover, strain 1 and strain 2 can coexist only for certain intermediate ranges of γ_1 and γ_2 . If both strains have sufficiently high detectabilities, it is less likely to have both strains coexist.

4.4 DISCUSSION

In this work, we consider a mathematical model on the epidemic dynamics of a disease transmission with n strains which follows an order of the competitive dominance according to the infection success in the host. In our model, when no superinfection occurs, the disease becomes eliminated or alternatively the endemic state arises with only the strain which has the largest basic reproduction number while all the other strains get eliminated. For the persisting strain with a sufficiently low transmissibility, the detectability of the infection with such a strain can reduce the endemic size. On the other hand, for the persisting strain with a sufficiently high transmissibility, the endemic size takes maximum value for a specific value of the detectability of the infection with such a strain.

For the two strain model with superinfection, we show a possibility that the system goes to the endemic equilibrium where there are infectives with strain 1 and infectives strain 2. If the transmissibility of strain 1 (or strain 2) is sufficiently large, the system approaches the endemic equilibrium which consists of infectives with only strain 1 (or strain 2). Only if the transmissibilities of strain 1 and strain 2 are under an intermediate ranges with $\Re_{0,2} > \Re_{0,1}$, these two strain can co-exist at the endemic equilibrium, that is, there exist infectives with strain 1 and infectives with strain 2 at the endemic equilibrium.

For the n strain model with superinfection, we find it is possible that there are more than one strain persisting at the endemic equilibrium. The model with n strain which have the same transmissibility to the susceptible and the possibility of superinfection to the infectives with subordinate strain shows that, if there is no detectable strain other than the most dominate strain (i.e., strain 1) with $\Re_{0,1} > 1$, it is likely to persist more than one strain at the endemic equilibrium state. If the unique detectable strain is not the most dominate strain but a subordinate strain k (k > 1) with $\Re_{0,k} > 1$, the system approaches the endemic equilibrium where there are the infectives with only strain 1.

With the presence of superinfection, the subordinate may persist at the endemic equilibrium. If the system approaches the endemic equilibrium at which not only one strain persists, surprisingly, the endemic size is not necessarily reduced by the superinfection possibility. Depending on the characteristics of the strains, superinfection could either support or suppress the endemic size.

CONLUSION

The emergence and spread of infectious diseases have posed significant challenges to public health concerns, and the social nature including social response, social sensitivity, community's policy, and detectability of infection has been considered as playing a critical role in the disease transmission.

Social behavior changes triggered by risk perception such as wearing masks, taking medication, vaccination, and keeping social distance can further alter the evolution of an epidemic outbreak. Especially, the social response with sensitivity and insensitivity could cause the emergence of recurring epidemic outbreaks, and the strong social insensitivity would stabilize the temporal variation of the infective size, certainly raising the endemic size.

People's displacement due to social and political unrest as well as the natural migration of disease vectors to new areas on the epidemic outbreak could have a significant influence on the spread of infectious diseases. The acceptance of visitors under such an epidemic dynamics with the risk of reinfection could not only support the endemicity but also suppress it by either shifting the endemicity or reducing the endemic size, depending on the nature of infectious disease. It is therefore essential to develop effective measures to mitigate the spread of an infectious disease, taking into account the characteristics of such a disease.

Moreover, the emergence of mutant or novel strains of infectious diseases could provide a challenge to clinical diagnosis due to the lack of knowledge about the strains or the limitation of testing techniques. Such strains are clearly under the exploitative competition for the host and for the reproduction in the host. Even when an individual is infected by a hardly detectable novel or mutant strain, the superinfection of another detectable strain could serve the infected individual to be diagnosed and quarantined, which in turn may help to suppress the disease spread.

The spread of transmissible diseases is a major public health concern that can result in significant morbidity and mortality, as seen in recent pandemics like COVID-19. Understanding of the complex relation between social nature and disease transmission could enable policymakers and healthcare professionals to develop targeted interventions to mitigate the spread of transmissible diseases and minimize the occurrence of repeated epidemic outbreaks. Such a kind of theoretical/mathematical works on the relation of social nature to the epidemics would also be valuable for reducing the burden of infectious diseases on global health.

BIBLIOGRAPHY

- P.E.M. Fine, J.A. Clarkson, Measles in England and Wales—I: an analysis of factors underlying seasonal patterns, Int. J. Epidemiol. 11 (1) (1982) 5–14. https://doi.org/10.1093/ije/11.1.5
- [2] Z. Grossman, Oscillatory phenomena in a model of infectious diseases, Theor. Popul. Biol. 18 (2) (1980) 204–243. https://doi.org/10.1016/0040-5809(80)90050-7
- [3] A.U.M. Shah, S.N.A. Safri, R. Thevadas, N.K. Noordin, A. Abd Rahman, Z. Sekawi, A. Ideris, M.T.H. Sultan, COVID-19 outbreak in Malaysia: Actions taken by the Malaysian government, Int. J. Infect. Dis. 97 (2020) 108–116. https://doi.org/10.1016/j.ijid.2020.05.093
- [4] H.E. Soper, The Interpretation of Periodicity in Disease Prevalence, J. R. Stat. Soc. 92 (1) (1929) 34–73. https://doi.org/10.2307/2341437
- [5] W.P. London, J.A. Yorke, Recurrent outbreaks of measles, chickenpox and mumps: I. Seasonal variation in contact rates, Am. J. Epidemiol. 98 (6) (1973) 453-468. https://doi.org/10.1093/oxfordjournals.aje.a121575
- [6] J.A. Yorke, N. Nathanson, G. Pianigiani, J. Martin, Seasonality and the requirements for perpetuation and eradication of viruses in populations, Am. J. Epidemiol. 109 (2) (1979) 103–123. https://doi.org/10.1093/oxfordjournals.aje.a112666
- [7] J.H. Epstein, J.T. Price, The significant but understudied impact of pathogen transmission from humans to animals, Mt. Sinai. J. Med. 76
 (5) (2009) 448–455. https://doi.org/10.1002/msj.20140
- [8] J. Torales, M. O'Higgins, J.M. Castaldelli-Maia, A. Ventriglio, The outbreak of COVID-19 coronavirus and its impact on global mental health, Int. J. Soc. Psychiatry 66 (4) (2020) 317-320. https://doi.org/10.1177/0020764020915212
- [9] M. Zheng, C. Wang, J. Zhou, M. Zhao, S. Guan, Y. Zou, Z. Liu, Nonperiodic outbreaks of recurrent epidemics and its network modelling, Sci. Rep. 5 (1) (2015) 16010. https://doi.org/10.1038/srep16010
- K. Dietz, The incidence of infectious diseases under the influence of seasonal fluctuations, in: J. Berger, W.J. Bühler, R. Repges, P. Tautu (Eds.), Mathematical Models in Medicine: Workshop, Mainz, March 1976, Springer-Verlag, Berlin, 1976, pp. 1–15. https://doi.org/10.1007/978-3-642-93048-5_1
- [11] S. Altizer, A. Dobson, P. Hosseini, P. Hudson, M. Pascual, P. Rohani, Seasonality and the dynamics of infectious diseases, Ecol. Lett. 9 (4) (2006) 467–484. https://doi.org/10.1111/j.1461-0248.2005.00879.x
- [12] N.C. Grassly, C. Fraser, Seasonal infectious disease epidemiology, Proc. R. Soc. B: Biol. Sci. 273 (1600) (2006) 2541–2550. https://doi.org/10.1098/rspb.2006.3604

- [13] G. Cruz-Pacheco, L. Esteva, C. Vargas, Seasonality and outbreaks in West Nile virus infection, Bull. Math. Biol. 71 (6) (2009) 1378–1393. https://doi.org/10.1007/s11538-009-9406-x
- [14] A. d'Onofrio, P. Manfredi, Information-related changes in contact patterns may trigger oscillations in the endemic prevalence of infectious diseases, J. Theor. Biol. 256 (3) (2009) 473–478. https://doi.org/10.1016/j.jtbi.2008.10.005
- [15] C. Han, M, Li, N. Haihambo, P. Babuna, Q. Liu, X. Zhao, C. Jaeger, Y. Li, S. Yang, Mechanisms of recurrent outbreak of COVID-19: a model-based study, Nonlinear Dyn. 106 (2) (2021) 1169–1185. https://doi.org/10.1007/s11071-021-06371-w
- [16] F. Caccioli, D. De Martino, Epidemic oscillations induced by social network control, J. Stat. Mech.: Theory Exp. 2022 (1) (2022) 013404. https://doi.org/10.1088/1742-5468/ac4804
- Z. Qiu, B. Espinoza, V.V. Vasconcelos, C. Chen, S.M. Constantino, S.A. Crabtree. L. Yang, A. Vuliknti, J. Chen, J. Weibull, K. Basu, A. Dixit, S.A. Levin, M.V. Marathe, Understanding the coevolution of mask wearing and epidemics: A network perspective, Proc. Natl. Acad. Sci. USA 119 (26) (2022) e2123355119. https://doi.org/10.1073/pnas.2123355119
- [18] Y. Iwasa, R. Hayashi, Waves of infection emerging from coupled social and epidemiological dynamics, J. Theor. Biol. 558 (2023) 111366. https://doi.org/10.1016/j.jtbi.2022.111366
- [19] M. Ochab, P. Manfredi, K. Puszynski, A. d'Onofrio, Multiple epidemic waves as the outcome of stochastic SIR epidemics with behavioral responses: a hybrid modeling approach, Nonlinear Dyn. 111 (1) (2023) 887–926. https://doi.org/10.1007/s11071-022-07317-6
- [20] J.R. Davies, E.A. Grilli, A.J. Smith, Influenza A: infection and reinfection, Epidemiol. Infect. 92 (1) (1984) 125–127. https://doi.org/10.1017/S002217240006410X
- [21] A.J. Hay, V. Gregory, A.R. Douglas, Y.P. Lin, The evolution of human influenza viruses, Philos. Trans. R. Soc. Lond. B Biol. Sci. 356 (1416) (2001) 1861–1870. https://doi.org/10.1098/rstb.2001.0999
- [22] D.J. Earn, J. Dushoff, S.A. Levin, Ecology and evolution of the flu, Trends Ecol. Evol. 17 (7) (2002) 334–340. https://doi.org/10.1016/S0169-5347(02)02502-8
- [23] O. Price, F. Birrell, E. Mifsud, S. Sullivan, Epidemiology of repeat influenza infection in Queensland, Australia, 2005–2017, Epidemiol. Infect. 150 (2018) E144. https://doi.org/10.1017/S0950268822001157
- [24] J. Wang, L. Jiang, Y. Xu, W. He, C. Zhang, F. Bi, Y. Tan, C. Ning, Epidemiology of influenza virus reinfection in Guangxi, China: a retrospective analysis of a nine-year influenza surveillance data: Characteristics of influenza virus reinfection, Int. J. Infect. Dis. 120 (2022) 135–141. https://doi.org/10.1016/j.ijid.2022.04.045
- [25] H.W. Hethcote, Simulations of pertussis epidemiology in the United States: effects of adult booster vaccinations, Math. Biosci. 158 (1) (1999) 47–73. https://doi.org/10.1016/S0025-5564(99)00004-8

- [26] M. van Boven, H.E. de Melker, J.F. Schellekens, M. Kretzschmar, Waning immunity and sub-clinical infection in an epidemic model: implications for pertussis in The Netherlands, Math. Biosci. 164 (2) (2000) 161–182. https://doi.org/10.1016/S0025-5564(00)00009-2
- [27] R.B. Nadelman, K. Hanincová, P. Mukherjee, D. Liveris, J. Nowakowski, D. McKenna, D. Brisson, D. Cooper, S. Bittker, G. Madison, D. Holmgren, I. Schwartz, G.P. Wormser, Differentiation of reinfection from relapse in recurrent Lyme disease, N. Engl. J. Med. 367 (20) (2012) 1883–1890. https://doi.org/10.1056/NEJM0a1114362
- [28] Q. Zhang, M. Zhou, Y. Yang, E. You, J. Wu, W. Zhang, J. Jin, F. Huang, Short-term effects of extreme meteorological factors on childhood hand, foot, and mouth disease reinfection in Hefei, China: A distributed lag non-linear analysis, Sci. Total Environ. 653 (2019) 839–848. https://doi.org/10.1016/j.scitotenv.2018.10.349
- [29] C.F. Arias, F.J. Acosta, C. Fernandez-Arias, Killing the competition: a theoretical framework for liver-stage malaria, Open Biol. 12 (2022) 210341. https://doi.org/10.1098/rsob.210341
- [30] A.U. Rehman, R. Singh, J. Singh, Mathematical analysis of multi-compartmental malaria transmission model with Solitons reinfection, Chaos Fractals 163 (2022)112527. https://doi.org/10.1016/j.chaos.2022.112527
- [31] E. Vynnycky, P.E. Fine, The natural history of tuberculosis: the implications of age-dependent risks of disease and the role of reinfection, Epidemiol. Infect. 119 (2) (1997) 183–201. https://doi.org/10.1017/S0950268897007917
- [32] C.R. Horsburgh Jr, Y. Jo, B. Nichols, H.E. Jenkins, C.A. Russell, L.F. White, Contribution of Reinfection to Annual Rate of Tuberculosis Infection (ARI) and Incidence of Tuberculosis Disease, Clin. Infect. Dis. 76 (3) 2023 e965–e972. https://doi.org/10.1093/cid/ciac451
- [33] B. Qiu, Z. Wu, B. Tao, Z. Li, H. Song, D. Tian, J. Wu, M. Zhan, J. Wang, Risk factors for types of recurrent tuberculosis (reactivation versus reinfection): A global systematic review and meta-analysis, Int. J. Infect. Dis. 116 (2022) 14–20. https://doi.org/10.1016/j.ijid.2021.12.344
- [34] C.R. MacIntyre, A.A. Chughtai, Recurrence and reinfection a new paradigm for the management of Ebola virus disease, Int. J. Infect. Dis. 43 (2016) 58–61. https://doi.org/10.1016/j.ijid.2015.12.011
- [35] F.B. Agusto, Mathematical model of Ebola transmission dynamics with relapse and reinfection, Math. Biosci. 283 (2017) 48–59. https://doi.org/10.1016/j.mbs.2016.11.002
- [36] H-K. Yum, I-N. Park, B-M. Shin, S-J. Choi, Recurrent Pseudomonas aeruginosa Infection in Chronic Lung Diseases: Relapse or Reinfection?, Tuberc. Respir. Dis. 77 (4) (2014) 172–177. https://doi.org/10.4046/trd.2014.77.4.172
- [37] M. Lipsitch, Vaccination against colonizing bacteria with multiple serotypes, Proc. Natl. Acad. Sci. USA 94 (12) (1997) 6571–6576. https://doi.org/10.1073/pnas.94.12.6571

- [38] S. Gupta, M.C. Maiden, Exploring the evolution of diversity in pathogen populations, Trends Microbiol. 9 (4) (2001) 181–185. https://doi.org/10.1016/S0966-842X(01)01986-2
- [39] N.W. Crawford, Importance of understanding the reinfection risk of COVID-19 in children, The Lancet Child & Adolescent Health 6 (6) (2022) 355–357. https://doi.org/10.1016/S2352-4642(22)00093-1
- [40] A. Kumar, N. Shiwalkar, J.D. Shaikh, R. Kaur, A.F.V. Leite, P. Persaud, A. Kichloo, A.J. Rodríguez-Morales, R. Sah, Reinfection after SARS-CoV-2 infection: A looming concern, J. Exp. Biol. Agric. Sci. 8 (Spl-1-SARS-CoV-2) (2020) S114–S118. https://doi.org/10.18006/2020.8(Spl-1-SARS-CoV-2).S114.S118
- [41] M. Le Page, The risks from covid-19 reinfection, New Sci. 255 (3394) (2022) 21. https://doi.org/doi.org/10.1016/S0262-4079(22)01208-8
- [42] A.A. Mensah, J. Lacy, J. Stowe, G. Seghezzo, R. Sachdeva, R. Simmons, A. Bukasa, S. O'Boyle, N. Andrews, M. Ramsay, H. Campbell, K. Brown, Disease severity during SARS-COV-2 reinfection: a nationwide study, J. Infect. 84 (4) (2022) 542–550. https://doi.org/10.1016/j.jinf.2022.01.012
- [43] N.N. Nguyen, L. Houhamdi, V.T. Hoang, J. Delerce, L. Delorme, P. Colson, P. Brouqui, P-E. Fournier, D. Raoult, P. Gautret, SARS-CoV-2 reinfection and COVID-19 severity, Emerg. Microbes Infect. 11 (1) (2022) 894–901. https://doi.org/10.1080/22221751.2022.2052358
- [44] X. Ren, J. Zhou, J. Guo, C. Hao, M. Zheng, R. Zhang, Q. Huang, X. Yao, R. Li, Y. Jin, Reinfection in patients with COVID-19: a systematic review, Glob. Health Res. Policy 7 (2022) 12. https://doi.org/10.1186/s41256-022-00245-3
- [45] C.M. Saad-Roy, J.E. Metcalf, B.T. Grenfell, Immuno-epidemiology and the predictability of viral evolution, Science 376 (6598) (2022) 1161–1162. https://doi.org/10.1126/science.abn9410
- [46] H.J. Salzer, M. Neuböck, S. Heldt, I. Haug, C. Paar, B. Lamprecht, Emerging COVID-19 reinfection four months after primary SARS-CoV-2 infection, Wiener Medizinische Wochenschrift 172 (3) (2022) 74–76. https://doi.org/10.1007/s10354-021-00813-1
- [47] N.A. Shaheen, R. Sambas, M. Alenezi, N.K. Alharbi, O. Aldibasi, M. Bosaeed, COVID-19 reinfection: A multicenter retrospective study in Saudi Arabia, Ann. Thorac. Med. 17 (2) (2022) 81–86. https://doi.org/10.4103/atm.atm_74_22
- [48] S. Cobey, M. Lipsitch, Pathogen diversity and hidden regimes of apparent competition, Am. Sci. 181 (1) (2013) 12–24. https://doi.org/10.1086/668598
- [49] A. Rynda-Apple, K.M. Robinson, J.F. Alcorn, Influenza and bacterial superinfection: illuminating the immunologic mechanisms of disease, Infect. Immun. 83 (10) (2015) 3764–3770 https://doi.org/10.1128/iai.00298-15
- [50] S. Gupta, N. Ferguson, R. Anderson, Chaos, persistence, and evolution of strain structure in antigenically diverse infectious agents, Science 280 (5365) (1998) 912–915. https://doi.org/10.1126/science.280.5365.912

- [51] T.M.T. Le, E. Gjini, S. Madec, Quasi-neutral dynamics in a coinfection system with N strains and asymmetries along multiple traits, J. Math. Biol. 87 (3) (2023) 48. https://doi.org/10.1007/s00285-023-01977-7
- [52] M. Martcheva, An introduction to mathematical epidemiology, Springer, New York, 2015. https://doi.org/10.1007/978-1-4899-7612-3
- [53] M.A. Nowak, R.M. May, Superinfection and the evolution of parasite virulence, Proc. Royal Soc. B: Biol. Sci. 255 (1342) (1994) 81–89. https://doi.org/10.1098/rspb.1994.0012
- [54] Q. Wu, M. Small, H. Liu, Superinfection Behaviors on Scale-Free Networks with Competing Strains, J. Nonlinear Sci. 23 (2013) 113–127. https://doi.org/10.1007/s00332-012-9146-1
- [55] W.J. Edmunds, N.J. Gay, M. Kretzschmar, R.G. Pebody, H. Wachmann, The pre-vaccination epidemiology of measles, mumps and rubella in Europe: implications for modelling studies, Epidemiol. Infect. 125 (3) (2000) 635–650. https://doi.org/10.1017/S0950268800004672
- [56] S. Locarnini, A. Hatzakis, D.S. Chen, A. Lok, Strategies to control hepatitis B: Public policy, epidemiology, vaccine and drugs, J. Hepatol. 62 (1) (2015) S76–S86. https://doi.org/10.1016/j.jhep.2015.01.018
- [57] N.M. Ferguson, D.A.T. Cummings, S. Cauchemez, C. Fraser, S. Riley, A. Meeyai, S. Iamsirithaworn, D.S. Burke, Strategies for containing an emerging influenza pandemic in Southeast Asia, Nat. 437 (7056) (2005) 209–214. https://doi.org/10.1038/nature04017
- [58] S. Funk, E. Gilad, C. Watkins, V.A.A. Jansen, The spread of awareness and its impact on epidemic outbreaks, Proc. Natl. Acad. Sci. USA 106 (16) (2009) 6872–6877. https://doi.org/10.1073/pnas.0810762106
- [59] A. Perisic, C.T. Bauch, Social contact networks and disease eradicability under voluntary vaccination, PLoS Comput. Biol. 5 (2) (2009) e1000280. https://doi.org/10.1371/journal.pcbi.1000280
- [60] S. Funk, M. Salathé, V.A.A. Jansen, Modelling the influence of human behaviour on the spread of infectious diseases: a review, J. R. Soc. Interface 7 (5) (2010) 1247–1256. https://doi.org/10.1098/rsif.2010.0142
- [61] C. Bauch, A. d'Onofrio, P. Manfredi, Behavioral epidemiology of infectious diseases: an overview, in: P. Manfredi, A. d'Onofrio (Eds.), Modeling the Interplay Between Human Behavior and the Spread of Infectious Diseases, Springer New York, New York, 2013, pp. 1–19. https://doi.org/10.1007/978-1-4614-5474-8
- [62] I.Z. Kiss, J. Cassell, M. Recker, P.L. Simon, The impact of information transmission on epidemic outbreaks, Math. Biosci. 225 (1) (2010) 1–10. https://doi.org/10.1016/j.mbs.2009.11.009
- [63] J.T.F. Lau, X. Yang, H.Y. Tsui, J.H. Kim, Impacts of SARS on health-seeking behaviors in general population in Hong Kong, Prev. Med. 41
 (2) (2005) 454–462. https://doi.org/10.1016/j.ypmed.2004.11.023
- [64] J.L. Brower, The threat and response to infectious diseases (Revised), Microb. Ecol. 76 (1) (2018) 19–36. https://doi.org/10.1007/s00248-016-0806-9

- [65] R. West, S. Michie, G.J. Rubin, R. Amlôt, Applying principles of behaviour change to reduce SARS-CoV-2 transmission, Nat. Hum. Behav. 4 (5) (2020) 451–459. https://doi.org/10.1038/s41562-020-0887-9
- [66] B. Yanti, E. Mulyadi, Wahiduddin, R.G.H. Novika, Y.M.D. Arina, N.S. Martani, Nawan, Community knowledge, attitudes, and behavior towards social distancing policy as a means of preventing transmission of COVID-19 in Indonesia. J. Adm. Kesehat Indones 8 (1) (2020) 4-14. https://doi.org/10.20473/jaki.v8i2.2020.4-14
- [67] L.B. Shaw, I.B. Schwartz, Enhanced vaccine control of epidemics in adaptive networks, Phys. Rev. E 81 (4) (2010) 046120. https://doi.org/10.1103/PhysRevE.81.046120
- [68] T.C. Reluga, Game theory of social distancing in response to an epidemic, PLoS Comput. Biol. 6 (5) (2020) e1000793. https://doi.org/10.1371/journal.pcbi.1000793
- [69] L. Kyne, M.B. Hamel, R. Polavaram, C.P. Kelly, Health care costs and mortality associated with nosocomial diarrhea due to Clostridium difficile, Clin. Infect. Dis. 34 (3) (2002) 346–353. https://doi.org/10.1086/338260
- [70] A. Haakenstad, A.C. Harle, G. Tsakalos, A.E. Micah, T. Tao, M. Anjomshoa, J. Cohen, N. Fullman, S.I. Hay, T. Mestrovic, S. Mohammed, S.M. Mousavi, M.R. Nixon, D. Pigott, K. Tran, C.J.L. Murray, J.L. Dieleman, Tracking spending on malaria by source in 106 countries, 2000-16: an economic modelling study, Lancet Infect. Dis. 19 (7) (2019) 703–716. https://doi.org/10.1016/S1473-3099(19)30165-3
- [71] R. Fairlie, The impact of COVID-19 on small business owners: Evidence from the first three months after widespread socialdistancing restrictions, J. Econ. Manag. Strategy 29 (4) (2020) 727–740. http://dx.doi.org/10.1111/jems.12400
- [72] R. Rastegar, F. Higgins-Desbiolles, L. Ruhanen, COVID-19 and a justice framework to guide tourism recovery, Ann. Tour. Res. 91 (2021) 103161. https://doi.org/10.1016/j.annals.2021.103161
- [73] M. Volgger, R. Taplin, A. Aebli, Recovery of domestic tourism during the COVID-19 pandemic: An experimental comparison of interventions, J. Hosp. Tour. Manag. 48 (2021) 428–440. https://doi.org/10.1016/j.jhtm.2021.07.015
- [74] H. Yan, H. Wei, M. Wei, Exploring tourism recovery in the post-covid-19 period: An evolutionary game theory approach, Sustainability 13 (16) (2021) 9162. https://doi.org/10.3390/su13169162
- [75] P. Jones, A review of the UK's tourism recovery plans post COVID-19, Athens Journal of Tourism 9 (1) (2022) 9–18. https://doi.org/10.30958/ajt.9-1-1
- [76] L. Okafor, E. Yan, Covid-19 vaccines, rules, deaths, and tourism recovery, Ann. Tour. Res. 95 (2022) 103424. https://doi.org/10.1016/j.annals.2022.103424
- [77] P. Palese, J.F. Young, Variation of influenza A, B, and C viruses, Science 215 (4539) (1982) 1468–1474. https://doi.org/10.1126/science.7038875

- [78] R.G. Webster, W.G. Laver, G.M. Air, G.C. Schild, Molecular mechanisms of variation in influenza viruses, Nature 296 (5853) (1982) 115–121. https://doi.org/10.1038/296115a0
- [79] V. M. Corman, O. Landt, M. Kaiser, R. Molenkamp, A. Meijer, D.K. Chu, T. Bleicker, S. Brünink, J. Schneider, M.L. Schmidt, D.G. Mulders, B.L. Haagmans, B. van der Veer, S. van den Brink, L. Wijsman, G. Goderski, J.L. Romette, J. Ellis, M. Zambon, M. Peiris, H. Goossens, C. Reusken, M.P.G. Koopmans, C. Drosten, Detection of 2019 novel coronavirus (2019-nCoV) by real-time RT-PCR, Euro surveillance: bulletin Europeen sur les maladies transmissibles = European communicable disease bulletin 25 (3) (2020) 2000045. https://doi.org/10.2807/1560-7917.ES.2020.25.3.2000045
- [80] A. Sharma, I. Ahmad Farouk, S.K. Lal, COVID-19: a review on the novel coronavirus disease evolution, transmission, detection, control and prevention, Viruses 13 (2) (2021) 202. https://doi.org/10.3390/v13020202
- [81] M. Cabrera, F. Córdova-Lepe, J.P. Gutiérrez-Jara, K. Vogt-Geisse, An SIR-type epidemiological model that integrates social distancing as a dynamic law based on point prevalence and socio-behavioral factors, Sci. Rep. 11 (1) (2021) 10170. https://doi.org/10.1038/s41598-021-89492-x
- [82] S. Funk, E. Gilad, V.A.A. Jansen, Endemic disease, awareness, and local behavioural response, J. Theor. Biol. 264 (2) (2010) 501–509. https://doi.org/10.1016/j.jtbi.2010.02.032
- [83] G.O. Agaba, Y.N. Kyrychko, K.B. Blyuss, Mathematical model for the impact of awareness on the dynamics of infectious diseases, Math. Biosci. 286 (2017) 22–30. https://doi.org/10.1016/j.mbs.2017.01.009
- [84] A.K. Misra, A. Sharma, J.B. Shukla, Modeling and analysis of effects of awareness programs by media on the spread of infectious diseases, Math. Comput. Model. 53 (5) (2011) 1221–1228. https://doi.org/10.1016/j.mcm.2010.12.005
- [85] F.A. Basir, S. Raya, E. Venturino, Role of media coverage and delay in controlling infectious diseases: A mathematical model, Appl. Math. Comput. 337 (2018) 372–385. https://doi.org/10.1016/j.amc.2018.05.042
- [86] M.E. Wilson, Global travel and emerging infections, Infectious Disease Movement in a Borderless World: Workshop Summary, 2010 90–104.
- [87] K-Y. Wang, How change of public transportation usage reveals fear of the SARS virus in a city, PLoS One 9 (3) (2014) e89405. https://doi.org/10.1371/journal.pone. 0089405
- [88] World Health Organization (WHO), Managing epidemics: key facts about major deadly diseases. World Health Organization. https://apps.who.int/iris/handle/10665/272442, 2018. License: CC BY-NC-SA 3.0 IGO.
- [89] H. Seno, An SIS model for the epidemic dynamics with two phases of the human day-to-day activity, J. Math. Biol. 80 (7) (2020) 2109-2140. https://doi.org/10.1007/s00285-020-01491-0

- [90] Centre for Disease Control and Prevention (CDC), COVID-19. https://www.cdc.gov/coronavirus/2019-ncov/index.html, 2022 (accessed 21 June 2022).
- [91] European Centre for Disease Prevention and Control (ECDC), COVID-19. https://www.ecdc.europa.eu/en/covid-19, 2022 (accessed 21 June 2022)
- [92] World Health Organization (WHO), Coronavirus disease (COVID-19) pandemic. https://www.who.int/emergencies/diseases/novelcoronavirus-2019, 2022 (accessed 21 June 2022)
- [93] N. Parikh, M. Youssef, S. Swarup, S. Eubank, Modeling the effect of transient populations on epidemics in Washington DC, Sci. Rep. 3 3152 (2013) 8pp. https://doi.org/10.1038/srep03152
- [94] D.M. Tompkins, S. Carver, M.E. Jones, M. Krkošek, L.F. Skerratt, Emerging infectious diseases of wildlife: a critical perspective, Trends in Parasitology 31 (4) (2015) 149–159. https://doi.org/10.1016/j.pt.2015.01.007
- [95] L. Mur, J.M. Sánchez-Vizcaíno, E. Fernández-Carrión, C. Jurado, S. Rolesu, F. Feliziani, A. Laddomada, B. Martínez-López, Understanding African Swine Fever infection dynamics in Sardinia using a spatially explicit transmission model in domestic pig farms, Transbound. Emerg. Dis. 65 (2018) 123–134. https://doi.org/10.1111/tbed.12636
- [96] A. Postel, S. Austermann-Busch, A. Petrov, V, Moennig, P. Becher, Epidemiology, diagnosis and control of classical swine fever: Recent developments and future challenges, Transbound. Emerg. Dis. 65 (2018) 248–261. https://doi.org/10.1111/tbed.12676
- [97] P.Y. Geoffard, T. Philipson, Rational epidemics and their public control, Int. Econ. Rev. 37 (3) (1996) 603–624. http://dx.doi.org/10.2307/2527443
- [98] F.H. Chen, Rational behavioral response and the transmission of STDs, Theor. Popul. Biol. 66 (4) (2004) 307–316. http://dx.doi.org/10.1016/j.tpb.2004.07.004
- [99] B. Buonomo, Effects of information-dependent vaccination behavior on coronavirus outbreak: insights from a SIRI model, Ric. Mat. 69 (2020) 483–499. https://doi.org/10.1007/s11587-020-00506-8
- [100] Z. Chladná, J. Kopfová, D. Rachinskii, S. Rouf, Global dynamics of SIR model with switched transmission rate, J. Math. Biol. 80 (2020) 1209–1233. http://dx.doi.org/10.1007/s00285-019-01460-2
- [101] D. Rachinskii, S. Rouf, Dynamics of SIR model with heterogeneous response to intervention policy, Theor. Popul. Biol. 146 (2022) 71–85. https://doi.org/10.1016/j.tpb.2022.06.003
- [102] A.F. Filippov, Differential Equations with Discontinuous Righthand Sides: Control Systems, Springer Science+Business Media, Dordrecht, 1988. https://doi.org/10.1007/978-94-015-7793-9
- [103] Y.A. Kuznetsov, S. Rinaldi, A. Gragnani, One-parameter bifurcations in planar Filippov systems, Int. J. Bifurc. Chaos 13 (8) (2003) 2157–2188. https://doi.org/10.1142/S0218127403007874

- [104] M. di Bernardo, C. Budd, A.R. Champneys, P. Kowalczyk, Piecewisesmooth Dynamical Systems: Theory and Applications, Springer-Verlag, London, 2008. https://doi.org/10.1007/978-1-84628-708-4
- [105] M. di Bernardo, S.J. Hogan, Discontinuity-induced bifurcations of piecewise smooth dynamical systems, Philos. Trans. R. Soc. A 368 (1930) (2010) 4915–4935. https://doi.org/10.1098/rsta.2010.0198
- [106] M. Antali, G. Stepan, Sliding and crossing dynamics in extended Filippov systems, SIAM J. Appl. Dyn. Syst. 17 (2018) 823–858. https://doi.org/10.1137/17M1110328
- [107] I. Belykh, R. Kuske, M. Porfiri, D.J. Simpson, Beyond the Bristol book: Advances and perspectives in non-smooth dynamics and applications, Chaos 33 (1) (2023) 010402. https://doi.org/10.1063/5.0138169
- [108] B. Buonomo, A. d'Onofrio, D. Lacitignola, Globally stable endemicity for infectious diseases with information-related changes in contact patterns, Appl. Math. Lett. 25 (7) (2012) 1056–1060. https://doi.org/10.1016/j.aml.2012.03.016
- [109] C. Vargas-De-León, A. d'Onofrio, Global stability of infectious disease models with contact rate as a function of prevalence index, Math. Biosci. Eng. 14 (4) (2017) 1019–1033. https://doi.org/10.3934/mbe.2017053
- [110] R. Hu, L. Liu, X. Ren, X. Liu, Global stability of an information-related epidemic model with age-dependent latency and relapse, Ecol. Complex. 36 (2018) 30–47. https://doi.org/10.1016/j.ecocom.2018.06.006
- [111] R.W. Wilda, T. Trisilowati, M.A. Imron, Sensitivity and stability analysis of a SEIR epidemic model with information, J. Exp. Life Sci. 9 (1) (2019) 47–53. https://doi.org/10.21776/ub.jels.2019.009.01.08
- [112] A. P.K. Srivastava, Kumar, Υ. Takeuchi, Modeling the role of information and limited optimal treatment on prevalence, J. Theor. Biol. disease 414 (2017), 103-119. http://dx.doi.org/10.1016/j.jtbi.2016.11.016
- [113] B. Buonomo, R. Della Marca, Effects of information-induced behavioural changes during the COVID-19 lockdowns: the case of Italy, R. Soc. Open Sci. 7 (10) (2020) 201635. http://dx.doi.org/10.1098/rsos.201635
- [114] X. Jin, J. Jia, Qualitative study of a stochastic SIRS epidemic model with information intervention, Phys. A: Stat. Mech. Appl. 547 (2020) 123866. https://doi.org/10.1016/j.physa.2019.123866
- [115] S. Saha, G.P. Samanta, J.J. Nieto, Epidemic model of COVID-19 outbreak by inducing behavioural response in population, Nonlinear Dyn. 102 (2020) 455–487. https://doi.org/10.1007/s11071-020-05896-w
- [116] A. d'Onofrio, P. Manfredi, Behavioral SIR models with incidencebased social-distancing, Chaos Solitons Fractals 159 (2022) 112072. https://doi.org/10.1016/j.chaos.2022.112072
- [117] B. Tang, W. Zhou, X. Wang, H. Wu, Y. Xiao, Controlling multiple COVID-19 epidemic waves: an insight from a multi-scale model linking the behaviour change dynamics to the disease transmission dynamics,

Bull. Math. Biol. 84 (10) (2022) 106. https://doi.org/10.1007/s11538-022-01061-z

- [118] I.M. Bulai, F. Montefusco, M.G. Pedersen, Stability analysis of a model of epidemic dynamics with nonlinear feedback producing recurrent infection waves, Appl. Math. Lett. 136 (2023) 108455. https://doi.org/10.1016/j.aml.2022.108455
- [119] B. Buonomo, A. Giacobbe, Oscillations in SIR behavioural epidemic models: The interplay between behaviour and overexposure to infection, Chaos Solitons Fractals 174 (2023) 113782. https://doi.org/10.1016/j.chaos.2023.113782
- [120] S. Guerrero-Flores, O. Osuna, C. Vargas-De-León, Periodic solutions of seasonal epidemiological models with informationdependent vaccination, Math. Meth. Appl. Sci. 2023 (2023) 1–12. https://doi.org/10.1002/mma.9728
- [121] O. Diekmann, H. Heesterbeek, T. Britton, Mathematical Tools for Understanding Infectious Disease Dynamics, Princeton Series in Theoretical and Computational Biology, Princeton University Press, Princeton, 2013
- [122] P.L. Delamater, E.J. Street, T.F. Leslie, Y.T. Yang, K.H. Jacobsen, Complexity of the basic reproduction number (R₀), Emerg. Infect. Dis. 25 (1) (2019) 1–4. https://doi.org/10.3201/eid2501.171901
- [123] M.A. Lewis, Z. Shuai, P. van den Driessche, A general theory for target reproduction numbers with applications to ecology and epidemiology, J. Math. Biol. 78 (2019) 2317–2339. https://doi.org/10.1007/s00285-019-01345-4
- [124] H. Seno, A Primer on Population Dynamics Modeling: Basic Ideas for Mathematical Formulation, Springer, Singapore, 2022. https://doi.org/10.1007/978-981-19-6016-1
- [125] M. Iannelli, A. Pugliese, An Introduction to Mathematical Population Dynamics: Along the trail of Volterra and Lotka, Springer, Cham, 2014. https://doi.org/10.1007/978-3-319-03026-5
- [126] L.A. Segel, M. Slemrod, The quasi-steady-state assumption: a case study in perturbation, SIAM Rev. 31 (3) (1989) 446–477. https://doi.org/10.1137/1031091
- [127] D. Aldila, H. Seno, A population dynamics model of mosquito-borne disease transmission, focusing on mosquitoes' biased distribution and mosquito repellent use, Bull. Math. Biol. 81 (12) (2019) 4977–5008. https://doi.org/10.1007/s11538-019-00666-1
- [128] S. Samanta, S. Rana, A. Sharma, A.K. Misra, J. Chattopadhyay, Effect of awareness programs by media on the epidemic outbreaks: A mathematical model, Appl. Math. Comput. 219 (12) (2013) 6965–6977. https://doi.org/10.1016/j.amc.2013.01.009
- [129] S. Samanta, Effects of awareness program and delay in the epidemic outbreak, Math. Meth. Appl. Sci. 40 (5) (2017) 1679–1695. https://doi.org/10.1002/mma.4089
- [130] M.G. Gomes, L.J. White, G.F. Medley, Infection, reinfection, and vaccination under suboptimal immune protection: epidemiological perspectives, J. Theor. Biol. 228 (4) (2004) 539–549. https://doi.org/10.1016/j.jtbi.2004.02.015
- [131] M.G. Gomes, L.J. White, G.F. Medley, The reinfection threshold, J. Theor. Biol. 236 (1) (2005) 111–113. https://doi.org/10.1016/j.jtbi.2005.03.001
- [132] D. Gökaydin, J.B. Oliveira-Martins, I. Gordo, M.G. Gomes, The reinfection threshold regulates pathogen diversity: the case of influenza, J. R. Soc. Interface 4 (12) (2007) 137–142. https://doi.org/10.1098/rsif.2006.0159
- [133] N. Stollenwerk, J. Martins, A. Pinto, The phase transition lines in pair approximation for the basic reinfection model SIRI, Phys. Lett. A 371 (5-6) (2007) 379–388. https://doi.org/10.1016/j.physleta.2007.06.040
- [134] J. Martins, A. Pinto, N. Stollenwerk, A scaling analysis in the SIRI epidemiological model, J. Biol. Dyn. 3 (5) (2009) 479-496. https://doi.org/10.1080/17513750802601058
- [135] A. Pinto, M. Aguiar, J. Martins, N. Stollenwerk, Dynamics of epidemiological models, Acta Biotheor. 58 (2010) 381–389. https://doi.org/10.1007/s10441-010-9116-7
- [136] L.P. Song, Z. Jin, G.Q. Sun, Reinfection induced disease in a spatial SIRI model, J. Biol. Phys. 37 (2011) 133–140. https://doi.org/10.1007/s10867-010-9204-6
- [137] P. Georgescu, H. Zhang, A Lyapunov functional for a SIRI model with nonlinear incidence of infection and relapse, Appl. Math. Comput. 219 (2013) (16) 8496–8507. https://doi.org/10.1016/j.amc.2013.02.044
- [138] P. Guo, X. Yang, Z. Yang, Dynamical behaviors of an SIRI epidemic model with nonlinear incidence and latent period, Advances in Difference Equations 2014 (2014) 164. https://doi.org/10.1186/1687-1847-2014-164
- [139] R, Pagliara, B. Dey, N.E. Leonard, Bistability and resurgent epidemics in reinfection models, IEEE Control Syst. Lett. 2 (2) (2018) 290–295. https://doi.org/10.1109/LCSYS.2018.2832063
- [140] B. Buonomo, Effects of information-dependent vaccination behavior on coronavirus outbreak: insights from a SIRI model, Ric. Mat. 69 (2020) 483–499. https://doi.org/10.1007/s11587-020-00506-8
- [141] M. Ghosh, S. Olaniyi, O.S. Obabiyi, Mathematical analysis of reinfection and relapse in malaria dynamics, Appl. Math. Comput. 373 (2020) 125044. https://doi.org/10.1016/j.amc.2020.125044
- [142] S. Wang, Dynamic Assessment of Population by Using SIRI Model, In: 4th International Conference on Biometric Engineering and Applications 2021 May 25, 2021 (pp. 93-97). https://doi.org/10.1145/3476779.3476795
- [143] A. Srivastava, P.K. Sonu & Srivastava, Nonlinear dynamics of a SIRI model incorporating the impact of information and saturated treatment with optimal control, Eur. Phys. J. Plus 137 (9) (2022) 1028. https://doi.org/10.1140/epjp/s13360-022-03201-9

- [144] D.L. Heymann, Social, behavioural and environmental factors and their impact on infectious disease outbreaks, Journal of public health policy 26 (1) (2005) 133-139. https://doi.org/10.1057/palgrave.jphp.3200004
- [145] P. Manfredi, A. D'Onofrio (eds.), Modeling the Interplay Between Human Behavior and the Spread of Infectious Diseases, Springer New York, New York, 2013. https://doi.org/10.1007/978-1-4614-5474-8
- [146] J.J. Bavel, K. Baicker, P.S. Boggio, V. Capraro, A. Cichocka, M. Cikara, M.J. Crockett, A.J. Crum, K.M. Douglas, J.N. Druckman, J. Drury, Using social and behavioural science to support COVID-19 pandemic response, Nat. Hum. Behav. 4 (5) (2020) 460–471. https://doi.org/10.1038/s41562-020-0884-z
- [147] M. Kapitány-Fövény, M. Sulyok, Social markers of a pandemic: modeling the association between cultural norms and COVID-19 spread data, Humanities and Social Sciences Communications 7 (1) (2020) 1–9. https://doi.org/10.1057/s41599-020-00590-z
- [148] T.P. Thu, P.N. Ngoc, N.M. Hai, Effect of the social distancing measures on the spread of COVID-19 in 10 highly infected countries, Sci. Total Environ. 742 (2020) 140430. https://doi.org/10.1016/j.scitotenv.2020.140430
- [149] V.H. Smith, R.D. Holt, Resource competition and withinhost disease dynamics, Trends Ecol. Evol. 11(9) (1996) 386–389. https://doi.org/10.1016/0169-5347(96)20067-9
- [150] M.E. Hood, Dynamics of multiple infection and within-host competition by the anther-smut pathogen, Am. Nat. 162(1) (2003) 122–133. https://doi.org/10.1086/375539
- [151] J.C. De Roode, M.E. Helinski, M.A. Anwar, A.F. Read, Dynamics of multiple infection and within-host competition in genetically diverse malaria infections, Am. Nat. 166(5) (2005) 531–542. https://doi.org/10.1086/491659
- [152] J.C. De Roode, R. Pansini, S.J. Cheesman, M.E. Helinski, S. Huijben, A.R. Wargo, A.S. Bell, H.K. Chan, A. Walliker, A.F. Read, Virulence and competitive ability in genetically diverse malaria infections, Proc. Natl. Acad. Sci. USA 102(21) (2005) 7624–7628. https://doi.org/10.1073/pnas.0500078102
- [153] A. S. Bell, J.C. De. Roode, D. Sim, A.F. Read, Within-host competition in genetically diverse malaria infections: parasite virulence and competitive success, Evol. 60(7) (2006) 1358–1371. https://doi.org/10.1111/j.0014-3820.2006.tb01215.x
- [154] A. Gold, T. Giraud, M.E. Hood, Within-host competitive exclusion among species of the anther smut pathogen, BMC Ecol. 9 (2009) 1–7. https://doi.org/10.1186/1472-6785-9-11



A.1 PROOF OF LEMMA 2.1 IN SECTION 2.3.3

From (8), it is necessary for the existence of E_+ that the following equation in terms of M has a positive root $M = M^*$:

$$\frac{\delta}{\eta}M = 1 - \frac{1}{\mathscr{R}_0} \frac{\beta_0}{\beta(M)}.$$
 (A.1)

Since $\beta(M)$ is continuous, positive, and decreasing in terms of $M \in [0, \infty)$ from the assumption introduced in Section 2.1 and $\beta(0) = \beta_0$, the right side of (A.1) is a decreasing function of M and its supremum value is given by $1 - 1/\Re_0$ for M = 0. Hence, if $1 - 1/\Re_0 \leq 0$, the right side of (A.1) is non-positive for any M > 0, so that we do not have any positive root $M = M^*$ for the equation (A.1). Inversely, if $1 - 1/\Re_0 > 0$, the right side of (A.1) is continuously decreasing from $1 - 1/\Re_0 > 0$ at M = 0, while the left side of (A.1) is linearly increasing from 0 at M = 0. Thus we find that the equation (A.1) necessarily has a unique positive root $M = M^*$ in this case. Consequently, if and only if $1 - 1/\Re_0 > 0$, that is, $\Re_0 > 1$, the equation (A.1) has a unique positive root $M = M^*$.

When $M^* > 0$ exists with $\Re_0 > 1$, it is satisfied from (8) that

$$0 < \nu^* = 1 - \frac{1}{\mathscr{R}_0} \frac{\beta_0}{\beta(M^*)} < 1 - \frac{1}{\mathscr{R}_0} \frac{\beta_0}{\beta(0)} = 1 - \frac{1}{\mathscr{R}_0} < 1$$

Therefore, with a positive root $M = M^*$ for (A.1) with $\Re_{01} > 1$, we have a reasonable value $v^* \in (0, 1)$ at the endemic equilibrium E_+ . These arguments prove Lemma 2.1.

A.2 PROOF OF LEMMA 2.2 IN SECTION 2.3.4

First, it is easily seen from the system (4) that the disease-free equilibrium always exists. Next, suppose that an endemic equilibrium with $M^* = 0$ exists. From (9), we have $v^* = 1 - 1/\mathscr{R}_0$ and $G(v^*) = 0$. Therefore, for $v^* > 0$, it is necessary that $\mathscr{R}_0 > 1$. Besides, from $G(v^*) = 0$, it is necessary that $v^* = 1 - 1/\mathscr{R}_0 \le \theta_c$, that is, $\mathscr{R}_0 \le (1 - \theta_c)^{-1}$. Inversely, consider the case where $1 < \mathscr{R}_0 \le (1 - \theta_c)^{-1}$. From (9), we have

$$\nu^* = 1 - \frac{1}{\mathscr{R}_0} \frac{\beta_0}{\beta(M^*)} \leqslant 1 - \frac{1}{\mathscr{R}_0} \frac{\beta_0}{\beta(0)} = 1 - \frac{1}{\mathscr{R}_0},$$

since $\beta(M)$ is a decreasing function of M with $\beta(0) = \beta_0$. Then, from $\mathscr{R}_0 \leq (1 - \theta_c)^{-1}$, we have $1 - 1/\mathscr{R}_0 \leq \theta_c$, so that $\nu^* \leq \theta_c$. This means that $G(\nu^*) = 0$, and subsequently $M^* = G(\nu^*)/\delta = 0$ from (9). These arguments prove (ii) of Lemma 2.2.

Now suppose that an endemic equilibrium with v^* and $M^* > 0$ exists. From $M^* = G(v^*)/\delta > 0$ in (9), v^* must be larger than θ_c to satisfy that $G(v^*) > 0$. Then, $G(v^*) = \eta(v^* - \theta_c) > 0$ from (5), and $v^* = \eta M^*/\delta + \theta_c$ from (9). Therefore, from (9), the following equation in terms of M must have a positive root $M = M^*$:

$$\frac{\delta}{\eta}M + \theta_{c} = 1 - \frac{1}{\mathscr{R}_{0}} \frac{\beta_{0}}{\beta(M)}.$$
(A.2)



Figure 24: Sketches of the vector flow in the (v, M)-phase plane for the system (4) with the nullclines. (a) $\mathscr{R}_0 \leq 1$; (b) $1 < \mathscr{R}_0 \leq (1 - \theta_c)^{-1}$; (c) $\mathscr{R}_0 > (1 - \theta_c)^{-1}$.

Since the right side of (A.2) is monotonically decreasing in terms of M while the left side is linearly increasing, we find that a positive root $M = M^*$ uniquely exists if and only if

$$\theta_{\rm c} < 1 - \frac{1}{\mathscr{R}_0} \frac{\beta_0}{\beta(0)} = 1 - \frac{1}{\mathscr{R}_0},$$

that is, $\Re_0 > (1 - \theta_c)^{-1}$. From the former equation in (9) and $\nu^* = \eta M^* / \delta + \theta_c$, we have

$$\theta_{\rm c} < \nu^* = 1 - \frac{1}{\mathscr{R}_0} \frac{\beta_0}{\beta(M^*)} < 1 - \frac{1}{\mathscr{R}_0} \tag{A.3}$$

for all $M^* > 0$. Hence, when a positive root $M = M^*$ for the equation (A.2) exists, we have $v^* \in (\theta_c, 1 - 1/\Re_0)$ at the same time. Hence (iii) of Lemma 2.2 is proved.

Since $1 - 1/\mathscr{R}_0 \to \theta_c$ as $\mathscr{R}_0 \to (1 - \theta_c)^{-1} + 0$, we find from (A.3) that $\nu^* \to \theta_c$ as $\mathscr{R}_0 \to (1 - \theta_c)^{-1} + 0$ Then, from (A.2), we have $M^* \to 0$ as $\mathscr{R}_0 \to (1 - \theta_c)^{-1} + 0$. These arguments prove (iv) of Lemma 2.2.

A.3 PROOF OF THEOREM 2.2 AND COROLLARY 2.2.2 IN SECTION 2.3.4

When $\Re_0 \leq 1$, the vector flow in the (ν, M) -phase plane indicates that $(\nu(\tau), M(\tau))$ necessarily approaches the unique equilibrium E_0 , and then E_0 is globally asymptotically stable (Figure 24(a)). When $1 < \Re_0 \leq (1 - \theta_c)^{-1}$, the vector flow in the phase plane shows that $(\nu(\tau), M(\tau))$ goes far away from the equilibrium E_0 and approaches the equilibrium E_{+0} , where E_0 is unstable while E_{+0} is globally asymptotically stable (Figure 24(b)).

Next let us consider the case of $\mathscr{R}_0 > (1 - \theta_c)^{-1}$ (Figure 24). From (4), we have

$$\frac{\mathrm{d}\nu}{\mathrm{d}\tau} = \mathscr{R}_{0}\nu\Big\{\frac{\beta(\mathsf{M})}{\beta_{0}}(1-\nu) - \frac{1}{\mathscr{R}_{0}}\Big\} < \mathscr{R}_{0}\nu\Big(1-\nu-\frac{1}{\mathscr{R}_{0}}\Big),$$

for any M > 0 and $v \in (0, 1)$ since $\beta(M)$ is monotonically decreasing in terms of M and $\beta(0) = \beta_0$. Thus we have $dv/d\tau < 0$ for $v \ge 1 - 1/\mathscr{R}_0$, so that the trajectory must go to the left direction in the (v, M)-phase plane as long as $v > 1 - 1/\mathscr{R}_0$. Thus the trajectory with $v(0) \ge 1 - 1/\mathscr{R}_0$ must enter the region where $v < 1 - 1/\mathscr{R}_0$ at a certain moment. If $v(\tau_1) < 1 - 1/\mathscr{R}_0$ at a $\tau_1 > 0$, $v(\tau)$ is always less than $1 - 1/\mathscr{R}_0$ for any $\tau \ge \tau_1$, because $v(\tau)$ cannot become larger than $1 - 1/\mathscr{R}_0$ again.

In the region where $\nu \leq \theta_c$, we have $dM/d\tau = -\delta M < 0$ for any M > 0. In the region where $\theta_c < \nu < 1 - 1/\Re_0$ and $M \ge (1 - 1/\Re_0 - \theta_c)\eta/\delta$, we have

$$\begin{aligned} \frac{dM}{d\tau} &= \eta(\nu - \theta_c) - \delta M \leqslant \eta(\nu - \theta_c) - \left(1 - \frac{1}{\mathcal{R}_0} - \theta_c\right) \eta \\ &= \eta \left(\nu - 1 + \frac{1}{\mathcal{R}_0}\right) < 0 \end{aligned}$$

from (4), so that the trajectory must go down in the region of the (ν, M) -phase plane where $0 < \nu < 1 - 1/\mathscr{R}_0$ and $M \ge (1 - 1/\mathscr{R}_0 - \theta_c)\eta/\delta$. Thus the trajectory with $M(0) \ge (1 - 1/\mathscr{R}_0 - \theta_c)\eta/\delta$ must enter the region where $M < (1 - 1/\mathscr{R}_0 - \theta_c)\eta/\delta$. Consequently we find that the trajectory from any initial condition $(\nu(0), M(0)) \in U = (0, 1 - 1/\mathscr{R}_0) \times [0, (1 - 1/\mathscr{R}_0 - \theta_c)\eta/\delta)$ must remain in U for any $\tau \ge 0$. From Lemma 2.2. there is only the equilibrium E_{++} in the domain U when $\mathscr{R}_0 > (1 - \theta_c)^{-1}$. Further, making use of the standard eigenvalue analysis on the Jacobi matrix $J(E_{++})$ for E_{++} , we can show that, if the equilibrium E_{++} exists, it is locally asymptotically stable.

As a result of these arguments with Lemma 2.2, the unique equilibrium E_{++} in the domain U is locally asymptotically stable when $\mathscr{R}_0 > (1 - \theta_c)^{-1}$. Therefore, from Poincaré–Bendixson Theorem, the equilibrium E_{++} for the system (4) is globally asymptotically stable when $\mathscr{R}_0 > (1 - \theta_c)^{-1}$ for any $M(0) \ge 0$ and $\nu(0) > 0$. This proves Theorem 2.2.

When $\mathscr{R}_0 \leq (1 - \theta_c)^{-1}$, from the vector flow in the (ν, M) -phase plane shown in Figure 24(a, b), it is easily seen that the trajectory must monotonically approach the equilibrium E_0 when $\mathscr{R}_0 \leq 1$, and the equilibrium E_{+0} when $1 < \mathscr{R}_0 \leq (1 - \theta_c)^{-1}$.

Further, when $\Re_0 > (1 - \theta_c)^{-1}$, we have the discriminant of the characteristic equation for the Jacobi matrix $J(E_{++})$, Δ given by (11). If $\Delta > 0$, the eigenvalues λ_1 and λ_2 are both negative as shown in the above. Then the system (4) monotonically approaches E_{++} . If $\Delta < 0$, the eigenvalues λ_1 and λ_2 are imaginary with a negative real part. In this case, the system (4) approaches E_{++} with a damped oscillation. This proves Corollary 2.2.2.

A.4 PROOF OF THEOREM 2.3 IN SECTION 2.3.5

First let us consider the case where the social response changes much faster than the epidemic dynamics. Then we apply the QSSA to put $dM/d\tau \approx 0$. Thus we assume

$$M(\tau) \approx \frac{G(\nu(\tau))}{\delta}$$
 (A.4)

for any $\tau \ge 0$. Substituting (A.4) for (4), we get the approximated dynamics about *v*:

$$\frac{d\nu}{d\tau} = \varphi(\nu)\nu := \left\{\frac{\mathscr{R}_0}{\beta_0}\beta\left(\frac{G(\nu)}{\delta}\right)(1-\nu) - 1\right\}\nu.$$
(A.5)

When $\nu \leq \theta_c$, we have $G(\nu) = 0$, and then the equation (A.5) becomes a logistic equation same as (6). When $\nu > \theta_c$, we have $G(\nu) = \eta(\nu - \theta_c)$. In this case, we can easily find that $d\phi/d\nu < 0$ for any $\nu > 0$ because of $\beta'(M) < 0$ for any M > 0. Since $\phi(\theta_c) = \mathscr{R}_0(1 - \theta_c) - 1$, we have $\phi(\nu) < 0$ for any $\nu > \theta_c$ if $\phi(\theta_c) \leq 0$, that is, if $\mathscr{R}_0 \leq (1 - \theta_c)^{-1}$. On the other hand, we have $\phi(1) = -1 < 0$. Hence, if $\phi(\theta_c) > 0$, that is, if $\mathscr{R}_0 > (1 - \theta_c)^{-1}$, there is a unique value of ν , $\nu = \nu^* \in (\theta_c, 1)$ such that $\phi(\nu^*) = 0$, $\phi(\nu) > 0$ for $\nu < \nu^*$, and $\phi(\nu) < 0$ for $\nu > \nu^*$.

As a result, we find that, if $\mathscr{R}_0 \leq 1$, $d\nu/d\tau < 0$ for any $\tau > 0$. Then $\nu \to 0$ as $\tau \to \infty$ in a monotonic manner. If $1 < \mathscr{R}_0 \leq (1 - \theta_c)^{-1}$, ν must reach θ_c at a certain finite time for any $\nu(0) > \theta_c$ because $d\nu/d\tau < 0$ for any $\nu > \theta_c$. For $\nu(0) \leq \theta_c$, the equation (A.5) becomes the logistic equation (6) as mentioned above. Then ν monotonically approaches $1 - 1/\mathscr{R}_0 < \theta_c$ when $1 < \mathscr{R}_0 < (1 - \theta_c)^{-1}$. When $\mathscr{R}_0 = (1 - \theta_c)^{-1}$, we have $d\nu/d\tau = 0$ for $\nu = \theta_c = 1 - 1/\mathscr{R}_0$. This means that $\nu = \theta_c$ is the equilibrium for (A.5). Hence we find that $\nu \to \theta_c = 1 - 1/\mathscr{R}_0$ as $\tau \to \infty$ when $\mathscr{R}_0 = (1 - \theta_c)^{-1}$.

In contrast, if $\Re_0 > (1 - \theta_c)^{-1}$, we have $d\nu/d\tau > 0$ in (6) for any positive $\nu \leq \theta_c$. Hence ν must necessarily become greater than θ_c from any positive $\nu(0) < \theta_c$. Once ν becomes greater than θ_c , $\nu \rightarrow \nu^*$ as $\tau \rightarrow \infty$ in a monotonic manner because of the sign of $\varphi(\nu)$ and $\varphi(\nu^*) = 0$ as shown above. Consequently, these arguments prove Theorem 2.4 for the case where the social response changes extremely fast.

Next let us consider the case where the social response changes much slower than the epidemic dynamics. Then we apply the QSSA to put $d\nu/d\tau \approx 0$ with $\nu \in [0, 1)$. Thus we have

$$u(\tau) \approx 0 \quad \text{or} \quad v(\tau) \approx 1 - \frac{1}{\mathscr{R}_0} \frac{\beta_0}{\beta(M(\tau))} \leqslant 1 - \frac{1}{\mathscr{R}_0}$$
(A.6)

for any $\tau \ge 0$. Since we have the initial value M(0) = 0, we find that $v(\tau) \approx 0$ if $\mathscr{R}_0 \le 1$, while $v(\tau)$ is approximated by the latter in (A.6) if $\mathscr{R}_0 > 1$. Thus, when $\mathscr{R}_0 \le 1$, we have $M \approx 0$ for any $\tau > 0$, because G(v) = 0 so that $dM/dt \approx 0$ for any $\tau > 0$ with M(0) = 0. Let us consider hereafter the case of $\mathscr{R}_0 > 1$. When $v \le \theta_c$, we have G(v) = 0, and $dM/d\tau = -\delta M < 0$ for any M > 0. When $v > \theta_c$, substituting the latter approximation in (A.6) for (4), we get the approximated dynamics about M:

$$\frac{\mathrm{d}M}{\mathrm{d}\tau} = \psi(M) := \eta \left(1 - \frac{1}{\mathscr{R}_0} \frac{\beta_0}{\beta(M)} - \theta_c \right) - \delta M.$$

We can easily find that $d\psi(M)/\tau < 0$ for any M > 0 because $\beta'(M) < 0$ for any M > 0. Since $\psi(0) = \eta(1 - 1/\mathscr{R}_0 - \theta_c)$, if $\psi(0) \leq 0$, that is, if $\mathscr{R}_0 \leq (1 - \theta_c)^{-1}$, we have $\psi(M) < 0$ for any M > 0. On the other hand, we have $\psi(M) \to -\infty$ as $M \to \infty$. Hence, if $\psi(0) > 0$, that is, if $\mathscr{R}_0 > (1 - \theta_c)^{-1}$, there is a unique value of M, $M = M^* > 0$ such that $\psi(M^*) = 0$, $\psi(M) > 0$ for $M < M^*$, and $\psi(M) < 0$ for $M > M^*$.

As a result, we find that, if $\mathscr{R}_0 \leq 1$, $M \to 0$ as $\tau \to \infty$ in a monotonic manner. If $1 < \mathscr{R}_0 \leq (1 - \theta_c)^{-1}$, we have $\nu < 1 - 1/\mathscr{R}_0 < \theta_c$ for any $\tau > 0$ from (A.6). Then we have $dM/d\tau < 0$ for any M > 0. Thus, we have $M \to 0$ as $\tau \to \infty$ in a monotonic manner for any positive M(0). If $\mathscr{R}_0 > (1 - \theta_c)^{-1}$, we also have $\nu < 1 - 1/\mathscr{R}_0 < \theta_c$ for any $\tau > 0$ from (A.6). Then we have $dM/d\tau > 0$ for any $M < M^*$, and $dM/d\tau < 0$ for any $M > M^*$. Thus $M \to M^*$ as $\tau \to \infty$ in a monotonic manner. Finally we have Theorem 2.4.

A.5 PROOF OF THEOREM 2.4 IN SECTION 2.3.6

The discriminant Δ given by (15) can be rewritten as

$$\Delta = f(x) := x^2 - 2\left(\delta + \frac{2}{\mathscr{R}_0}a\eta\right)x + \delta^2 \tag{A.7}$$

with

$$x = \frac{\nu^*}{1 - \nu^*} = \frac{(\mathscr{R}_0 - 1)\delta + a\eta\theta_c}{\delta + a\eta(1 - \theta_c)}.$$
 (A.8)

There are two distinct positive roots x_- and x_+ ($x_- < x_+$) of the equation f(x) = 0 as given by (17). We have f(x) < 0 for and only for $x \in (x_-, x_+)$. Hence, from the inequality $x_- < x < x_+$ with (A.8), we can derive the condition $\theta_-^c < \theta_c < \theta_+^c$ with (16), since f(x) < 0 means the negative discriminant (15) so that the eigenvalue for E_{++} is imaginary. Therefore, a damped oscillation occurs if (16) is satisfied, These arguments prove Theorem 2.4.

A.6 PROOF OF COROLLARY 2.4.1 IN SECTION 2.3.6

When $\theta_{\pm}^{c} = 1 - 1/\Re_{0}$, we have $x_{\pm} = \Re_{0} - 1$ from (16) and (17), where x_{\pm} is defined by the roots of the equation f(x) = 0 given by (A.7) in A.5. We have $dx_{-}/d\Re_{0} > 0$ and $dx_{+}/d\Re_{0} < 0$ from (17). Then there are uniquely determined positive roots \Re_{0}^{inf} and \Re_{0}^{cc} of equations $x_{\pm} = \Re_{0} - 1$ with respect to \Re_{0} , where $\Re_{0}^{cc} > \Re_{0}^{inf} > 1$. We have $\Re_{0} - 1 < x_{-}$ for $\Re_{0} < \Re_{0}^{inf}$, $x_{-} < \Re_{0} - 1 < x_{+}$ for $\Re_{0}^{inf} < \Re_{0} < \Re_{0}^{cc}$, and $\Re_{0} - 1 > x_{+}$ for $\Re_{0} > \Re_{0}^{cc}$. Therefore, we have $\theta_{-}^{c} > 1 - 1/\Re_{0}$ for $\Re_{0} < \Re_{0}^{inf}$, $\theta_{-}^{c} < 1 - 1/\Re_{0} < \theta_{+}^{c}$ for $\Re_{0}^{inf} < \Re_{0} < \Re_{0}^{cc}$. The damped oscillation does not occur for $\Re_{0} < \Re_{0}^{inf}$ independently of θ_{c} .

For $\theta_{\pm}^{c} = 0$, we have $x_{\pm} = (\mathscr{R}_{0} - 1)\delta/(\delta + a\eta)$ from (16) and (17). There are uniquely determined positive roots \mathscr{R}_{0}^{c} and \mathscr{R}_{0}^{sup} of equations $x_{\pm} = (\mathscr{R}_{0} - 1)\delta/(\delta + a\eta)$ with respect to \mathscr{R}_{0} , where $\mathscr{R}_{0}^{sup} > \mathscr{R}_{0}^{c} > 1$. We have $(\mathscr{R}_{0} - 1)\delta/(\delta + a\eta) < x_{-}$ for $\mathscr{R}_{0} < \mathscr{R}_{0}^{c}$, $x_{-} < (\mathscr{R}_{0} - 1)\delta/(\delta + a\eta) < x_{+}$ for $\mathscr{R}_{0}^{c} < \mathscr{R}_{0} < \mathscr{R}_{0}^{c}$, $x_{-} < (\mathscr{R}_{0} - 1)\delta/(\delta + a\eta) < x_{+}$ for $\mathscr{R}_{0}^{c} < \mathscr{R}_{0} < \mathscr{R}_{0}^{sup}$. Therefore we have $\theta_{-}^{c} > 0$ for $\mathscr{R}_{0} < \mathscr{R}_{0}^{c}$, $\theta_{-}^{c} < 0 < \theta_{+}^{c}$ for $\mathscr{R}_{0}^{cc} < \mathscr{R}_{0} < \mathscr{R}_{0}^{sup}$, and $\theta_{+}^{c} < 0$ for $\mathscr{R}_{0} > \mathscr{R}_{0}^{sup}$. The damped oscillation does not occur for $\mathscr{R}_{0} > \mathscr{R}_{0}^{sup}$ independently of θ_{c} .

We note that there is no case where $0 < \theta_{-}^{c} < \theta_{+}^{c} < 1 - 1/\mathscr{R}_{0}$ from the definition of θ_{\pm}^{c} . Therefore, we can subsequently find it not valid that $\mathscr{R}_{0}^{cc} \leq \mathscr{R}_{0}^{c}$. Hence, as a consequence, $\mathscr{R}_{0}^{sup} > \mathscr{R}_{0}^{cc} > \mathscr{R}_{0}^{c} > \mathscr{R}_{0}^{inf} > 1$. Further, we have $1 - \mathscr{R}_{0} < \theta_{-}^{c}$ for $\mathscr{R}_{0} < \mathscr{R}_{0}^{inf}$, $1 - \mathscr{R}_{0} > \theta_{-}^{c} > 0$ for $\mathscr{R}_{0}^{o} < \mathscr{R}_{0}^{c}$, and $\theta_{-}^{c} < 0$ for $\mathscr{R}_{0} < \mathscr{R}_{0}^{c}$. On the other hand, we have $1 - \mathscr{R}_{0} < \theta_{+}^{c}$ for $\mathscr{R}_{0} < \mathscr{R}_{0}^{cc}$, $1 - \mathscr{R}_{0} > \theta_{+}^{c} > 0$ for $\mathscr{R}_{0}^{cc} < \mathscr{R}_{0} < \mathscr{R}_{0}^{cc}$, and $\theta_{-}^{c} < 0$ for $\mathscr{R}_{0} > \mathscr{R}_{0}^{cc}$, when $\theta_{-} < 0$ for $\mathscr{R}_{0} < \mathscr{R}_{0}^{cc}$. When $\theta_{c} \ge 1 - 1/\mathscr{R}_{0}^{cc}$, if $\mathscr{R}_{0} \le \mathscr{R}_{0}^{cc}$, then E_{++} does not exist since $\mathscr{R}_{0} \le \mathscr{R}_{0}^{cc} \le (1 - \theta_{c})^{-1}$. If $\mathscr{R}_{0} > \mathscr{R}_{0}^{cc}$, we have $\mathscr{R}_{0} - 1 > x_{+}$, that is, $\theta_{+}^{c} < 1 - 1/\mathscr{R}_{0}$. There is no damped oscillation since $\theta_{c} \ge 1 - 1/\mathscr{R}_{0}^{cc} > 1 - 1/\mathscr{R}_{0} > \theta_{+}^{c}$. Therefore the damped oscillation does not occur for $\theta_{c} \ge 1 - 1/\mathscr{R}_{0}^{cc}$ independently of \mathscr{R}_{0}^{cc} . These arguments prove the Corollary 2.4.1.

A.7 PROOF OF COROLLARY 2.4.2 IN SECTION 2.3

If $\delta = 0$, we have $v^* = \theta_c$ and the discriminant (11) as

$$\Delta = \frac{\theta_{\rm c}}{1 - \theta_{\rm c}} \left(\frac{\theta_{\rm c}}{1 - \theta_{\rm c}} - \frac{4 \alpha \eta}{\mathscr{R}_0} \right).$$

The discriminant Δ becomes negative if and only if

$$\theta_{\rm c} < \theta_+^{\rm c} = \frac{4a\eta}{\mathscr{R}_0 + 4a\eta}.\tag{A.9}$$

We have $\theta_+^c \ge 1 - 1/\mathscr{R}_0$ if and only if

$$\mathscr{R}_0 \leqslant \mathscr{R}_0^{cc} = \frac{1}{2}(1 + \sqrt{1 + 16a\eta}). \tag{A.10}$$

Hence, when the condition (A.10) is satisfied for $\Re_0 > 1$, the condition (A.9) holds for any $\theta_c < 1 - 1/\Re_0$. That is, $\Delta < 0$ whenever E_{++} exists with $\theta_c \in (0, 1 - 1/\Re_0)$. From Corollary 2.2.2, the damped oscillation does not occur for $\theta_c \ge 1 - 1/\Re_0$. This proves Corollary 2.4.2(i). Unless (A.10) is not satisfied, that is, if $\Re_0 > \Re_0^{cc}$, then we have $\theta_+^c < 1 - 1/\Re_0$. Hence in this case, from Corollary 2.2.2, we obtain the result of Corollary 2.4.2(ii). For the other case, we have $\theta_c \ge 1 - 1/\Re_0$ or $\theta_c \ge \theta_+^c$, that is, E_{++} does not exist or it exists with $\Delta \ge 0$. Therefore a damped oscillation occurs only in the cases described in Corollary 2.4.2.

A.8 PROOF OF COROLLARY 2.4.3 IN SECTION 2.3

From (A.8) as $a\eta \rightarrow \infty$, we have $x \rightarrow \theta_c/(1-\theta_c)$ and

$$\Delta \to \left(\frac{\theta_{c}}{1-\theta_{c}}\right) + \delta^{2} - \left(\frac{\theta_{c}}{1-\theta_{c}}\right) \lim_{\alpha\eta\to\infty} \left(\delta + \frac{1}{\mathscr{R}_{0}}\alpha\eta\right) < 0.$$

Besides, $x \to \mathscr{R}_0 - 1$ and

$$\Delta \to (\mathscr{R}_0 - 1 - \delta)^2 \ge 0$$

as $a\eta \rightarrow +0$ or $\delta \rightarrow \infty$. These results prove Corollary 2.4.3.

A.9 PROOF OF COROLLARY 2.4.4 IN SECTION 2.3.6

With $\theta_c = 0$, the discriminant (15) becomes

$$\Delta|_{\theta_{c}=0} = \frac{\delta}{\mathscr{R}_{0}^{2}(a\eta+\delta)^{2}}g(a\eta),$$

where

$$g(a\eta) := \left\{ \delta - 4\left(1 - \frac{1}{\mathscr{R}_0}\right) \right\} (a\eta)^2$$
$$-2\delta \left\{ 2\left(1 - \frac{1}{\mathscr{R}_0}\right) + (\mathscr{R}_0 - 1 - \delta) \right\} a\eta + \delta(\mathscr{R}_0 - 1 - \delta)^2$$

with (14). For $\delta \neq 4(1-1/\mathscr{R}_0)$, we have $g(a\eta) < 0$ when $4(1-1/\mathscr{R}_0) < \delta < \min\{\mathscr{R}_0, (\mathscr{R}_0+2)(1-1/\mathscr{R}_0)\}$ with $\mathscr{R}_0 > 2$ and $(a\eta)_-^c < a\eta < (a\eta)_+^c$, or when $\delta < 4(1-1/\mathscr{R}_0)$ and $a\eta > (a\eta)_+^c$, where $(a\eta)_{\pm}^c$ are given in (18). For $\delta = 4(1-1/\mathscr{R}_0)$, we have

$$g(a\eta) = 4\left(1 - \frac{1}{\mathscr{R}_0}\right)^2 \left\{ (\mathscr{R}_0 - 4)^2 \left(1 - \frac{1}{\mathscr{R}_0}\right) - 2(\mathscr{R}_0 - 2)a\eta \right\}.$$

Then $g(a\eta) < 0$ for $a\eta > (\mathscr{R}_0 - 4)^2 (\mathscr{R}_0 - 1)/\{2\mathscr{R}_0 (\mathscr{R}_0 - 2)\} > 0$ with $\mathscr{R}_0 > 2$. These prove Corollary 2.4.4.

B.1 DERIVATION OF \mathscr{R}_0 IN SECTION 3.2

In our model, the new cases consist of residents and visitors. Therefore, from the conceptual definition and the mathematical feature of the basic reproduction number, we shall derive it here from the following conditions:

$$\frac{d(\mathrm{I}_v+\mathrm{I}_r)}{dt}\bigg|_{0<\mathrm{I}_v+\mathrm{I}_r\ll 1}>0 \text{ for } \mathscr{R}_0>1; \ \frac{d(\mathrm{I}_v+\mathrm{I}_r)}{dt}\bigg|_{0<\mathrm{I}_v+\mathrm{I}_r\ll 1}<0 \text{ for } \mathscr{R}_0<1,$$

for the initial condition (21) with $0 < I_v(0) + I_r(0) = 0 + I_{r0} \ll 1$. This is because the basic reproduction number is defined as the expected number of new infectives produces by one infective individual in an environment consisting of only susceptibles. Following Assumption H₃ in Section 3.1.1, we shall adopt the initial condition (21) in order to define the basic reproduction number. Then, in place of the above conditions, we can use the followings:

$$\left.\frac{d(I_v+I_r)}{dt}\right|_{t=0} > 0 \text{ for } \mathscr{R}_0 > 1; \left.\frac{d(I_v+I_r)}{dt}\right|_{t=0} < 0 \text{ for } \mathscr{R}_0 < 1$$

for the initial condition (21) with $0 < I_v(0) + I_r(0) = 0 + I_{r0} \ll 1$. Remark that, in this context about the situation to define the basic reproduction number of our model, the initial infective must be a resident, which matches Assumption H₃ in Section 3.1.1. Then, making use of (21), we have

$$\frac{d(I_v+I_r)}{dt}\bigg|_{t=0} = \Big\{\beta\frac{(1-\rho)\mathfrak{m}+S_{r0}}{N+\mathfrak{m}} + \varepsilon\beta\frac{\rho\mathfrak{m}+R_{r0}}{N+\mathfrak{m}} - \gamma\Big\}I_{r0}.$$

Thus we find that

$$\frac{d(I_v + I_r)}{dt} \bigg|_{t=0} > 0 \text{ if and only if } \frac{\beta}{\gamma} \frac{(1-\rho)m + S_{r0}}{N+m} + \frac{\varepsilon\beta}{\gamma} \frac{\rho m + R_{r0}}{N+m} > 1.$$

Consequently we can define the basic reproduction number \mathcal{R}_0 as follows:

$$\begin{aligned} \mathscr{R}_{0} &= \sup_{(S_{r0},R_{r0})} \left\{ \frac{\beta}{\gamma} \frac{(1-\rho)m + S_{r0}}{N+m} + \frac{\epsilon\beta}{\gamma} \frac{\rho m + R_{r0}}{N+m} \right\} \\ &= \sup_{S_{r0}} \left\{ \frac{\beta}{\gamma} \frac{(1-\rho)m + S_{r0}}{N+m} + \frac{\epsilon\beta}{\gamma} \frac{\rho m + N - S_{r0}}{N+m} \right\} \\ &= \sup_{S_{r0}} \left\{ \frac{\beta}{\gamma} \frac{(1-\rho)m}{N+m} + \frac{\epsilon\beta}{\gamma} \frac{\rho m + N}{N+m} + \frac{(1-\epsilon)\beta}{\gamma} \frac{S_{r0}}{N+m} \right\} \\ &= \frac{\beta}{\gamma} \frac{(1-\rho)m + N}{N+m} + \frac{\epsilon\beta}{\gamma} \frac{\rho m}{N+m}. \end{aligned}$$

This formula can be rewritten as given by (22) to clarify the meaning.

B.2 PROOF OF THEOREM 3.2 IN SECTION 3.3.2

From the arguments in the first paragraph of Section 3.3.2, we find that the dynamics given by (23) with $\epsilon = 0$ necessarily approaches the dynamics with the following limiting system in terms of the visitor population:

$$\frac{d\tilde{x}_{v}}{d\tau} = (1-\rho)c - \mathscr{R}_{00} \frac{\mu}{1+\mu} \tilde{y}_{v} \tilde{x}_{v} - c\tilde{x}_{v};$$

$$\frac{d\tilde{y}_{v}}{d\tau} = \mathscr{R}_{00} \frac{\mu}{1+\mu} \tilde{y}_{v} \tilde{x}_{v} - (1+c) \tilde{y}_{v}.$$
(B.1)

The feasible equilibria are $\tilde{E}_0(1-\rho, 0)$ and $\tilde{E}_+(\tilde{x}^*_v, \tilde{y}^*_v)$ with (24). The former \tilde{E}_0 corresponds to the disease-eliminated equilibrium for the system (23), $E_{00}(1-\rho,0,0,0)$, and so does the latter \tilde{E}_+ to the endemic equilibrium $E_{+0}(\tilde{x}^*_v, \tilde{y}^*_v, 0, 0)$. The endemic equilibrium E_{+0} can exist when and only when the condition (25) is satisfied. By the local stability analysis with the eigenvalues of the Jacobi matrix at the equilibrium, we can easily find that the endemic equilibrium E_{+0} is locally asymptotically stable when it exists. In the following part, we shall consider its global stability.

First we set the following mathematical result on the boundedness for the solution of the system (B.1):

Lemma B.1. For any initial condition $(\tilde{x}_v(0), \tilde{y}_v(0))$ in the domain

$$D := \left\{ \left(\widetilde{x}_{v}, \widetilde{y}_{v} \right) \mid \widetilde{x}_{v} > 0, \ \widetilde{y}_{v} > 0, \ \widetilde{x}_{v} + \widetilde{y}_{v} < 1 - \rho \right\}, \tag{B.2}$$

the solution $(\tilde{x}_v(\tau), \tilde{y}_v(\tau))$ of (B.1) stays in D for any $\tau > 0$.

Proof. We can obtain the following features from (B.1) for the initial condition $(\tilde{x}_v(0), \tilde{y}_v(0)) \in D$:

$$\frac{d\widetilde{x}_{v}}{d\tau}\bigg|_{x_{v}=0} = 1-\rho > 0; \ \widetilde{y}_{v}(\tau) = \widetilde{y}_{v}(0) \exp\left[\int_{0}^{\tau} \left\{\mathscr{R}_{00}\frac{\mu}{1+\mu}\widetilde{x}_{v}(s) - (1+c)\right\}ds\right] > 0.$$

The first inequality indicates that \tilde{x}_v cannot reach 0 from the initial value $\tilde{x}_v(0) > 0$. The second equation indicates that \tilde{y}_v is necessarily positive for any $\tau > 0$ and $\tilde{y}_v(0) > 0$. Then, from

$$\left. \frac{d \big(\widetilde{x}_v + \widetilde{y}_v \big)}{d \tau} \right|_{\widetilde{x}_v + \widetilde{y}_v = 1 - \rho} = - \widetilde{y}_v < 0$$

for $\tilde{y}_v > 0$, we can find that $\tilde{x}_v + \tilde{y}_v$ cannot become $\tilde{x}_v + \tilde{y}_v = 1 - \rho$ for any $\tau > 0$ and initial condition in D.

Lemma B.1 means that the domain D is invariant for the dynamics given by (B.1). Further, when the endemic equilibrium \tilde{E}_+ exists, satisfying the condition (25), we find from (24) that

$$\begin{split} \widetilde{x}_{v}^{*} + \widetilde{y}_{v}^{*} &= (1-\rho)\frac{c}{c+1} + \frac{1+\mu}{\mu}\frac{1}{\mathscr{R}_{00}} \\ &< (1-\rho)\frac{c}{c+1} + \frac{1+\mu}{\mu}\frac{1}{c+1}(1-\rho)\frac{\mu}{1+\mu} = 1-\rho. \end{split}$$

Hence we have the following result:



Figure 25: Application of the isocline method for the system (B.1) when the condition (25) is (a) not satisfied; (b) satisfied.

Lemma B.2. When $\widetilde{E}_{+}(\widetilde{x}_{v}^{*}, \widetilde{y}_{v}^{*})$ exists, it must belong to the domain D defined by (B.2).

When the condition (25) is not satisfied, that is, when the endemic equilibrium does not exist, we can easily find that the disease-eliminated equilibrium is globally asymptotically stable, making use of the isocline method shown by Figure 25(a). In contrast, as seen from Figure 25(b), when the condition (25) is satisfied and the endemic equilibrium \tilde{E}_+ exists, the stability cannot be determined only by the isocline method.

When the endemic equilibrium \tilde{E}_+ exists, let us consider the following function of $(\tilde{x}_v, \tilde{y}_v)$ in the domain D defined by Lemma B.1:

$$V(\widetilde{\mathbf{x}}_{v}, \widetilde{\mathbf{y}}_{v}) := \frac{1}{2} \left\{ (\widetilde{\mathbf{x}}_{v}^{*} - \widetilde{\mathbf{x}}_{v}) + (\widetilde{\mathbf{y}}_{v}^{*} - \widetilde{\mathbf{y}}_{v}) \right\}^{2} + \left(\widetilde{\mathbf{x}}_{v}^{*} + c \frac{1 + \mu}{\mathscr{R}_{00} \mu} \right) \left(\widetilde{\mathbf{y}}_{v} - \widetilde{\mathbf{y}}_{v}^{*} - \widetilde{\mathbf{y}}_{v}^{*} \ln \frac{\widetilde{\mathbf{y}}_{v}}{\widetilde{\mathbf{y}}_{v}^{*}} \right).$$
(B.3)

It can be easily found that $V(\tilde{x}_v^*, \tilde{y}_v^*) = 0$ and $V(\tilde{x}_v, \tilde{y}_v) > 0$ for any $(\tilde{x}_v, \tilde{y}_v) \neq (\tilde{x}_v^*, \tilde{y}_v^*)$ in D. Further, making use of (B.1), we can derive

$$\frac{dV}{d\tau} = -\left(\widetilde{x}_{v}^{*} - \widetilde{x}_{v}\right)^{2} - \frac{\mu}{1+\mu}\widetilde{x}_{v}^{*}\left(\widetilde{y}_{v}^{*} - \widetilde{y}_{v}\right)^{2},$$

which becomes zero only for $(\tilde{x}_v, \tilde{y}_v) = (\tilde{x}_v^*, \tilde{y}_v^*)$, and negative for any $(\tilde{x}_v, \tilde{y}_v) \neq (\tilde{x}_v^*, \tilde{y}_v^*)$ in D. These features of V indicates that it is a Lyapunov function according to the endemic equilibrium \tilde{E}_+ for the system (B.1). Therefore, when the endemic equilibrium \tilde{E}_+ exists for the system (B.1), it is globally asymptotically stable with respect to the dynamics given by (B.1) with the initial condition in D. Since the dynamics of (23) with $\epsilon = 0$ must eventually approach that of (B.1), this result shows the global stability of the endemic equilibrium for (23).

B.3 PROOF OF THEOREM 3.4 IN SECTION 3.3.4

The characteristic equation $det(J_{00} - \lambda E) = 0$ with the Jacobian matrix J_{00} according to the disease-eliminated equilibrium $E_{00}(1 - \rho, 0, 0, 0)$ for the system (23) becomes

$$(-1-\lambda)(-\omega-\lambda) \begin{vmatrix} \{1-(1-\varepsilon)\rho\}\mathscr{R}_{00}\frac{\mu}{1+\mu}-(1+\varepsilon)-\lambda & \{1-(1-\varepsilon)\rho\}\mathscr{R}_{00}\frac{\mu}{1+\mu}\\ \varepsilon\mathscr{R}_{00}\frac{\mu}{1+\mu} & \varepsilon\mathscr{R}_{00}\frac{1}{1+\mu}-1-\lambda \end{vmatrix} = 0.$$

Hence we have two negative eigenvalues -1 and $-\omega$ with the other two given by the roots of the equation that the above 2×2 determinant is equal to zero. Both of them have negative real parts if and only if

$$\begin{cases} \left\{1 - (1 - \epsilon)\rho\right\} \mathscr{R}_{00} \frac{\mu}{1 + \mu} - (1 + c) + \epsilon \mathscr{R}_{00} \frac{1}{1 + \mu} - 1 < 0; \\ \left[\left\{1 - (1 - \epsilon)\rho\right\} \mathscr{R}_{00} \frac{\mu}{1 + \mu} - (1 + c)\right] \left(\epsilon \mathscr{R}_{00} \frac{1}{1 + \mu} - 1\right) \\ - \left\{1 - (1 - \epsilon)\rho\right\} \mathscr{R}_{00} \frac{\mu}{1 + \mu} \cdot \epsilon \mathscr{R}_{00} \frac{\mu}{1 + \mu} > 0, \end{cases}$$

that is,

$$\begin{cases} \left\{1 - (1 - \epsilon)\rho\right\} \mathscr{R}_{00} \frac{\mu}{1 + \mu} - (1 + c) + \epsilon \mathscr{R}_{00} \frac{1}{1 + \mu} - 1 < 0; \\ \left\{1 - (1 - \epsilon)\rho\right\} \mathscr{R}_{00} \frac{\mu}{1 + \mu} - (1 + c) + (1 + c)\epsilon \mathscr{R}_{00} \frac{1}{1 + \mu} < 0. \end{cases} \end{cases}$$

Therefore we find that the second inequality gives the necessary and sufficient condition that every eigenvalue has a negative real part. It can be expressed as the inverse inequality of (27). At the same time, we can find that, if the inverse of the above second inequality is satisfied, there exists an eigenvalue with a positive real part. Then the disease eliminated equilibrium E_{00} is unstable.

B.4 PROOF OF COROLLARY 3.4.1 IN SECTION 3.3.4

It is sufficient to show that $\{G(\mu, \rho)\}^{-1} < 1/(\epsilon \mathscr{R}_{00})$ when $\mathscr{R}_0 \leq 1$. From the definition of \mathscr{R}_0 defined by (22), we have $\mathscr{R}_0 \leq 1$ if and only if

$$\frac{1}{\varepsilon \mathscr{R}_{00}} \ge \frac{1}{\varepsilon} \Big\{ 1 - (1 - \varepsilon) \rho \frac{\mu}{1 + \mu} \Big\}.$$

Then we can derive

.

$$\begin{split} \frac{1}{\varepsilon} \bigg\{ 1 - (1 - \varepsilon)\rho \frac{\mu}{1 + \mu} \bigg\} &- \{G(\mu, \rho)\}^{-1} \\ &= \frac{1 - \varepsilon}{\varepsilon} + \frac{\mu}{1 + \mu} \bigg\{ 1 - \frac{1}{\varepsilon} \frac{1}{1 + \varepsilon} - \rho \frac{1 - \varepsilon}{\varepsilon} \big(1 - \frac{1}{1 + \varepsilon} \big) \bigg\} \\ &> \frac{1 - \varepsilon}{\varepsilon} + \frac{\mu}{1 + \mu} \Big(1 - \frac{1}{\varepsilon} - \rho \frac{1 - \varepsilon}{\varepsilon} \Big) = \frac{1 - \varepsilon}{\varepsilon} \bigg\{ 1 - (1 - \rho) \frac{\mu}{1 + \mu} \bigg\} \ge 0. \end{split}$$

Therefore, we find that $\{G(\mu, \rho)\}^{-1} < 1/(\varepsilon \mathscr{R}_{00})$ when $\mathscr{R}_0 \leq 1$, and then the inverse equality of (27) is satisfied.

B.5 PROOF OF THEOREM 3.5 IN SECTION 3.3.4

First we consider the case of $\rho = 1$. From (29), we can derive the following equations about y_v^* and y_r^* :

$$y_r^* = \frac{(1+c)y_v^*}{1+cy_v^*}; \ \phi(y_v^*) \coloneqq \frac{\varepsilon \mathscr{R}_{00}}{1+\mu} \Big(\frac{1+c}{1+cy_v^*} + \mu\Big)(1-y_v^*) - (1+c) = 0. \ (B.4)$$

The function $\varphi(y)$ is monotonically decreasing in terms of y > 0, and $\varphi(1) = -(1+c) < 0$. Hence, if and only if $\varphi(0) > 0$, the equation $\varphi(y) = 0$ has a unique positive root $y = y_v^* < 1$. We remark from the first equation

of (B.4) that y_r^* is uniquely determined for each positive $y_v^* < 1$ such that $0 < y_r^* < 1$. Therefore, if and only if $\varphi(0) > 0$, the endemic equilibrium E_{++} exists when $\rho = 1$. It is easy to show that the condition that $\varphi(0) > 0$ is necessary and sufficient to make (27) hold with $\rho = 1$.

Next, from (28) in the case of $\mu > 0$ and $\rho < 1$, we can derive an equation $\psi(\zeta^*) = 0$ in terms of $\zeta^* := (1 - \rho - x_v^*)/x_v^*$ with

$$\psi(\zeta) := \frac{c(1+\mu)}{\mathscr{R}_{00}} - \frac{\varepsilon}{1/c + \varepsilon\zeta} - \frac{\mu}{1/c + 1 - \varepsilon} \Big\{ \varepsilon \frac{1/c + \rho(1-\varepsilon)}{1/c + 1 + \varepsilon\zeta} + \frac{(1-\varepsilon)(1-\rho)}{1+\zeta} \Big\}.$$
(B.5)

Since $x_v^* \in (0, 1-\rho)$ for the endemic equilibrium, we have $\zeta^* \in (0, \infty)$. We can easily find that $\psi(\zeta)$ is monotonically increasing in terms of $\zeta > 0$. Further $\psi(\zeta) \rightarrow c(1+\mu)/\mathscr{R}_{00}$ as $\zeta \rightarrow \infty$. Hence the equation $\psi(\zeta) = 0$ necessarily has a unique positive root ζ^* if and only if $\psi(0) < 0$, which can be easily proved to be equivalent to the condition (27). With $\zeta^* > 0$, the equilibrium value x_v^* is uniquely determined by $x_v^* = (1-\rho)/(1+\zeta^*)$.

On the other hand, with the second equation of (28), we can derive

$$\begin{split} x_v^* + y_v^* &= g(x_v^*) \coloneqq \frac{1/c}{1 + 1/c - \varepsilon} x_v^* + \frac{1 - \varepsilon}{1 + 1/c - \varepsilon} \Big\{ \frac{(1 - \rho)(1 + 1/c)}{1 + 1/c - \varepsilon} - \frac{\varepsilon}{1 - \varepsilon} \Big\} \\ &+ \frac{\varepsilon(1 - \rho)(1 + 1/c)}{(1 + 1/c - \varepsilon)^2} \frac{1/c + (1 - \varepsilon)\rho}{(1 + 1/c - \varepsilon)x_v^* + \varepsilon(1 - \rho)}. \end{split}$$

Then we can easily find that g(x) is concave with g(0) = 1 and $g(1 - \rho) = 1 - \rho < 1$, so that g(x) < 1 for $x \in (0, 1 - \rho)$. This result indicates that, if the equation $\psi(x) = 0$ has a unique positive root x_v^* such that $0 < x_v^* < 1 - \rho$, the value y_v^* is reasonably determined by the second equation of (28) such that $x_v^* + y_v^* < 1$. Moreover, by the third equation of (28), the value y_r^* is reasonably determined at the same time such that $0 < y_r^* < 1$. Finally, from these arguments, if and only if $\psi(0) < 0$, which is equivalent to the condition (27), the endemic equilibrium E_{++} uniquely exists when $\rho < 1$.

B.6 PROOF OF THEOREM 3.6 IN SECTION 3.3.4

The Jacobian matrix about the endemic equilibrium E_{++} becomes

$$J(E_{++}) := \begin{pmatrix} -(B+c) & -\frac{\mu}{1+\mu} \mathscr{R}_{00} x_v^* & 0 & -\frac{1}{1+\mu} \mathscr{R}_{00} x_v^* \\ (1-\varepsilon)B & \mu \Xi - (\varepsilon B + 1 + c) & 0 & \Xi \\ 0 & 0 & -(B+\omega) & 0 \\ 0 & \mu \Phi & (1-\varepsilon)B & \Phi - \varepsilon B - 1 \end{pmatrix}$$

where

$$\begin{split} B &:= \mathscr{R}_{00} \frac{y_r^* + \mu y_v^*}{1 + \mu}; \ \Phi := \frac{y_r^*}{y_r^* + \mu y_v^*}; \\ \Xi &:= \frac{1 + c}{\mu} \frac{\mu y_v^*}{y_r^* + \mu y_v^*} = \frac{1 + c}{\mu} \Big(1 - \frac{y_r^*}{y_r^* + \mu y_v^*} \Big) = \frac{1 + c}{\mu} (1 - \Phi). \end{split}$$

Then the characteristic polynomial for the eigenvalue λ about E_{++} can be obtained as $\left|J(E_{++})-\lambda E\right|=(B+\omega+\lambda)h(\lambda)$ with

$$h(\lambda) := - \begin{vmatrix} -(B+c+\lambda) & 0 & -\frac{1}{1+\mu} \mathscr{R}_{00} x_v^* \\ (1-\varepsilon)B & -(\varepsilon B+1+c+\lambda) & \Xi \\ 0 & \mu(\varepsilon B+1+\lambda) & \Phi-\varepsilon B-1-\lambda \end{vmatrix}$$
$$= \lambda^3 + a_2 \lambda^2 + a_1 \lambda + a_0, \qquad (B.6)$$

where

$$\begin{split} \mathfrak{a}_{2} &= c\Phi + 1 + c + B + 2\varepsilon B;\\ \mathfrak{a}_{1} &= (c + B + \varepsilon B)c\Phi + (B + c)(2\varepsilon B + 1) + \varepsilon B(\varepsilon B + 1) + \frac{(1 - \varepsilon)\mu}{1 + \mu} \mathscr{R}_{00} x_{v}^{*}B;\\ \mathfrak{a}_{0} &= \varepsilon B(B + c)(\varepsilon B + 1 + c\Phi) + \frac{(1 - \varepsilon)\mu}{1 + \mu} \mathscr{R}_{00} x_{v}^{*}B(\varepsilon B + 1). \end{split}$$

Thus we have a negative eigenvalue $\lambda = -(B + \omega)$ and the cubic equation $h(\lambda) = 0$ given by (B.6) to determine the other three eigenvalues for E_{++} .

Every coefficient of $h(\lambda)$ is positive: $a_2 > 0$, $a_1 > 0$, and $a_0 > 0$. Since $0 < \Phi < 1$, we have

$$\begin{split} a_2 > a'_2 &:= 1 + c + B + 2\varepsilon B > 0; \\ a_1 > a'_1 &:= (B + c)(2\varepsilon B + 1) + \varepsilon B(\varepsilon B + 1) + \frac{(1 - \varepsilon)\mu}{1 + \mu} \mathscr{R}_{00} x_v^* B > 0; \\ a_0 < a'_0 &:= \varepsilon B(B + c)(\varepsilon B + 1 + c) + \frac{(1 - \varepsilon)\mu}{1 + \mu} \mathscr{R}_{00} x_v^* B(\varepsilon B + c), \end{split}$$

and subsequently find that

$$\begin{split} a_{2}a_{1}-a_{0} > a_{2}^{\prime}a_{1}^{\prime}-a_{0}^{\prime} &= (\varepsilon B+1+c)(\varepsilon B+1)(c+B+\varepsilon B) \\ &+ (1+\varepsilon)B\big\{(B+c)(2\varepsilon B+1)+\varepsilon B(\varepsilon B+1)\big\} \\ &+ \frac{(1-\varepsilon)\mu}{1+\mu}\big\{1+(1+\varepsilon)B\big\}\mathscr{R}_{00}x_{v}^{*}B > 0. \end{split}$$

$$(B.7)$$

Consequently from the Routh-Hurwitz criterion, we can find that all roots of $h(\lambda) = 0$ have negative real parts. Therefore it has been proved that every eigenvalue for E_{++} has negative real part. This result shows Theorem 3.6 about the local stability of E_{++} when it exists.

B.7 PROOF OF THEOREM 3.10 AND COROLLARY 3.10.3 IN SECTION 3.3.5

As shown in the proof for Theorem 3.5 (Appendix B.5), the endemic size can be determined by the unique positive root $\zeta = \zeta^*$ of the equation $\psi(\zeta) = 0$ with (B.5) for $\rho < 1$ under the condition (27). From the equation $\psi(\zeta^*) = 0$, we can derive

$$\frac{\partial \zeta^*}{\partial \mu} = \frac{1 + \varepsilon c \zeta^* - \varepsilon \mathscr{R}_{00}}{K \mu \mathscr{R}_{00} (1/c + \varepsilon \zeta^*)}$$
(B.8)

with

$$\mathsf{K}:=\frac{\varepsilon^2}{(1/c+\varepsilon\zeta^*)^2}+\frac{\mu}{1/c+1-\varepsilon}\Big\{\varepsilon^2\frac{1/c+(1-\varepsilon)\rho}{(1/c+1+\varepsilon\zeta^*)^2}+\frac{(1-\varepsilon)(1-\rho)}{(1+\zeta^*)^2}\Big\}>0.$$

First for $\epsilon \mathscr{R}_{00} \leq 1$, we find from (B.8) that $\partial \zeta^* / \partial \mu > 0$ for any $\zeta^* > 0$ and $\mu > 0$. Since $x_v^* = (1 - \rho)/(1 + \zeta^*)$, we then have $\partial x_v^* / \partial \mu < 0$. Lastly we can get the following result:

Lemma B.3. x_v^* is monotonically decreasing in terms of $\mu > 0$ when $\rho < 1$ and $\varepsilon \mathscr{R}_{00} \leqslant 1$.

Next for $\epsilon \mathscr{R}_{00} > 1$, the partial derivative (B.8) indicates that $\partial \zeta^* / \partial \mu > 0$ if and only if $\zeta^* > \zeta_c := (\epsilon \mathscr{R}_{00} - 1)/(\epsilon c)$, while $\partial \zeta^* / \partial \mu < 0$ if and only if $\zeta^* < \zeta_c$. Since $\psi(\zeta)$ is monotonically increasing in terms of $\zeta > 0$, such that $\psi(\zeta) < 0$ for $\zeta < \zeta^*$ and $\psi(\zeta) > 0$ for $\zeta > \zeta^*$ under the condition (27), we find that $\partial \zeta^* / \partial \mu > 0$ if and only if $\psi(\zeta_c) < 0$, while $\partial \zeta^* / \partial \mu < 0$ if and only if $\psi(\zeta_c) > 0$. Since ζ_c is independent of μ , this result indicates that the sign of $\partial \zeta^* / \partial \mu$ is determined independently of μ . Hence we obtain the following lemma:

Lemma B.4. $x_v^* > 0$ is monotonic in terms of $\mu > 0$ when $\rho < 1$.

Now we can derive

$$\psi(\zeta_{c}) = \frac{\mu\{\epsilon c - 1 + \epsilon^{2} \mathscr{R}_{00} + \epsilon(1 - \epsilon) \mathscr{R}_{00} \rho\}}{\epsilon \mathscr{R}_{00}(1 + \epsilon \mathscr{R}_{00}/c)\{1 + (\epsilon \mathscr{R}_{00} - 1)/(\epsilon c)\}}.$$
 (B.9)

Thus we can result that $\psi(\zeta_c) < 0$ if and only if $\rho < \rho_c$ where ρ_c is defined by (37). That is, $\partial \zeta^* / \partial \mu > 0$ if and only if $\rho < \rho_c$. Lastly we have

Lemma B.5. $x_v^* > 0$ is monotonically decreasing in terms of $\mu > 0$ if and only if $\rho < \rho_c$ when $\rho < 1$ and $\epsilon \mathscr{R}_{00} > 1$.

It can be easily found that $\rho_c < 1$ when $\epsilon \mathscr{R}_{00} > 1$.

In the critical case of $\rho = \rho_c > 0$ with $\epsilon \Re_{00} > 1$, we have $\psi(\zeta_c) = 0$ in (B.9), which means that $\zeta^* = \zeta_c$ and subsequently $\partial \zeta^* / \partial \mu = 0$. Actually, from $\zeta^* = \zeta_c$, we find that $x_v^* = (1 - \rho)/(1 + \zeta_c)$, the endemic sizes y_r^* , y_v^* , and z^* defined by (36) are independent of μ , as given by (38) which are derived by (28). This result gives Corollary 3.10.3.

On the other hand, from (28), we can easily find that the endemic sizes y_r^* , y_v^* , and z^* are monotonically decreasing in terms of $x_v^* > 0$. Therefore, from Lemma B.4, we can get the following lemma:

Lemma B.6. The endemic sizes y_r^* , y_v^* , and z^* are monotonic in terms of μ when $\rho < 1$.

Consequently, from Lemmas B.3, B.4, B.5, and B.6, we can obtain the result of Theorem 3.10 for $\rho < 1$.

B.8 PROOF OF COROLLARY 3.10.1 IN SECTION 3.3.5

In case of $\rho = 1$, the unique positive root of $\varphi(y) = 0$ with (B.4) gives the endemic size y_v^* under the condition (27) with $\rho = 1$ (Appendix B.5). Then, from the equation $\varphi(y_v^*) = 0$, we can derive

$$\frac{\partial y_{v}^{*}}{\partial \mu} = \frac{1 - y_{v}^{*} - (1 + c)/(\epsilon \mathscr{R}_{00})}{(1 + c)^{2}/(1 + cy_{v}^{*})^{2} + \mu}.$$
(B.10)

When $\epsilon \mathscr{R}_{00} \leq 1 + c$, we have $\partial y_v^* / \partial \mu < 0$ for any $\mu > 0$. Therefore we obtain the following result:

Lemma B.7. The endemic sizes y_r^* and y_v^* are monotonically decreasing in terms of μ when $\rho = 1$ and $\epsilon \mathscr{R}_{00} \leqslant 1 + c$.

From the first equation in (B.4), we note that the endemic size y_r^* is monotonically increasing in terms of y_v^* .

When $\epsilon \mathscr{R}_{00} > 1 + c$, $\partial y_v^* / \partial \mu > 0$ if and only if $y_v^* < y_v^c := 1 - (1 + c)/(\epsilon \mathscr{R}_{00})$, while $\partial y_v^* / \partial \mu < 0$ if and only if $y_v^* > y_v^c$. Since $\varphi(y)$ is monotonically decreasing in terms of y > 0, such that $\varphi(y) > 0$ for $y < y_v^*$ and $\varphi(y) < 0$ for $y > y_v^*$. Hence, $\partial y_v^* / \partial \mu > 0$ if and only if $\varphi(y_v^c) < 0$, while $\partial y_v^* / \partial \mu < 0$ if and only if $\varphi(y_v^c) > 0$. Now we can derive

$$\phi(\boldsymbol{y}_v^c) = \frac{c(1+c)}{(1+\mu)(\varepsilon \mathscr{R}_{00} - c)} > 0$$

for $\epsilon \mathscr{R}_{00} > 1 + c$. Therefore we have the following result:

Lemma B.8. The endemic sizes y_r^* and y_v^* are monotonically decreasing in terms of μ when $\rho = 1$ and $\epsilon \mathscr{R}_{00} > 1 + c$.

Consequently, from Lemmas B.7 and B.8, the endemic sizes y_r^* and y_v^* are monotonically decreasing in terms of μ when $\rho = 1$.

C.1 PROOF OF THEOREM 4.1 IN SECTION 4.3.2

Since the first (n + 1) equations equation in the model (44) is closed, we shall consider the reduced system of the first (n + 1) equations in model (44). We can derive the Jacobi matrix $J(E_0)$ for the equilibrium E_0 is given by

$$J(E_0) = \begin{pmatrix} -1 & -b_1 & -b_2 & \cdots & -b_n \\ 0 & b_1 \left(1 - \frac{1}{\mathcal{R}_{0,1}} \right) & 0 & \cdots & 0 \\ \vdots & \vdots & \vdots & \ddots & \cdots \\ 0 & 0 & 0 & \cdots & b_n \left(1 - \frac{1}{\mathcal{R}_{0,n}} \right) \end{pmatrix}.$$

It is clear that the eigenvalues of $J(E_0)$ are $\lambda_0 = -1 < 0$ and $\lambda_k = b(1 - 1/\Re_{0,k})$ (k = 1, 2, 3, ..., n). Thus E_0 is unstable if there exists ℓ where $\ell \in \Omega$, that is, $\Re_0 > 1$. For $\Re_0 \leq 1$, from $du/d\tau$, if $u \to 1$, we have $\sum_{k=1}^n b\nu_i \to 0$. Thus we have $\nu_k \to 0$ as $\tau \to \infty$ (k = 1, 2, ..., n). If $u \to u^* < 1$, since

$$\sum_{k=1}^n \frac{d\nu_k}{d\tau} < \sum_{k=1}^n b\nu_k(u-1) < 0$$

for $\mathscr{R}_0 \leq 1$, then $\nu_k \to 0$ as $\tau \to \infty$ (k = 1,2,...,n). Thus we have $u \to 1$ as $\tau \to \infty$ from $du/d\tau$. Therefore, we have shown that E_0 is globally asymptotically stable if and only if $\mathscr{R}_0 \leq 1$, while it is unstable when $\mathscr{R}_0 > 1$.

Next we consider the local stability of the endemic equilibrium E_{ℓ} for the model (44). Making use of (45), we derive the characteristic equation of the Jacobi matrix $J(E_{\ell})$ for the equilibrium E_{ℓ}

$$\left\{\lambda^{2} + \mathscr{R}_{0,\ell}\lambda + (1+\gamma_{\ell}+\eta_{\ell})\left(1-\frac{1}{\mathscr{R}_{0,\ell}}\right)\right\} \prod_{k\neq\ell} \left\{(1+\gamma_{k}+\eta_{k})\left(\frac{\mathscr{R}_{0,k}}{\mathscr{R}_{0,\ell}}-1\right)-\lambda\right\} = 0$$

There are two eigenvalues λ_1 and λ_2 satisfying

$$\lambda_1 + \lambda_2 = -\mathscr{R}_{0,\ell} < 0 \quad \text{and} \quad \lambda_1 \lambda_2 = (1 + \gamma_\ell + \eta_\ell) \left(1 - \frac{1}{\mathscr{R}_{0,\ell}} \right) > 0.$$

Thus λ_1 and λ_2 have negative real parts. The other eigenvalues are negative if and only if $\mathscr{R}_{0,\ell} > \mathscr{R}_{0,k}$ for any k = 1, 2, ..., n and $k \neq \ell$. Thus the endemic equilibrium E_ℓ is locally asymptotically stable. If there exist some k such that $\mathscr{R}_{0,k} > \mathscr{R}_{0,\ell}$, there is a positive eigenvalue for the characteristic equation of the Jacobi matrix $J(E_\ell)$ for the endemic equilibrium E_ℓ . Thus we have E_ℓ is unstable. Therefore, there is a unique locally asymptotically stable endemic equilibrium E_ℓ with $\mathscr{R}_{0,\ell} = \mathscr{R}_0$. These arguments show the proof of Theorem 4.1.

C.2 PROOF OF THEOREM 4.2 IN SECTION 4.3.2

From (46), we can derive

$$\frac{\partial z^*}{\partial \gamma_{\ell}} = \frac{1}{\mathfrak{b}_{\ell}(\mathfrak{a}_{\ell}+1)(1+\gamma_{\ell}+\eta_{\ell})^2} \{-(1+\gamma_{\ell}+\eta_{\ell})^2 + \mathfrak{b}_{\ell}(\eta_{\ell}-\mathfrak{a}_{\ell})\}.$$

If

$$\mathfrak{b}_{\ell}(\mathfrak{\eta}_{\ell}-\mathfrak{a}_{\ell})\leqslant (1+\mathfrak{\eta}_{\ell})^2,$$

we have $\partial z^* / \partial \gamma_{\ell} < 0$ for any $\gamma_{\ell} > 0$. Then z^* is monotonically decreasing in terms of γ_{ℓ} . If

$$\mathfrak{b}_{\ell}(\mathfrak{q}_{\ell}-\mathfrak{a}_{\ell})>(1+\mathfrak{q}_{\ell})^{2},$$

since $1 + \gamma_{\ell} + \eta_{\ell} < b_{\ell}$, we have

$$\begin{split} \frac{\partial z^*}{\partial \gamma_{\ell}} \bigg|_{\gamma_{\ell} \to b_{\ell} - (1+\eta_{\ell}) - 0} &= \frac{1}{b_{\ell} (a_{\ell} + 1)(1+\gamma_{\ell} + \eta_{\ell})^2} \{ -b_{\ell}^2 + b_{\ell} (\eta_{\ell} - a_{\ell}) \} \\ &= \frac{1}{(a_{\ell} + 1)(1+\gamma_{\ell} + \eta_{\ell})^2} \{ -b_{\ell} + (\eta_{\ell} - a_{\ell}) \} \\ &< \frac{1}{(a_{\ell} + 1)(1+\gamma_{\ell} + \eta_{\ell})^2} \{ -(1+\eta_{\ell}) + (\eta_{\ell} - a_{\ell}) \} \\ &= -\frac{1}{(1+\gamma_{\ell} + \eta_{\ell})^2}. \end{split}$$

Thus there exists a unique positive root of $\partial z^* / \partial \gamma_{\ell} = 0$ which is given by

$$\gamma_{\ell} = -(1+\eta_{\ell}) + \sqrt{b_{\ell}(\eta_{\ell} - a_{\ell})}$$

where z^* takes a maximum value. We have shown the proof of Theorem 4.2.

C.3 PROOF OF LEMMA 4.1 IN SECTION 4.3.3

From the equations of v_1 , v_j (j = 2, 3, ..., n-1), and v_n in the reduced system of model (42), we have

$$\begin{aligned} \nu_{1}(\tau) &= \nu_{1}(0) \exp\left[b\int_{0}^{\tau}u(s) + \varepsilon\left\{\sum_{k=2}^{n}\nu_{k}(s)\right\}ds - \frac{\tau}{\mathcal{R}_{0,1}}\right];\\ \nu_{j}(\tau) &= \nu_{j}(0) \exp\left[b\int_{0}^{\tau}u(s) + \varepsilon\left\{\sum_{k=j+1}^{n}\nu_{k}(s) - \sum_{k=1}^{j-1}\nu_{k}(s)\right\}ds - \frac{\tau}{\mathcal{R}_{0,j}}\right];\\ \nu_{n}(\tau) &= \nu_{n}(0) \exp\left[b\int_{0}^{\tau}u(s) - \varepsilon\left\{\sum_{k=1}^{n-1}\nu_{k}(s)\right\}ds - \frac{\tau}{\mathcal{R}_{0,n}}\right].\end{aligned}$$

Hence it holds that $v_1(\tau) > 0$, $v_j(\tau) > 0$, and $v_n(\tau) > 0$ for any $\tau > 0$ with $v_1(0) = v_1^0 > 0, v_j(0) = v_j^0 > 0 \ (j = 2, 3, ..., n-1), v_n(0) = v_n^0 > 0.$ Besides, since $\frac{du}{d\tau}\big|_{u=0} = 1 > 0$, we have $u(\tau) > 0$ for any $\tau > 0$ with $u(0) = u^0 > 0$. Next, from (42), we have

$$\begin{aligned} \frac{d}{d\tau} \Big(u + \sum_{k=1}^{n} v_k \Big) \Big|_{u + \sum_{k=1}^{n} v_k \ge 1} &= \left[1 - u - \sum_{k=1}^{n} (1 + \gamma_k + \eta_k) v_k \right]_{u + \sum_{k=1}^{n} v_k \ge 1} \\ &= \left[1 - u - \sum_{k=1}^{n} v_k - \sum_{k=1}^{n} (\gamma_k + \eta_k) v_k \right]_{1 - u - \sum_{k=1}^{n} v_k \le 0} < 0. \end{aligned}$$

so that $u + \sum_{k=1}^{n} v_k < 1$ for any $\tau > 0$ with $u(0) + \sum_{k=1}^{n} v_k(0) = u^0 +$ $\sum_{k=1}^{n} v_k^0 = \overline{1. \text{ Consequently we have Lemma 4.1.}}$

C.4 PROOF OF LEMMA 4.2 IN SECTION 4.3.3

From (42), we have

$$\begin{aligned} \frac{dv_{1}}{d\tau} < b\left(u + \sum_{k=2}^{n} v_{k} - \frac{1}{\mathscr{R}_{0,1}}\right) v_{1} = b\left(u + \sum_{k=1}^{n} v_{k} - v_{1} - \frac{1}{\mathscr{R}_{0,1}}\right) v_{1} < b\left(1 - v_{1} - \frac{1}{\mathscr{R}_{0,1}}\right) v_{1}; \\ (C.1) \\ \frac{dv_{j}}{d\tau} < b\left(u + \sum_{\substack{k=1\\k\neq j}}^{n} v_{k} - \frac{1}{\mathscr{R}_{0,j}}\right) v_{j} = b\left(u + \sum_{k=1}^{n} v_{k} - v_{j} - \frac{1}{\mathscr{R}_{0,j}}\right) v_{j} < b\left(1 - v_{j} - \frac{1}{\mathscr{R}_{0,j}}\right) v_{j}; \\ (C.2) \\ (C.3) \end{aligned}$$

for any $\tau > 0$ because $u + \sum_{k=1}^{n} v_j < 1$ for any $\tau > 0$ from Lemma 4.1. Thus we find that the right sides of (C.1), (C.2), and (C.3) are always nonpositive when $\Re_{0,k} \leq 1$ for all k. Therefore, when $\Re_{0,k} \leq 1$ for all k, we have $dv_k/d\tau < 0$ for all k and any $\tau > 0$. That is, $v_k \to 0$ as $\tau \to \infty$ for all k if $\Re_{0,k} \leq 1$ for all k. We have shown the proof of Lemma 4.2.

C.5 PROOF OF LEMMA 4.3 IN SECTION 4.3.3

The Jacobi matrix $J(\bullet)$ for the reduced system of model (42) is given by

$$J(\bullet) = \begin{pmatrix} -\frac{1}{u^*} & -bu^* & -bu^* & -bu^* & -bu^* & -bu^* & -bu^* \\ bv_1^* & \Delta_1(\bullet) & \varepsilon bv_1^* & \varepsilon bv_1^* & \cdots & \varepsilon bv_1^* & \varepsilon bv_1^* \\ bv_2^* & -\varepsilon bv_2^* & \Delta_2(\bullet) & \varepsilon bv_2^* & \cdots & \varepsilon bv_2^* & \varepsilon bv_2^* \\ bv_3^* & -\varepsilon bv_3^* & -\varepsilon bv_3^* & \Delta_3(\bullet) & \cdots & \varepsilon bv_3^* & \varepsilon bv_3^* \\ \vdots & \vdots & \vdots & \vdots & \ddots & \vdots & \vdots \\ bv_n^* & -\varepsilon bv_{n-1}^* & -\varepsilon bv_{n-1}^* & -\varepsilon bv_{n-1}^* & \cdots & \Delta_{n-1}(\bullet) & \varepsilon bv_{n-1}^* \\ bv_n^* & -\varepsilon bv_n^* & -\varepsilon bv_n^* & -\varepsilon bv_n^* & \cdots & -\varepsilon bv_n^* & \Delta_n(\bullet) \end{pmatrix}$$

where

$$\begin{split} &\Delta_{1}(\bullet) = bu^{*} + \varepsilon b \sum_{k=2}^{n} v_{k}^{*} - (1 + \gamma_{1} + \eta_{1}); \\ &\Delta_{j}(\bullet) = bu^{*} + \varepsilon b \left(\sum_{k=j+1}^{n} v_{k}^{*} - \sum_{k=1}^{j-1} v_{k}^{*} \right) - (1 + \gamma_{j} + \eta_{j}) \quad (1 < j < n); \ (C.5) \\ &\Delta_{n}(\bullet) = bu^{*} - \varepsilon b \sum_{k=1}^{n-1} v_{k}^{*} - (1 + \gamma_{n} + \eta_{n}). \end{split}$$

The characteristic equation of the Jacobi matrix $J(E_0)$ for the disease-free equilibrium E_0 is given by

$$(\lambda+1)\prod_{k=1}^{n}\left\{b\left(1-\frac{1}{\mathscr{R}_{0,k}}\right)-\lambda\right\}=0.$$

If $\Re_0 < 1$, all eigenvalues of $J(E_0)$ have negative real parts. Thus we have E_0 is locally asymptotically stable if $\Re_0 < 1$. If $\Re_0 > 1$, that is, if there exists k such that $\Re_{0,k} > 1$, there exists a positive eigenvalue $b(1 - 1/\Re_{0,k})$ for the Jacobi matrix $J(E_0)$. Thus we have the disease-free equilibrium E_0 is unstable if $\Re_0 > 1$. Taking account of Lemma 4.3, we have shown the proof of Theorem 4.3.

c.6 proof of theorem 4.4 in section 4.3.3

We can get the following result of the local stability of E_{ℓ} :

Lemma C.1. The single strain endemic equilibrium \mathbb{E}_{ℓ} exists and is locally asymptotically stable if conditions (49) and (50) are satisfied with $\mathscr{R}_{0,\ell} > 1$. If the condition with the inverse inequality of (49) or (50) is satisfied, it is unstable.

Proof. For the endemic equilibrium E_{ℓ} where only the infectives with strain ℓ persist and there is no infectives with strain k (k $\neq \ell$), from (C.4) with (C.5), we have the characteristic equation of the Jacobi matrix $J(E_{\ell})$ is given by

$$\left\{\lambda^{2} + \mathscr{R}_{0,\ell}\lambda + b\left(1 - \frac{1}{\mathscr{R}_{0,\ell}}\right)\right\} \prod_{k \neq \ell} \left(\Delta_{k}(\mathsf{E}_{\ell}) - \lambda\right) = 0.$$

There exists two eigenvalues λ_1 and λ_2 satisfying

$$\lambda_1 + \lambda_2 = -\mathscr{R}_{0,\ell} < 0 \quad \text{and} \quad \lambda_1 \lambda_2 = b\left(1 - \frac{1}{\mathscr{R}_{0,\ell}}\right) > 0.$$

Thus λ_1 and λ_2 have negative real parts with $\mathscr{R}_{0,\ell} > 1$. The other eigenvalues are negative if and only if $\Delta_k(\mathsf{E}_\ell) < 0$ for any $k \neq \ell$, that is,

Then we have the endemic equilibrium E_{ℓ} is locally asymptotically stable if the above conditions are satisfied. If there exists i where $k \neq \ell$ such that $\Delta_k(E_{\ell}) > 0$, that is, if the inverse inequality of (49) or (50) is satisfied, the endemic equilibrium E_{ℓ} is unstable.

Let us consider the following function of u, v_{ℓ} , and v_k when E_{ℓ} exists $(k = 1, 2, ..., n \text{ and } k \neq \ell)$:

$$V(u, v_1, ..., v_{\ell}, ..., v_n) := \frac{1}{2u^*} (u - u^*)^2 + \left(v_{\ell} - v_{\ell}^* - v_{\ell}^* \ln \frac{v_{\ell}}{v_{\ell}^*} \right) + \sum_{k \neq \ell} v_k$$

with $u^* = 1/\Re_{0,\ell}$, $v_{\ell}^* = (\Re_{0,\ell} - 1)/b$. Taking account of Lemma 4.1, it holds that V > 0 for any $(u, v_1, \dots, v_n) \in D \setminus \{E_{\ell}\}$ and V = 0 for E_{ℓ} . Then, making use of the equations in (42) we can derive the following time-derivative:

$$\frac{dV}{d\tau} = -\frac{1}{u^*}(u-u^*)^2 \left(1+b\sum_{k=1}^n v_k\right) + \left\{ \left(\frac{1}{\mathscr{R}_{0,\ell}} - \frac{1}{\mathscr{R}_{0,k}}\right) + \frac{\varepsilon}{b}(\mathscr{R}_{0,\ell}-1) \right\} \sum_{k=1}^{\ell-1} \frac{bv_k}{v_\ell} + \left\{ \left(\frac{1}{\mathscr{R}_{0,\ell}} - \frac{1}{\mathscr{R}_{0,k}}\right) - \frac{\varepsilon}{b}(\mathscr{R}_{0,\ell}-1) \right\} \sum_{k=\ell+1}^n \frac{bv_k}{v_\ell}.$$
(C.6)

The second term of (C.6) is negative when the condition (49) is satisfied, while the third term of (C.6) is negative when the condition (50) is satisfied. Then the derivative $dV/d\tau$ is always negative under the conditions (49) and (50), taking account of Lemma 4.1. Besides, when the corresponding left and right sides of (49) and (50) are equal to each other, the second and third terms of (C.6) becomes zero, and the derivative $dV/d\tau$ is negative unless $u = u^*$. Since the value u temporally varies as long as $(u, v_1, \ldots, v_\ell, \ldots, v_n) \neq (u^*, 0, \ldots, v_\ell^*, \ldots, 0)$ even when $u = u^*$ at a certain finite moment, we note from (C.6) that the derivative $dV/d\tau$ is negative for almost every time $\tau > 0$ unless $(u, v_1, \ldots, v_\ell, \ldots, v_n) \neq (u^*, 0, \ldots, v_\ell^*, \ldots, 0)$.

As a result, the derivative $dV/d\tau$ is always negative under the conditions (49) and (50), and then the function V is always decreasing as time passes.

C.7 PROOF OF LEMMA 4.4 IN SECTION 4.3.5

Lemma 4.4 (i) and (ii) can be easily shown. Let us consider the existence of the endemic equilibrium E_{12} . Since the first three equations in the model (51) is closed, we can consider the endemic equilibrium $\overline{E}_{12}(u^*, v_1^*, v_2^*)$ for the system of the first three equations in the model (51), where u^* , v_1^* , and v_2^* are given in (55). Let us consider first the positiveness of u^* , v_1^* , and v_2^* . From (55), the positiveness holds if and only if

$$\frac{1}{\mathcal{R}_{0,2}} < u^* < \frac{1}{\mathcal{R}_{0,1}}.$$
 (C.7)

We remark that, if the condition (C.7) is satisfied, then $\Re_{0,1} < \Re_{0,2}$, $0 < u^* < 1$, and $\Re_{0,2} > 1$. We have the first inequality of the condition (C.7) can be rewritten as the inequality (52), and the second inequality of the condition (C.7) can be rewritten as the inequality (53) with (54). If the condition (52) is satisfied, since $\Re_{0,2} > 1$, we have

$$\frac{1}{1-\hat{\varepsilon}} \Big(\frac{1}{\mathscr{R}_{0,1}} - \hat{\varepsilon} \Big) < \frac{1}{\mathscr{R}_{0,2}} < 1$$

Thus it is necessary that $\Re_{0,1} > 1$ to satisfy the condition (52). Since

$$\frac{1}{(1-\hat{\varepsilon})+\hat{\varepsilon}\mathscr{R}_{0,1}}\cdot\frac{1}{\mathscr{R}_{0,1}}-\frac{1}{1-\hat{\varepsilon}}\Big(\frac{1}{\mathscr{R}_{0,1}}-\hat{\varepsilon}\Big)=\frac{1}{1-\hat{\varepsilon}}\cdot\frac{\hat{\varepsilon}(\mathscr{R}_{0,1}-1)}{(1-\hat{\varepsilon})+\hat{\varepsilon}\mathscr{R}_{0,1}}>0$$

when $\Re_{0,1} > 1$, we have the condition (C.7) is satisfied if and only if the inequalities (53) and (52) are satisfied with $\Re_{0,2} > \Re_{0,1} > 1$. Thus \overline{E}_{12} exists only if the conditions (53) and (52) are satisfied. Since the corresponding equilibrium E_{12} for the system (51) satisfies that $u^* + v_1^* + v_2^* + q_1^* + q_2^* + w^* = 1$, we have the consistent equilibrium values for E_{12} exists if and only if u^* , v_1^* , and v_2^* are positive. Therefore, the endemic equilibrium \overline{E}_2 for the system of the first three equations in the model (51), and the corresponding endemic equilibrium E_2 for the system (51) exist if and only if the conditions (53) and (52) are satisfied with $\Re_{0,2} > \Re_{0,1} > 1$. When E_{12} exists, E_1 and E_2 is necessarily exist. We have shown the proof of Lemma 4.4 (iii).

c.8 proof of theorem 4.7 in section 4.3.5

Theorem 4.7 (i) and (ii) can be shown by Theorem 4.4. As for the equilibrium E_{12} , we can derive the Jacobi matrix $J(E_{12})$

$$J(E_{12}) = \begin{pmatrix} -\frac{1}{u^*} & -bu^* & -bu^* \\ bv_1^* & 0 & \epsilon b_1 v_1^* \\ bv_2^* & -\epsilon bv_2^* & 0 \end{pmatrix}$$

From (55), we can derive the characteristic equation given by

$$\lambda^{3} + \frac{1}{u^{*}}\lambda^{2} + b^{2}(u^{*}v_{1}^{*} + u^{*}v_{2}^{*} + \varepsilon^{2}v_{1}^{*}v_{2}^{*})\lambda + \frac{1}{u^{*}}\varepsilon^{2}b^{2}v_{1}^{*}v_{2}^{*} = 0.$$
 (C.8)

Since all coefficients of equation (C.8) are positive and they satisfy that

$$\frac{b^2}{u^*}(u^*\nu_1^*+u^*\nu_2^*+\varepsilon^2\nu_1^*\nu_2^*)-\frac{b^2}{u^*}\varepsilon^2\nu_1^*\nu_2^*=b^2(\nu_1^*+\nu_2^*)>0,$$

according to Routh-Hurwitz stability criterion, the endemic equilibrium E_{12} is locally asymptotically stable. Then we consider the following function of u, v_1 , and v_2 when E_{12} exists:

$$V(u,v_1,v_2) := \frac{1}{2u^*}(u-u^*)^2 + \left(v_1 - v_1^* - v_1^* \ln \frac{v_1}{v_1^*}\right) + \left(v_2 - v_2^* - v_2^* \ln \frac{v_2}{v_2^*}\right).$$

It is easily found that V > 0 for any $(u, v_1, v_2) \in \overline{D} \setminus \{E_{12}\}$ and V = 0 for E_{12} , where $\overline{D} := \{(u, v_1, v_2) \mid u > 0, v_1 > 0, v_2 > 0, u + v_1 + v_2 < 1\}$. Then, making use of the first three equations in (51) we can derive the following time-derivative:

$$\frac{dV}{d\tau} = -\frac{1}{u^*}(u - u^*)^2(b_1v_1 + b_2v_2 + 1).$$
(C.9)

Since the value u temporally varies as long as $(u,v_1,v_2) \neq (u^*,v_1^*,v_2^*)$ even when $u = u^*$ at a certain finite moment, we note from (C.9) that the derivative $dV/d\tau$ is negative for almost every time $\tau > 0$ unless $(u,v_1,v_2) = (u^*,v_1^*,v_2^*)$. Hence the function V is always decreasing as time passes. Thus the function V defines a Lyapunov function for the equilibrium E_{12} . Finally we have that, whenever the two strain endemic equilibrium E_{12} exists, it is globally asymptotically stable. The above arguments show the proof of Theorem 4.7 (iii).